

# Transmission Congestion Management and Security Cost Optimization in Deregulated Electricity Markets

A thesis presented for the degree of

Doctor of Philosophy

by

Jiawei Zhao

Supervisor: Professor K.L.Lo

Power System Research Group

Department of Electronic and Electrical Engineering

University of Strathclyde

2019

### Declaration

This thesis is the result of the author's original research. It has been completed by the author and has not been previously submitted for examination which has led to the award of a degree.

The copyright of this thesis belongs to the author under the terms of the United Kingdom Copyright Acts, as qualified by University of Strathclyde Regulation 3.50. Due acknowledgement must always be made of the use of any material contained in, or derived from, this thesis.

Signature:

Date:

### Acknowledgement

I would like to express my deeply appreciation to my supervisor, Professor K.L.Lo, the head of the Power System Research Group (PSRG), University of Strathclyde, who offered me this precious opportunity of studying as a PhD. I am truly grateful for his patient guidance and constant support not only on my research work but beyond.

My gratitude also goes to my colleagues, who are and were studying inside and outside the PSRG. I am thankful for the friendship, the company and the unforgettable memories over these challenging years. Here I will give my special thanks to Dr Jianfeng Lu, for his patient help and strong encouragement during my PhD study.

Last but not least, my sincere gratitude goes to my family especially my parents, for their unlimited love, unconditional support and firm expectation in my life.

### Abstract

Two sensitivity factors, namely the power transfer distribution factor (PTDF) and the impedance based sensitivity factor (ZS), are proposed respectively in the thesis to calculate the power flow contribution of each bus to the transmission network congestion. Generation re-dispatch and the corresponding security cost allocation are based on the calculated power flow contribution. The proposed generation re-dispatch method and the proposed security cost allocation method both with two sensitivity factors are validated respectively on four test systems including a modified IEEE-14 bus system, a modified IEEE-30 bus system, a modified IEEE-57 bus system and a modified IEEE-118 bus system. The experimental results are obtained including the generation re-dispatch MW amount of each generator selecting different locations of slack bus, the resulted security cost and the security cost allocation to each load. The comparisons between two sensitivity-based methods on both generation re-dispatch and security cost allocation show that the PTDF-based method and the ZS-based method are both dependent on the location of slack bus selection, which increase the risks of inaccuracy and complexity. The ZS-based method involves more generators in re-dispatch stage than the PTDF-based method, consequently results in higher security cost. Moreover, with intense level of congestion, such security cost difference between the two methods becomes larger. Compared with the ZS-based method, the PTDF-based method is more reliable on security cost allocation since in large bus system, for example, the 118-bus system, the allocation results by the ZS-based method appears unreasonable.

## Contents

Contents		iv
List of Fig	ures	X
List of Tal	bles	xiv
Nomencla	ure	xvii
Chapter 1		1
Introdu	ction	1
1.1.	Overview	1
1.2.	Objectives of the Thesis	5
1.3.	Original Contribution of the Thesis	6
1.4.	Outline of the Thesis	7
1.5.	Publication	10
Chapter 2		11
Chapter 2 Deregul	ated Electricity Market	11 11
Chapter 2 Deregul 2.1.	ated Electricity Market Introduction	<b>11</b> <b>11</b> 11
Chapter 2 Deregul 2.1. 2.2.	ated Electricity Market Introduction Traditional Vertically Integrated Electricity Utility	<ul> <li>11</li> <li>11</li> <li>11</li> <li>13</li> </ul>
Chapter 2 Deregul 2.1. 2.2. 2.3.	ated Electricity Market Introduction Traditional Vertically Integrated Electricity Utility Generally Understanding of Electricity Industry Deregulation	<ul> <li>11</li> <li>11</li> <li>11</li> <li>13</li> <li>14</li> </ul>
Chapter 2 Deregul 2.1. 2.2. 2.3. 2.4.	ated Electricity Market Introduction Traditional Vertically Integrated Electricity Utility Generally Understanding of Electricity Industry Deregulation Difference between Previous and Existing Deregulated Electricity	11 11 13 14 y Market
Chapter 2 Deregul 2.1. 2.2. 2.3. 2.4. Struct	ated Electricity Market Introduction Traditional Vertically Integrated Electricity Utility Generally Understanding of Electricity Industry Deregulation Difference between Previous and Existing Deregulated Electricity ures	11 11 13 14 y Market 16
Chapter 2 Deregul 2.1. 2.2. 2.3. 2.4. Struct 2.4.	ated Electricity Market Introduction Traditional Vertically Integrated Electricity Utility Generally Understanding of Electricity Industry Deregulation Difference between Previous and Existing Deregulated Electricity ures 1. Pool Market	11 11 13 14 y Market 16 17
Chapter 2 Deregul 2.1. 2.2. 2.3. 2.4. Struct 2.4. 2.4.	<ul> <li>ated Electricity Market</li> <li>Introduction</li> <li>Traditional Vertically Integrated Electricity Utility</li> <li>Generally Understanding of Electricity Industry Deregulation</li> <li>Difference between Previous and Existing Deregulated Electricity</li> <li>ures</li> <li>Pool Market</li> <li>Bilateral Market</li> </ul>	11 11 13 14 y Market 16 17 18
Chapter 2 Deregul 2.1. 2.2. 2.3. 2.4. Struct 2.4. 2.4. 2.4.	<ul> <li>ated Electricity Market</li> <li>Introduction</li> <li>Traditional Vertically Integrated Electricity Utility</li> <li>Generally Understanding of Electricity Industry Deregulation</li> <li>Difference between Previous and Existing Deregulated Electricity</li> <li>ures</li> <li>Pool Market</li> <li>Bilateral Market</li> <li>Hybrid Market</li> </ul>	11 11 13 14 y Market 16 17 18 20
Chapter 2 Deregul 2.1. 2.2. 2.3. 2.4. Struct 2.4. 2.4. 2.4. 2.4. 2.5.	<ul> <li>ated Electricity Market</li> <li>Introduction</li> <li>Traditional Vertically Integrated Electricity Utility</li> <li>Generally Understanding of Electricity Industry Deregulation</li> <li>Difference between Previous and Existing Deregulated Electricity</li> <li>ures</li> <li>Pool Market</li> <li>Bilateral Market</li> <li>Hybrid Market</li> <li>Independent System Operator</li> </ul>	11 11 13 14 y Market 16 17 18 20 21

2.6.1. UK	23
2.6.1.1. The England & Wales Pool	23
2.6.1.2. British Electricity Trading and Transmission Arrangements	
(BETTA)	27
2.6.2. Nordic Countries	31
2.6.3. US	34
2.6.3.1. PJM Interconnection	34
2.6.3.2. California Crisis	37
2.7. Summary	39
Chapter 3	41
Transmission Congestion Management in Deregulated Electricity Market	41
3.1. Introduction	41
3.2. Market-based Transmission Congestion Management Schemes	42
3.2.1. Pricing Schemes	44
3.2.1.1. Uniform Marginal Price	44
3.2.1.2. Locational Marginal Price	48
3.2.1.3. Zonal Price	54
3.2.2. Transaction Curtailment	58
3.2.2.1. First Come First Serve	58
3.2.2.2. Pro Rata (Proportional)	59
3.2.2.3. Minimum-net Curtailment	59
3.2.2.4. Compensative price	60
3.2.2.5. Willingness to Pay	60
3.2.3. Financial Instruments	61
3.2.3.1. Contract for Difference	61
3.2.3.2. Financial Transmission Right	62
3.3. Case Study	64
3.3.1. The Modified IEEE-14 Bus System	64
3.3.2. Results	66

3.3.2.1. Congestion Cost Allocation by Uniform Marginal Price	68
3.3.2.2. Congestion Cost Allocation by Locational Marginal Price	69
3.3.3. Comparison	71
3.3.4. Discussion	72
3.4. Summary	73
Chapter 4	74
Re-dispatch Generation in Deregulated Electricity Market with Power Transf	er
Distribution Factor	74
4.1. Introduction	74
4.2. Proposed Generation Re-dispatch Method in Deregulated Electricity Mar	ket
	75
4.2.1. Conventional Re-dispatch Generation in Electricity Market	75
4.2.2. Adjustments in the Proposed Re-dispatch Scheme	76
4.2.3. Illustration of the proposed re-dispatch generation scheme	78
4.3. The Algorithm of Power Transfer Distribution Factor	79
4.4. Mathematical Model for the Proposed Method with Power Transfer	
Distribution Factor	83
4.5. Case Studies	85
4.5.1. Test Description	86
4.5.2. Modified IEEE-14 Bus System	87
4.5.3. Modified IEEE-30 Bus System	92
4.5.4. Modified IEEE-57 Bus System	97
4.5.5. Modified IEEE-118 Bus System	103
4.5.6. Discussion	111
4.6. Summary	113
Chapter 5	114

### Re-dispatch Generation in Deregulated Electricity Market with Bus Impedance Matrix Sensitivity 114

5.1. Introduction	114
5.2. Proposed Generation Re-dispatch Method in Deregulated Electricity	Market
	115
5.3. The Admittance Matrix and the Impedance Matrix	115
5.3.1. The Admittance Matrix	115
5.3.2. The Impedance Matrix	117
5.4. The Algorithm of Bus Impedance Matrix Sensitivity	120
5.5. Mathematical Model for the Proposed Method with Bus Impedance	Matrix
Sensitivity	125
5.6. Case studies	127
5.6.1. Test Description	127
5.6.2. Modified IEEE-14 Bus System	128
5.6.3. Modified IEEE-30 Bus System	132
5.6.4. Modified IEEE-57 Bus System	136
5.6.5. Modified IEEE-118 Bus System	141
5.6.6. Discussion	145
5.7. Summary	146
Chapter 6	147
Security Cost Allocation in Deregulated Electricity market with Power Tr	ansfer
Distribution Factor	147
6.1. Introduction	147
6.2. Conventional Security Cost Allocation Schemes	148
6.2.1. Security Cost Allocation in Vertically Integrated Power Utility	148
6.2.2. Security Cost Allocation in Pool Electricity Market	148
6.2.3. Security Cost Allocation in Bilateral Electricity Market	149
6.3. Proposed Security Cost Allocation Method with Power Transfer	
Distribution Factor	151
6.3.1. Mechanism of the Proposed Security Cost Allocation Method with	n Power
Transfer Distribution Factor	151

6.3.2. Mathematical Model of the Proposed Security Cost Allocation Meth	od
with Power Transfer Distribution Factor	152
6.4. Case Studies	156
6.4.1. Test Description	156
6.4.2. Modified IEEE-14 Bus System	158
6.4.2.1. Comparison between Security Cost Allocation with Different S	lack
Bus Selections in the 14-bus Test System	159
6.4.2.2. Comparison between Pro Rata Method and PTDF Method in the	e 14-
bus Test System	162
6.4.3. Modified IEEE-30 Bus System	163
6.4.3.1. Comparison between Security Cost Allocation with Different S	lack
Bus Selections in the 30-bus Test System	165
6.4.3.2. Comparison between Pro Rata Method and PTDF Method in th	e 30-
bus Test System	170
6.4.4. Modified IEEE-57 Bus System	172
6.4.4.1. Comparison between Security Cost Allocation with Different S	lack
Bus Selections in the 57-bus Test System	173
6.4.4.2. Comparison between Pro Rata Method and PTDF Method in th	e 57-
bus Test System	179
6.4.5. Modified IEEE-118 Bus System	182
6.4.5.1. Comparison between Security Cost Allocation with Different S	lack
Bus Selections in the 118-bus Test System	185
6.4.5.2. Comparison between Pro Rata Method and PTDF Method in th	e
118-bus Test System	201
6.4.6. Discussion	206
6.5. Summary	207
Chapter 7	208
Security Cost Allocation in Deregulated Electricity Market with Bus Impeda	ince
Matrix Sensitivity	208

7.1. Introduction	208
7.2. Proposed Security Cost Allocation Method with Bus Impedance Matrix	
Sensitivity	209
7.2.1. Mechanism of the Proposed Security Cost Allocation Method with B	us
Impedance Matrix Sensitivity	209
7.2.2. Mathematical Model of the Proposed Security Cost Allocation Metho	od
with Bus Impedance Matrix Sensitivity	209
7.3. Case Studies	212
7.3.1. Test Description	212
7.3.2. Modified IEEE-14 Bus System	214
7.3.3. Modified IEEE-30 Bus System	216
7.3.4. Modified IEEE-57 Bus System	220
7.3.5. Modified IEEE-118 Bus System	225
7.3.6. Discussion	230
7.4. Summary	232
Chapter 8	233
Conclusion and Future Work	233
8.1. Conclusion	233
8.2. Future Work	235
References	239
Appendix	253
A. PTDF and ZS Sensitivity Factors in Modified IEEE-14 Bus System	253
B. PTDF and ZS Sensitivity Factors in Modified IEEE-30 Bus System	255
C. PTDF and ZS Sensitivity Factors in Modified IEEE-57 Bus System	258
D. PTDF and ZS Sensitivity Factors in Modified IEEE-118 Bus System	264

## **List of Figures**

Figure 2.1 Vertically Integrated Utility Structure	13
Figure 2.2 Structure of Unbundled Power Industry	15
Figure 2.3 the Structure of Electricity Industry at Privatization in England & Wales	24
Figure 2.4 the Industry Structure of the BETTA	28
Figure 2.5 the Diagrammatic Representation of BETTA Time Frame Process	29
Figure 2.6 the Diagrammatic Representation of Nord Pool Time Frame Process	33
Figure 2.7 Tree structure of PJM markets	35
Figure 3.1 Meshed network structure of a multi-nodes power system without congest	stion
	55
Figure 3.2 Multi-nodes power system split into three zones when congestion occurs	56
Figure 3.3 The modified IEEE-14 bus system	65
Figure 3.4 Comparison in congestion cost allocation	71
Figure 4.1 Illustration of the proposed generation re-dispatch scheme	78
Figure 4.2 The modified IEEE-14 bus system	88
Figure 4.3 14-bus system generator output variation in re-dispatch when LO=45%	91
Figure 4.4 14-bus system congestion costs of re-dispatch in five slack bus cases when	n LO
=45%	91
Figure 4.5 The modified IEEE-30 bus system	93
Figure 4.6 30-bus system generator output variation in re-dispatch when LO=52%	96
Figure 4.7 30-bus system congestion costs of re-dispatch in six slack bus cases w	when
LO=52%	97
Figure 4.8 The modified IEEE-57 bus system	98
Figure 4.9 57-bus system generator output variation in re-dispatch when LO=8%	101

Figure 4.10 57-bus system congestion costs of re-dispatch in seven slack bus cases	when
LO=8%	102
Figure 4.11 The modified IEEE-118 bus system	103
Figure 4.12 118-bus system generator output variation in re-dispatch when LO=999	6 109
Figure 4.13 118-bus system congestion cost of re-dispatch in all slack bus cases	when
LO=99%	110
Figure 5.1 $\pi$ Equivalent Circuit of Transmission Line j-k	120
Figure 5.2 The modified IEEE-14 bus system	128
Figure 5.3 14-bus system generator output variation in re-dispatch by two methods	when
LO=33%	130
Figure 5.4 14-bus system congestion cost of re-dispatch by two methods when LO	=33%
	130
Figure 5.5 Comparison of security cost by two methods in five different LO scenar	ios in
14-bus test system	131
Figure 5.6 The modified IEEE-30 bus system	132
Figure 5.7 30-bus system generator output variation in re-dispatch when LO=19%	134
Figure 5.8 30-bus system congestion cost of re-dispatch when LO=19%	134
Figure 5.9 Comparison of security cost by two methods in five different LO scenar	ios in
30-bus test system	135
Figure 5.10 The modified IEEE-57 bus system	137
Figure 5.11 57-bus system generator output variation in re-dispatch when LO=8%	139
Figure 5.12 57-bus system congestion cost of re-dispatch when LO=8%	139
Figure 5.13 Comparison of security cost by two methods in five different LO scenar	ios in
57-bus test system	140
Figure 5.14 The modified IEEE-118 bus system	141
Figure 5.15 118-bus system generator output variation in re-dispatch when LO=999	6 143
Figure 5.16 118-bus system congestion cost of re-dispatch when LO=99%	143
Figure 5.17 Comparison of security cost by two methods in five different LO scenar	ios in
118-bus test system	144
Figure 6.1 Test description for chapter 6	157

Figure 6.2 The modified IEEE-14 bus system	158
Figure 6.3 Security cost settings in the 14-bus test system	159
Figure 6.4 Security cost allocated to each load with different slack buses in t	he 14-bus test
system	160
Figure 6.5 Comparison between two methods on security cost allocation in t	he 14-bus test
system	163
Figure 6.6 The modified IEEE-30 bus system	164
Figure 6.7 Security cost settings in the 30-bus test system	165
Figure 6.8 Security cost allocated to each load with different slack buses in t	he 30-bus test
system	167
Figure 6.9 Comparison between two methods on security cost allocation in t	he 30-bus test
system	171
Figure 6.10 The modified IEEE-57 bus system	172
Figure 6.11 Security cost settings in the 57-bus test system	173
Figure 6.12 Security cost allocated to each load with different slack buses	in the 57-bus
test system	176
Figure 6.13 Comparison between two methods on security cost allocation	in the 57-bus
test system	181
Figure 6.14 The modified IEEE-118 bus system	182
Figure 6.15 Security cost settings in the 118-bus test system	184
Figure 6.16 Security cost allocated to each load with different slack buses i	n the 118-bus
test system	194
Figure 6.17 Security cost allocated to each load with different slack buses i	n the 118-bus
test system from L1 to L41	196
Figure 6.18 Security cost allocated to each load with different slack buses	in in the 118-
bus test system from L42 to L81	197
Figure 6.19 Security cost allocated to each load with different slack buses	in in the 118-
bus test system from L82 to L118	198
Figure 6.20 Comparison between two methods on security cost allocation i	n the 118-bus
test system	204

Figure 7.1 Test description for chapter 7	213
Figure 7.2 The modified IEEE-14 bus system	214
Figure 7.3 Comparison between three methods on security cost allocation in	the 14-bus
test system	215
Figure 7.4 The modified IEEE-30 bus system	217
Figure 7.5 Comparison between three methods on security cost allocation in	the 30-bus
test system	219
Figure 7.6 The modified IEEE-57 bus system	220
Figure 7.7 Comparison between three methods on security cost allocation in	the 57-bus
test system	223
Figure 7.8 The modified IEEE-118 bus system	225
Figure 7.9 Comparison between three methods on security cost allocation in t	the 118-bus
test system	229

## **List of Tables**

Table 3.1 Congestion Management Schemes in Different Market Structures	43
Table 3.2 Generation and Load Data	65
Table 3.3 Transmission Capacity of Each Transmission Line	66
Table 3.4 Generation Cost of Unconstrained Dispatch and Security-Constrained	Dispatch
	67
Table 3.5 Power Flows of Unconstrained Dispatch and Security-Constrained Dis	spatch 67
Table 3.6 Congestion Cost Allocation by Uniform Marginal Price	68
Table 3.7 Locational Marginal Prices in Unconstrained Dispatch and S	Security-
Constrained Dispatch	69
Table 3.8 Congestion Cost Allocation by Locational Marginal Price	70
Table 4.1 Generator and Load Data in 14 Bus System	88
Table 4.2 Generation Schedule and the Congested Line Power Flow in the Uncor	ıstrained
Dispatch in 14 Bus System	89
Table 4.3 Generator Output Variation in Each Slack Bus Selection Due to Re-dis	spatch in
14-Bus System when LO=45%	90
Table 4.4 Generator Data in 30 Bus System	93
Table 4.5 Load Data in 30 Bus System	94
Table 4.6 Generation Schedule and the Congested line power flow in the Uncor	istrained
Dispatch in 30 Bus System	95
Table 4.7 Generator Output Variation in Each Slack Bus Selection Due to Re-dis	spatch in
30-Bus System when LO=52%	96
Table 4.8 Generator Data in 57 Bus System	98
Table 4.9 Load Data in 57 Bus System	99
Table 4.10 Generation Schedule and the Congested Line Power Flow in the Uncor	ıstrained
Dispatch in 57 Bus System	100

Table 4.11 Generator Output Variation Due to Re-dispatch in 57-bus System when LO=8%
101
Table 4.12 Generator Data in 118 Bus System104
Table 4.13 Load Data in 118 Bus System104
Table 4.14 Generation Schedule and the Congested Line Power Flow in the Unconstrained
Dispatch in 118 Bus System 106
Table 4.15 Generator Output Variation in Each Slack Bus Selection Due to Re-dispatch
in 118 Bus System at LO=99% 108
Table 5.1 Generator Output Variation Due to Re-dispatch by Two Methods in 14-Bus
System When LO=33%129
Table 5.2 Generator Output Variation Due to Re-dispatch by Two Methods in 30-Bus
System When LO=19% 133
Table 5.3 Generator Output Variation Due to Re-dispatch by Two Methods in 57-Bus
System When LO=8% 138
Table 5.4 Generator Output Variation Due to Re-dispatch by Two Methods in 118-Bus
System When LO=99% 142
Table 6.1 Security Cost Allocated to Each Load with Different Slack Buses in the 14-bus
Test System 159
Table 6.2 Comparison between Two Methods on Security Cost Allocation in the 14-bus
Test System162
Table 6.3 Security Cost Allocated to Each Load with Different Slack Buses in the 30-bus
Test System 166
Table 6.4 Comparison between Two Methods on Security Cost Allocation in the 30-bus
Test System 170
Table 6.5 Security Cost Allocated to Each Load with Different Slack Buses in the 57-bus
Test System 174
Table 6.6 Comparison between Two Methods on Security Cost Allocation in 57-bus Test
System 179
Table 6.7 Security Cost Allocated to Each Load with Different Slack Buses in the 118-
bus Test System 186

XV

Table 6.8 Comparison between Two Methods on Security Cost Allocation in the	118-bus
Test System	202
Table 7.1 Comparison between Three Methods on Security Cost Allocation in the	e 14-bus
Test System	215
Table 7.2 Comparison between Three Methods on Security Cost Allocation in the	e 30-bus
Test System	218
Table 7.3 Comparison between Three Methods on Security Cost Allocation in the	e 57-bus
Test System	221
Table 7.4 Comparison between Three Methods on Security Cost Allocation in	the 118-
bus Test System	227
Table A. PTDF and ZS Sensitivity Factors Associated with Congested Transmiss	ion Line
3-4 in Modified IEEE-14 Bus System	253
Table B. PTDF and ZS Sensitivity Factors Associated with Congested Transmiss	ion Line
4-12 in Modified IEEE-30 Bus System	255
Table C. PTDF and ZS Sensitivity Factors Associated with Congested Transmiss	ion Line
7-29 in Modified IEEE-57 Bus System	258
Table D. PTDF and ZS Sensitivity Factors Associated with Congested Transmiss	ion Line
23-24 in Modified IEEE-118 Bus System	264

## Nomenclature

NETA	New Electricity Trading Arrangement
BETTA	British Electricity Trading and Transmission Arrangements
CfD	Contract for Difference
FTR	Financial Transmission Right
PTDF	Power Transfer Distribution Factor
ZS	Bus Impedance Based Sensitivity Factor
ISO	Independent System Operator
FERC	Federal Energy Regulatory Commission
CEGB	Central Electricity Generating Board
RECs	Regional Electricity Companies
GOAL	Generating Ordering and Loading
SMP	System Marginal Price
PPP	Pool Purchase Price
SMP	System Marginal Price
СР	Capacity Payment
LOLP	Loss of Load Probability
VOLL	Value of Loss Load

PSP	Pool Selling Price
NGC	National Grid Company
IPNs	Initial Physical Notifications
FPNs	Final Physical Notifications
SSP	System Sell Price
SBP	System Buy Price
BSUoS	Balancing Service Use of System
TSOs	Transmission System Operators
OTC	Over the Counter
RTO	Regional Transmission Organization
LMP	Locational Marginal Pricing
CPUC	California Public Utilities Commission
PX	Power Exchange
CAISO	California Independent System Operator
FACTs	Flexible AC Transmission System Devices
OPF	Optimal Power Flow
DCOPF	DC Optimal Power Flow
ACOPF	AC Optimal Power Flow
TCC	Transmission Congestion Contract
UMP	Uniform Marginal Price
LO	Level of Overloading
$C_{G_i}(P_{G_i})$	The bid-based generation cost of generator <i>i</i>

$P_{G_i}$	The active power generated by generator <i>i</i>
$P_{L_j}$	The demand prediction of load <i>j</i>
$N_G$	The total number of generators
$N_L$	The total number of loads
$P_{G_i}^{max}$	The maximum generation capacity of generator <i>i</i>
P <sub>linek</sub>	The power flow on the transmission line $k$
$P_{line_k}^{max}$	The maximum power flow transmission capacity of the transmission line $k$
bid price <sub>off</sub>	The bid price of the "constrained off" generator submitted to the system operator during the unconstrained dispatch
bid price <sub>on</sub>	The bid price of the "constrained on" generator submitted to the system operator during the unconstrained dispatch
$P_i^g$	The real power generated at node <i>i</i>
<i>C</i> <sub>0<i>i</i></sub>	The no-load cost coefficient of generator <i>i</i>
<i>C</i> <sub>1<i>i</i></sub>	The linear cost coefficient of generator <i>i</i>
<i>C</i> <sub>2<i>i</i></sub>	The quadratic cost coefficient of generator <i>i</i>
$P_i^d$	The real power demand at node <i>i</i>
<i>P<sub>ij</sub></i>	The power transmission flow from node $i$ to node $j$
P <sub>i,min</sub>	The minimum generation limitation of node <i>i</i>
P <sub>i,max</sub>	The maximum generation limitation of node <i>i</i>
$P_l$	The real transmission power flow on the line $l$
$P_l^{max}$	The real power maximum transmission capacity of line $l$

P <sub>loss</sub>	The transmission losses in the power system
$\lambda_0$	The Lagrangian multiplier of the whole system power balance constraint
Ν	The total number of nodes
L	The total number of transmission lines
$\mu_l$	The Lagrangian multiplier of transmission line constraint
$\pi_i^{min}$	The Lagrangian multipliers of minimum generation capacity of generator $i$
$\pi_i^{max}$	The Lagrangian multipliers of maximum generation capacity of generator $i$
$T_{i-1}$	The sensitivity factor for real power at node $i$ with line $l$ constraint
$LMP_i^{energy}$	The system marginal cost of node <i>i</i>
$LMP_i^{congestion}$	The cost of transmission congestion of node <i>i</i>
$LMP_i^{loss}$	The cost of transmission losses of node <i>i</i>
Power <sub>contract</sub>	The power volume signed in the CfD contract
price <sub>reference</sub>	The reference price which is often the spot market price
price <sub>strike</sub>	The agreed fixed price signed in the CfD contract
<i>Power<sub>FTR</sub></i>	The active power delivered from source node to sink node
price <sub>sink</sub>	The nodal price of the sink node
price <sub>source</sub>	The nodal price of the source node
$PTDF_i^{j-k}$	The PTDF of the line $j$ - $k$ respect to bus $i$
$\Delta P_{j-k}$	The active power flow variation of line <i>j</i> - <i>k</i>

$\Delta P_i$	The active power injection variation at bus <i>i</i>
$P_{j-k}$	The active power flow between bus $j$ and bus $k$
Vj	The voltage of bus <i>j</i>
$V_k$	The voltage of bus k
$ heta_j$	The phase angle of bus <i>j</i>
$ heta_k$	The phase angle of bus $k$
$Z_{j-k}$	The impedance connected between bus $j$ and bus $k$
$x_{j-k}$	The reactance connected between bus $j$ and bus $k$
$b_{j-k}$	The susceptance between bus $j$ and bus $k$
[ <i>P</i> ]	The matrix of active power injection at buses
[ heta]	The matrix of phase angle of bus voltages
[ <i>B</i> ]	The susceptance matrix
[ <i>B</i> *]	The matrix that eliminating one row and one column of $[B]$
$[M^*]$	The invert of [ <i>B</i> <sup>*</sup> ]
[ <i>M</i> ]	The matrix is from $[M^*]$ by adding one row and one column of 0 according to the corresponding reference bus or slack bus.
$M_{ji}$	The $j_{th}$ row and $i_{th}$ column element in the M matrix
$M_{ki}$	The $k_{th}$ row and $i_{th}$ column element in the M matrix
$C_{G_i}$	The incremental/decremental price of generator <i>i</i>
$\Delta P_{G_i}$	The incremental/decremental amount of active power injection of generator $i$

$PTDF_{G_i}^{j-k}$	The PTDF of the line $j$ - $k$ respect to bus $i$
$P_{j-k}^{lim}$	Transmission capacity limit of line <i>j</i> -k
$P_{j-k}^{uncon}$	The active power flow of line $j$ - $k$ in unconstraint dispatch
$P_{G_i}$	The generation amount of generator $i$ in unconstraint dispatch
$P_{G_i}^{max}$	The generation capacity of generator <i>i</i>
$P_{j-k}^{uncon}$	The active power of congested line $j$ - $k$ in unconstrained dispatch
$P_{j-k}^{lim}$	The limit value of active power transmission capacity of congested line $j$ - $k$
I <sub>i</sub>	The nodal current in bus <i>i</i>
Y <sub>ij</sub>	The admittance matrix of the n-bus power system
Vj	The voltage on bus <i>j</i>
Z <sub>ij</sub>	The impedance matrix of the n-bus power system
Y <sub>bus</sub> *	The reduced admittance matrix of the n-bus power system
$S_{j-k}$	The complex power flow of line $j$ - $k$ calculated at bus $j$
$V_j$	The bus voltage at bus <i>j</i>
$I_{j-k}^*$	The conjugate of current through line $j$ - $k$
Z <sub>ji</sub>	The $j_{th}$ row and $i_{th}$ column element in the impedance matrix
Ii	The nodal current at bus <i>i</i>
$I_{j-k}$	The current through line <i>j</i> - <i>k</i>
$V_k$	The bus voltage at bus k
$y_{j-k}$	The series admittance of the $\pi$ equivalent circuit of line <i>j</i> - <i>k</i>

$\mathcal{Y}_{j-k}^{sh}$	The shunt admittance of the $\pi$ equivalent circuit of line <i>j</i> - <i>k</i>
$Z_{ki}$	The $k_{th}$ row and $i_{th}$ column element in the impedance matrix
$I_i^*$	The conjugate of nodal current at bus <i>i</i>
P <sub>i</sub>	The active power injected at bus <i>i</i>
$Q_i$	The reactive power injected at bus <i>i</i>
V <sub>i</sub>	The voltage at bus <i>i</i>
$ZS_i^{j-k}$	The bus impedance matrix sensitivity of line $j$ - $k$ respect to real power injection at bus $i$
$\Delta P_{j-k}^i$	The portion of active power flow change through line $j$ - $k$ only associated with bus $i$ power injection change
$\Delta P_i$	The power injection change of bus <i>i</i>
$ZS_{G_i}^{j-k}$	The bus impedance matrix sensitivity of the line <i>j</i> - <i>k</i> respect to bus <i>i</i>
SC <sub>total</sub>	The total security cost
SC <sub>gens</sub>	The security cost portion for generators
SC <sub>loads</sub>	The security cost portion for loads
$SC_{G_i}$	The security cost allocated to generator <i>i</i>
$P_{G_i}$	The MW power generation of generator <i>i</i>
$SC_{L_j}$	The security cost allocated to load <i>j</i>
$P_{L_j}$	The MW power consumption of load <i>j</i>
$N_G$	The number of generators in the system

$N_L$	The number of loads in the system
$P_{j-k}$	The power flow MW value of the target transmission line $j-k$
$P_{j-k}^G$	The power flow MW value on line $j-k$ associated with all the generators
$P_{j-k}^L$	The power flow MW value on line $j$ - $k$ associated with all the loads
$PTDF_{L_i}^{j-k}$	The PTDF of the line $j-k$ respect to load bus $i$
$P_{L_i}$	The MW power energy withdrawn by load <i>i</i>
$P_{j-k}^{L_i}$	The power flow MW value associated with load <i>i</i>
$Ctb_{L_i}$	The power flow contribution percentage of load <i>i</i> to the congested line
$CA_{L_i}$	The allocated security cost to the load <i>i</i>
SC <sub>total</sub>	The total security cost due to congestion management
$ZS_{G_i}^{j-k}$	The ZS of the line $j$ - $k$ respect to generator bus $i$
$ZS_{L_i}^{j-k}$	The ZS of the line $j$ - $k$ respect to load bus $i$

## **Chapter 1**

### Introduction

#### 1.1. Overview

Power industry is initially controlled by one utility which is normally vertically integrated and regulated by the government. This utility controls generation, transmission and distribution services. Customers have no other choices but have to purchase electricity from the utility and the electricity price is at a set rate in the controlled area. The single utility holds all the information of power system operation, some of which are utilized to centrally dispatch and re-dispatch generation schedules. Moreover, all the planning activities of power system such as generation plant investment and maintenance of transmission network are also centrally supervised by the utility. This monopolistic utility model has been adopted for a long time. As time progresses, new requirements for power energy industry appear such as cost reducing, efficiency improvement and long-term development, which pushed the government to unbundle the vertically integrated utility into generation, transmission and distribution portions [1]. It is believed that with the restructuring of the power energy industry, a competitive atmosphere can be introduced to achieve the above requirements. Due to the market open access, incentives will be given so that investment from the society can contribute to long-term development of the power industry. After almost forty years discovery, global achievements on electricity industry restructuring have been gained such as the "England & Wales Pool" in the UK, which now is further evolved as the British Electricity Trading and Transmission Arrangements (BETTA). The original England & Wales Pool was a practical pool model electricity market while the BETTA is now an electricity market, which is built based on a bilateral market principle; the "Nord Pool" in Europe and the PJM interconnection in the US, which are both hybrid electricity models combining features of pool model and bilateral model [2-6].

In the power system, transmission lines are restricted by several constraints such as thermal constraint, voltage constraint and stability. To ensure the reliability of the transmission network system, steady state security needs to be assessed to monitor the temperature of transmission lines [7]. When the thermal constraint of the transmission line is exceeded, the transmission line will be overloaded, which is also known as transmission congestion. Congestion can cause cascade outages, increase losses, build up barrier to energy transactions, increase electricity price and result in market power, which above are threats to the system security, so congestion issue cannot be tolerated [8]. In the traditional vertically integrated structure, resolving the congestion can be done by re-dispatching the generation schedule because it is centrally controlled [9]. However, congestion management in deregulated electricity market becomes more complex and frequent since transmission network is based on an open access principle to all the market participants so power flows are determined by power trades. Market participants only consider the maximization of their economic interests rather than the transmission network physical limitations. Hence, congestion occurs when the transmission lines have not sufficient capacity to transfer the desired power in the open market [10]. The most direct way to solve transmission congestion is to expand transmission network to provide enough power transfer capacity. But planning and construction of a transmission line lasts a long time and the investment of transmission network costs a lot. Consequently, congestion management is regarded as a practical tool for efficiently utilizing the transmission network without violating the transmission constraints.

The aim of congestion management is to avoid congestion or to relieve congestion. A. Singla et al. [11] classifies the congestion management methodologies into non-market based methods and market based methods. The market based methods are implemented according to market mechanisms so economic signals can be sent to the market while this is not the case for non-market methods. Market based methods are more acceptable as they can enhance liquidity and transparency of the electricity market. The market based methods can be subdivided into pricing based methods, financial instruments, re-dispatch methods, load-side managements and sensitivity factors based methods. References [12] [13] and [14] give a detailed description of the locational marginal price (LMP) especially the components of the LMP. Zonal pricing method is explained in [15]. X.Ma et al. [16] conducts a comparison among the uniform marginal pricing (UMP), the LMP and the zonal pricing in the aspect of providing economic signals. To help market participants dampen the price volatility, some financial instruments are introduced into electricity market. References [17] and [18] have a description of the financial transmission rights (FTR), which is used to ensure efficient allocation of transmission access rights in the LMP-based market. Reference [19] introduces another financial instrument called contracts for differences (CfD), which is a mid or long-term two way contract between the generator and the consumer to hedge the price volatility. As known, re-dispatch method was widely used in the vertically integrated utility and the pool based market. Combined with corresponding market mechanism, it can still solve congestion issues. Reference [20] proposed a re-dispatch method combined with incremental and decremental bids submission mechanism to relieve the congestion. Apart from the generation-side congestion management methods, some methods attempt to solve the problem at the demand-side. References [21] and [22] respectively try to curtail the demand-side load to relieve congestion. Besides, some congestion management methods seek for the help of transmission network sensitivity factors to relieve congestion. In [23], the power transfer distribution factors (PTDF) is proved to be a useful index for congestion management since it can calculate the sensitivity of each bus contribution to the congestion. K. Singh et al. [24] presents another sensitivity factor for congestion management, namely bus impedance matrix based sensitivity.

This thesis firstly proposes a generation re-dispatch method to prevent the transmission congestion in deregulated electricity market. The mechanism of incremental and decremental bids submission for the generators is combined. This re-dispatch method

can be deemed as a minimum cost optimization associated with the optimal power flow (OPF) and the submitted bids from generators, at the same time, restricted by transmission capacity constraints and generation capacity constraints. During the optimization, two sensitivity factors, the power transfer distribution factor (PTDF) and the bus impedance based sensitivity factor (ZS), are introduced to respectively calculate the power flow contribution of each node in the power system to the congested transmission line. The similarity of these two sensitivity factors is that they are both trying to find out the relationship between active power change quantity of a particular transmission line and the active power change quantity at a particular bus. The difference between these two sensitivity factors is that they are derived from different algorithms. The PTDF-based factor is calculated according to the admittance matrix while the ZS-based factor is calculated according to the impedance matrix. Before tests, the connection between each sensitivity factor and the location of slack bus selection will be investigated. Then both sensitivity-based re-dispatch methods will be tested in four modified IEEE-bus systems ranging from 14-bus up to 118-bus. According to different locations of slack bus selection, results will be the re-dispatched MW amount of each generator and the total security cost due to re-dispatch. The result comparison between two sensitivity-based methods under the same slack bus selection and the same level of overloading of the congested transmission line will be obtained and discussed. Moreover, this thesis tries to balance the re-dispatch action and the resulted security cost. Hence, two security cost allocation methods respectively associated with the above two sensitivity factors in deregulated electricity market will be proposed. The resulted security cost due to the congestion will be allocated by the two methods respectively, which are based on the calculated power flow contribution by the PTDF factor and the ZS factor. Similarly, the security cost allocation results by PTDF-based method under different locations of slack bus selection will be indicated. Then the allocation performance comparison between two methods will be shown and discussed.

#### **1.2.** Objectives of the Thesis

The objectives of this thesis are shown as follows:

- To review former and current deregulated electricity market experiences across the world; discuss and compare the advantages and disadvantages of each deregulated electricity market model.
- To review typical transmission congestion management schemes in deregulated electricity market and to discuss and compare the pros and cons of each congestion management scheme.
- To propose a generation re-dispatch method associated with the sensitivity factor PTDF and the market bidding mechanism to relieve transmission congestion under a minimum cost principle. To investigate the influence of slack bus selection on the results of the PTDF-based method.
- To propose a generation re-dispatch method associated with the sensitivity factor ZS and the market bidding mechanism to solve the transmission congestion problem under a minimum cost principle, and to differentiate the influence of slack bus selection. Compare the re-dispatch results between the PTDF-based method and the ZS-based method.
- To develop a security cost allocation method associated with the sensitivity factor PTDF and to utilize it to calculate the power flow contribution of each node to the congested transmission line. Then allocate the security cost based on the calculated power flow contribution. To investigate the impact of slack bus selection on the results of the PTDF-based method.

• To develop a security cost allocation method associated with the sensitivity factor ZS. With the help of the ZS, to calculate the power flow contribution of each node to the congested transmission line. Then allocate the security cost based on the calculated power flow contribution. Compare the security cost allocation results between the PTDF-based method and the ZS-based method.

#### **1.3.** Original Contribution of the Thesis

The major original contributions of the thesis are listed as follows:

- A proposed generation re-dispatch method associated with the sensitivity factor PTDF and the market bidding mechanism is developed, which is used to relieve the transmission congestion in deregulated electricity market. The proposed method is capable to re-dispatch generation schedule for congestion management in four test systems. The effect of location of slack bus selection on the proposed PTDF-based method is verified.
- A proposed generation re-dispatch method associated with the sensitivity factor ZS and the market bidding mechanism is developed. The aim is to relieve the transmission congestion in deregulated electricity market. The method is tested in four test systems and proved to be capable. The dependency of the location of the slack bus selection of the ZS-based method is confirmed. Comparisons with the PTDF-based method indicate the generators involved in the ZS-based method for re-dispatch are more than the PTDF-based method. Consequently, the resulted security cost in the ZS-based method is more than the PTDF-based method. Moreover, with the intense congestion, the resulted security cost difference between two methods becomes larger.

- A proposed security cost allocation method associated with the sensitivity factor PTDF is developed. The method is based on the calculated power flow contribution of each bus to allocate the security cost as a result of congestion. The proposed method is applied to four test systems and the obtained results proved that the method is working. Nevertheless, the dependency of location of slack bus selection increases the risks of inaccuracy and complexity of security cost allocation.
- A proposed security cost allocation method using sensitivity factor ZS is developed, which is based on the calculated power flow contribution of each bus to allocate the security cost as a result of congestion. The method is tested in four systems and the results prove it is capable. Parallel comparisons between the ZS-based method and the PTDF-based method indicate that the above two methods have different ideas to allocate the same value of security cost. When implemented in large power system, the ZS-based method has the risk of providing unreasonable solutions.

#### **1.4.** Outline of the Thesis

Chapter 1 gives an introduction of the whole thesis. Firstly it provides a brief explanation about electricity market deregulation, transmission congestion, transmission congestion management and the work included in this thesis. Furthermore, the objectives, the major original contributions and the structure of the thesis are presented.

Chapter 2 describes the traditional vertically integrated electricity utility and its major drawbacks which can be regarded as the motivation of electricity industry restructuring. A description of the electricity industry deregulation is given including the history of deregulation, the new structure and the expectations of the deregulation. A pool

market model, a bilateral market model and a hybrid market model are introduced as the three typically deregulated electricity market models. The role of the independent system operator is explained together with its characteristics and functions. Finally, global experiences of major deregulated electricity market are introduced one by one including the old England and Wales Pool, the British Electricity Trading and Transmission Arrangements, the Nord Pool, the PJM interconnection and the California crisis.

Chapter 3 introduces and discusses different types of transmission congestion management schemes in deregulated electricity market. Firstly three pricing schemes including the uniform marginal price, the locational marginal price and the zonal price are illustrated. Then five transaction curtailments based on five different principles are introduced. Furthermore, two financial instruments: the contract for difference and the financial transmission right are described. Finally, a case study is included to test the performances of the uniform marginal price and the locational marginal price in a modified IEEE-14 bus test system.

Chapter 4 proposes a method using sensitivity index PTDF to re-dispatch generation when a transmission congestion occurs in deregulated electricity market. Firstly, the mechanism of the proposed re-dispatch method is introduced. Secondly the algorithm of the sensitivity factor PTDF is presented. Thirdly, the mathematic model of the proposed method including an objective function and three constraints, one of which includes the sensitivity factor PTDF are described. Finally, case studies in this chapter consist of four test systems including a modified IEEE 14-bus system, a modified IEEE 30-bus system, a modified IEEE 57-bus system and a modified IEEE 118-bus system. The proposed re-dispatch method with the PTDF is tested in the above four test systems and the test results are included.

Chapter 5 proposes another generation re-dispatch method using the bus impedance matrix sensitivity factor. Firstly the methodology of the proposed method and the algorithm of the sensitivity factor ZS are described. Then the mathematic model of the proposed method is given, which is similar with chapter 4, but the only difference is it uses the bus impedance matrix sensitivity factor as the sensitivity factor. Finally, the proposed method is tested in the same four systems as chapter 4. Result comparisons between the ZS-based method and the PTDF-based method are included.

Chapter 6 proposes a security cost allocation method with the sensitivity factor PTDF. Firstly some conventional security cost allocation schemes are reviewed. Then the mathematic model of the proposed security cost allocation method using the PTDF factor is given. Finally, in the case studies, under the same four test systems previously used in chapter 4 and 5 respectively, experimental results are presented to show distinctions of allocation results by selecting different locations of slack bus in the PTDF-based method and the comparison between the pro rata method and the PTDF-based method.

Chapter 7 proposes another security cost allocation method using the sensitivity factor ZS. Firstly, the mechanism and the mathematic model of the proposed method are described. Then again the same four test systems used in chapter 4, 5 and 6 respectively are used to recur the tests, which provide a briefly comparison among three security allocation methods: the pro rata based method, the PTDF-based method with the first generator bus as the slack bus and the ZS-based method using the same slack bus.

Chapter 8 gives a summary of the thesis and the possible future research directions.

#### 1.5. Publication

- Jiawei Zhao, Jianfeng Lu and K.L.Lo, "Variously worldwide types of deregulated electricity markets and their respective transmission congestion management schemes", in 51st International Universities' Power Engineering Conference (UPEC), Coimbra, Portugal, 2016.
- Jiawei Zhao, Jianfeng Lu and K.L.Lo, "Review of methods to calculate congestion cost allocation in deregulated electricity market", in 6th World Congress on Engineering and Technology (CET): 6th Power Engineering and Automation Conference (PEAM), Shanghai, China, 2016.
- Jiawei Zhao, Jianfeng Lu and K.L.Lo, "A transmission congestion cost allocation method in bilateral trading electricity market", in 9th Asia-Pacific Power and Energy Engineering Conference (APPEEC), Chengdu, China, 2017.
- Jiawei Zhao and K.L.Lo, "Two sensitivity-based methods to re-dispatch generation in the deregulated electricity market", under preparation for journal submission of the IET Journals Generation Transmission & Distribution.

### **Chapter 2**

### **Deregulated Electricity Market**

#### 2.1. Introduction

In modern life, electricity is a necessity. With the high speed development of human civilization, electricity capacity allocation has become a very essential topic for power system research. Several requirements came along for power energy industry such as reducing the cost, improving efficiency and maintaining the long-term development [1]. As a result, since 1980s, many countries in the world started to unbundle their vertically integrated power utility into a competitive market environment because of the development of energy industry and demand of customers. It is believed that with the introduction of competition, liberalized electricity market will theoretically be able to satisfy the above requirements thus electricity price could get close to the marginal cost of generation, companies will minimize their production cost and customers could get a cheaper price for electricity [9]. The restruction in other industries such as railway, telecommunication and wireless communication has obtained a series of successes which also stimulates the progress of electricity industry deregulation. However, deregulation of electricity market is not simple, for example, electricity is unable to be stored easily so supply and demand must be balanced all the time. As a consequence, there are issues in the deregulation route both from technical side and economic side. The objective of energy industry restructure is to create a competitive environment for electricity trading in order to improve social welfare [25]. In fact, the motivations for electricity reform are always complex and political. In most cases, restructure progress will be more or less influenced by party politics [26]. Introducing competition into electricity market represents that government is willing to set up a sound and orderly energy system which could be utilized to stabilize power system operation, reduce the cost of power generation, decrease the
electricity price to customers, promote high technology and maintain the long-term development of electric power system.

One of the earliest deregulated electricity markets in the world was set up by the United Kingdom in 1990, called the "England & Wales Pool" [2]. It has been regarded as a remarkable template for the worldwide electricity deregulation because it broke the traditional vertically integrated structure and introduced competition into the generation side. Ten years later, "English Pool" was replaced by the New Electricity Trading Arrangement (NETA), which was implemented in England and Wales and was based on bilateral principles to let generation side and demand side trade power by contracts [3]. Five years later, with the joining of Scotland electricity, NETA became the British Electricity Trading and Transmission Arrangements (BETTA) [4]. In Europe, there is another successful electricity market called "Nord Pool" which consists of several Nordic electricity markets [5]. It was the first multinational electricity market in the world. In North America, Americans also learnt to set up their own competitive electricity market. Different states adopted different approaches to manage their power system operation. One of the most famous electricity markets is the PJM interconnection which is the largest deregulated wholesale electricity market in the world dominates power systems in Pennsylvania, New Jersey, Maryland and other states in the eastern part of United States [6]. Unfortunately, all the deregulated electricity markets have to face challenges. For example, California Market, in the summer of 2001, a series of rolling blackouts were caused by electricity market crisis [27].

In this chapter, history of deregulated electricity market will be introduced including the contents such as traditional integrated electricity market in section 2.2 and motivation of reform old power utility in section 2.3. Also, categories of current electricity market models will be briefly explained in section 2.4 mainly at pool market, bilateral market and hybrid market. The role of independent system operator will be discussed in

section 2.5. An important part of this chapter is section 2.6 which includes the detailed contents about the type of deregulated market each country adopted. Finally, section 2.7 will give the summary of this chapter.

#### 2.2. Traditional Vertically Integrated Electricity Utility

In many countries, prior to deregulation, generation, transmission and distribution were dominated by one large central utility which was mostly a monopoly and state-owned by government [28]. Figure 2.1 shows the structure of vertically integrated electricity utility [9].



Figure 2.1 Vertically Integrated Utility Structure

These central companies have authority to design and control the whole operation of power system in their designated regions. Consumers in such region only have one choice of power supply. The traditional vertically integrated utility holds all the information and data of power system operation and they dispatch and re-dispatch generation units by optimal power flow software to match supply and demand in order to keep power balance. In the vertically integrated mode, the setting of electricity price should be based on the overall cost of generation, transmission and distribution and price is uniform in the controlled region. Not only setting the price, the centralized company also controls the power network planning such as long-term generation investment and transmission distribution network maintenance. It is claimed that state-owned electricity utility is not as good as private ownership in the domain of efficiency improvement since state-owned companies is willing to focus more on budget maximization than costs minimization [29]. Hence, large-scale investment always leads to waste and inefficiency rather than cost reduction and better service. The risk of large investment will be placed onto consumers' shoulders. Due to lack of competition in power market, there is a high possibility that consumers would be faced with quite a high price of power supply. To sum up, electricity industry really needs to be restructured and deregulated so as to improve the efficiency and decrease the electricity price closer to marginal price.

#### **2.3.** Generally Understanding of Electricity Industry Deregulation

In the early 1980s, industry restructuring, deregulation, liberalization and privatization as a combinational concept became an irresistible trend spreading all round the world. Many nationalized industries have been successfully deregulated and some achievements have been obtained so it was suggested that electricity industry should also be restructured, deregulated and liberalized therefore competitive trading can be introduced. Chile was the first country in the world to implement electricity sector reform which began in 1982 [30]. Then the consequence is, UK, United States and Nordic countries were involved in the electricity industry deregulation in 1990s to open their electricity markets.

The purpose of electricity industry deregulation is that the traditional vertically integrated industry should be unbundled into generation, transmission and distribution sections and in each section competition will exist. Figure 2.2 shows the structure of

unbundled electricity industry [3]. As a result, both buyers and sellers in electricity market can benefit from this utility structure renewal because electricity price would be pushed close to the marginal cost. Moreover, with the market opening, plenty of incentives will be created so that investments from society could ensure long-term development on power industry.



Figure 2.2 Structure of Unbundled Power Industry [3]

It is not a wise idea to set up a totally new structure of electricity market because it will require large amount of time and finance. As a consequence, the best thing to do is to break the former monopoly structure and introduce competition into basic components. After deregulation, the former monopoly generation company is divided into several smaller generation companies which start to compete with each other, in the meantime, some new established generation firms are able to have a chance to join in the competition. Distribution section follows the same rule for restructure. The dominated distribution company is replaced by many independent ones. However, due to inflexible feature of distribution hardware, a distribution company in a specific region operates in a monopolistic characteristic. A distribution company can assign a right to a supplier to use the distribution network or a distribution company itself can play the role of supplier for customers' power usage. With the naturally monopolistic feature, transmission network is unable to be split up into several regional small transmission companies. Since competition cannot economically sustain, transmission network utility still stay at monopoly status. In addition to this, electricity market deregulation causes the growth of transmission congestion inescapability. An independent agency called System Operator is set up to control and supervise the operation of transmission system. In some cases, Transmission Company also undertakes the job of system operator. The independence of system operator points out an effective approach to balance the conflicts between economic interests and transmission security [31].

### 2.4. Difference between Previous and Existing Deregulated Electricity Market Structures

It is well known that there is no effective way to store large quantity electricity efficiently and cheaply, moreover, and hence supply and demand must be balanced all the time. As a consequence, competitive market for electricity should be distinguished from any other deregulated utility markets. Due to the trend of deregulation, restructuring and privatization, competitive electricity marketplaces are expected to be established in the way of breaking the linkage of former monopoly market structure not rebuilding a whole new electricity market [32]. The most significant issue of electricity market reform is how to effectively arrange power transaction at the meantime ensure the security of system operation. In fact, there is no such simple solution that can solve diverse problems and issues once and for all because different nations have discriminating political rules and various market situations. However, each type of electricity market deregulation should always follow one principle that generation, transmission and distribution/supply will be split up to let competition to be introduced into market environment so that electricity sellers and buyers could be benefited from the structure reform. Even though the goal is identical, routes for that by different nations appear not exactly the same. Those leading

countries of electricity reform have already obtained their own successful achievement for competitive marketplace establishment in some degree. Each competitive structure has its own feature and characteristic. Typically, there are three templated reformed electricity market structures which are pool market, bilateral market and hybrid market. It is very necessary to understand similarity and distinction between each deregulated electricity marketplace structure. Each of them has both strength and shortage and it will be a wise way to use the strong points to offset the weaknesses.

#### 2.4.1. Pool Market

The pool concept is regarded as the first innovative step which has a historical significance for electricity market restructure. In this type of deregulated marketplace, the independent system operator is called pool operator who coordinates all the power supply and manages the whole process of power trading [33]. All the generation companies need to upload their planned generation output information to the system operator normally one-day ahead. At the meantime, generation companies should submit their electricity bid price to the pool operator respectively. Then system operator starts to forecast the total electricity demand quantity on the date of generation uploaded. Based on generation cost minimization principle, pool operator sets up a list to determine which generation companies should generate electricity onto the power transmission network system. The list must be ranked to follow an increasing order from cheapest bid price to a higher price until the forecasted demand can be satisfied. The pool operator will set a series of pool prices by software calculation based on the received bids from generators and it has the authority to explain the details of these price settings. Finally generators will be paid by a uniform buying price and customers will be charged by a uniform selling price.

Technically, pool markets can be distinguished into two groups: mandatory pool and voluntary pool. In mandatory pool, all the power energy trades must be transacted in the pool; Singapore, Australia, and the former UK Pool are three typical examples of mandatory mode [9]. While, in voluntary pool, electricity trades are allowed to be made outside the pool system. For example, the Nord pool could be considered as a typical voluntary pool market since it allows market participants make power transaction outside the pool market.

At the beginning, as the pioneer attempt of electricity utility reform, pool market has achieved some effective successes since its similarly centralized market structure allowing former regulated electricity industry to adapt to it in a short time. Nevertheless, pool structure still has two main disadvantages. Firstly, if there are several dominate generation companies in the market, they probably attempt to increase their economic profits through strategic biddings which are also considered as exercising a market power. In general, two approaches could be utilized to exercise market power: bidding high directly and withholding capacity during peak time. As a result, electricity price in the pool market will probably not decrease with the drop of generation cost so that customers are unable to benefit from industry deregulation [40]. Secondly, pool structure market only concentrates on the competition in generation end while demand side is paid much less attention [3]. Due to such above issues, a new mode of energy trading is expected to be introduced to overcome the problems which bothers the market participants in pool market.

#### 2.4.2. Bilateral Market

In an ideal bilateral marketplace, any power consumer can purchase electricity from any electricity generation company who can also sell electricity to any customer. Power trading could be done through bilateral contracts individually by buyer and seller.

By now, in real life, BETTA (British Electricity Trading and Transmission Arrangement) is a practical example of bilateral electricity market [3]. This time independent system operator only needs to concentrate on matching the real-time power balance and maintaining system operation security. Electricity price is bilaterally dealt by

generator and customer without the influence of the system operator. Contract could be signed in advance even for several years or as short as even for the last 24 hours before real delivery action. One hour before market gate closure, all the required information and data from both contractual sides should be handed in to the system operator such as power transaction quantities, transfer time and electricity offer/bid prices and such group of information is referred as the final physical notifications (FPNs). More details of the FPNs will be given in the following section. If all the power trades are accepted by the system operator, market participants who have contracts only need to follow the contractual details to pay and charge. But if any of the transactions caused transmission security problem, the system operator has the power to reject the energy trade and ask the market participants to adjust their contractual positions. The key point in bilateral market will be the imbalance settlement solution which takes place after the gate closure. System operator will make use of the above balancing mechanism to secure the power system operator [34].

In bilateral market, competition is directly introduced into the demand side which means demand-side customers can obtain more available price information and have more choices for their power source supply. With the increasing of market liquidity, the prices of electricity have been reduced which thanks to the bilateral mechanism. However, this advanced market scheme also has drawbacks. In order to avoid the penalty in balancing actions, some generators may intend to increase their power production and some suppliers may try to decrease their demand quantity. In this way, difficulty of power balance will be increased and the whole cost for power system operation will be increased. If so, imbalance settlement and penalty mechanisms will definitely be useless and the efficiency of bilateral market will no doubt drop down.

#### 2.4.3. Hybrid Market

In the Hybrid market, participants process their electricity transaction inside the pool market and they can also arrange contracts with other market participants outside the pool market. Hybrid market can be regarded as the combination of pool market and bilateral market. At the moment, there are two typically successful hybrid markets among the global power markets: the Nordic market and the PJM (Pennsylvania, New Jersey, and Maryland) market.

Take Nordic market as an example, it is a multinational electricity market consists of electricity markets from Nordic and Baltic countries. Each of them has an independent system operator named differently but do the same job. Market participants can conduct trades not only in the counter market, but also in the financial market of Nord pool. In the financial market, participants sign contracts for electricity trading so as to reduce the price in the future years. Nordic system also provides physical-delivery market in order to effectively control and manage real-time power balance. Indeed, multiple liberalized electricity markets help decreased the possibility of exercising market power by dominated participants. However, due to power transmission network constraints, exercising market power behaviour can never be totally avoided.

As with the Nordic market, the PJM market also allows market participants make transaction both inside and outside the pool-type market. In fact, there are four markets exist in PJM market system which are a capacity market, an energy market, a FTR (Financial Transmission Right) auction market and an ancillary service market. Moreover, the energy market in PJM can be divided into two markets: a day-ahead market and a realtime energy balance market. Market participants can not only purchase and sell electricity in the above market they can also feel free to make bilateral contacts with each other to complete power trades.

#### 2.5. Independent System Operator

After liberalisation of electricity utility, the control of the operation of the transmission system has been separated from the control of the operation of generation plants. In order to open access for market participants to the monopoly transmission system, a stand-alone independent system operator (ISO) has been created which has been regarded as the "soul of the power grid" and the "air traffic controller" of the power system [35]. The US Federal Energy Regulatory Commission (FERC) has outlined a set of characteristics and functions for the ISO which are described as follows [36]:

#### Characteristics:

- 1. Independence
- 2. Scope and Regional Configuration
- 3. Operational Authority
- 4. Short-term Reliability

#### Functions:

- 1. Tariff Administration and Design
- 2. Congestion Management
- 3. Parallel Path Flow
- 4. Ancillary Service
- 5. Total Transmission Capability and Available Transmission Capability
- 6. Market Monitoring
- 7. Planning and Expansion
- 8. Inter-regional Coordination

The ISO is an independent system operator who carry out generation schedule in electricity market [37]. By definition, all the independent system operators should undertake the task of controlling the whole operation of electricity system. However, not all the system operators perform all the above functions. They might peel off the operation of energy markets from the whole operation of the system and let a third party or their subsidiary to manage them [35]. The basic work of the system operator is to secure and manage the transmission network. In fact, ISOs in different electricity market models will perform different functions. For instance, in pool market, ISO should be responsible for energy trading, generation scheduling and prices setting. But in bilateral market, the ISO has nothing to do with energy market and the responsibility is narrowed to purely manage real-time system balance and provide ancillary service [9].

#### 2.6. Global Experience of Deregulated Electricity Market

Since the concept of electricity market deregulation has been put forward around the world for more or less thirty years, more and more nations on their own have gained achievements on electricity industry restructure in various degrees. Some of those countries even successfully established sets of complete market systems in order to make full use of electricity market mechanism. In those reformed markets, rules and orders have been clearly formulated and performed. Market participants only need to follow the instruction and then to enjoy the benefits from competition mechanism. Electricity industry would obtain a long term development and improvement under new utility structures. Nevertheless, particularly several deregulated markets did not smoothly go through the tough process of reform. During the process, they suffered a series of issues, troubles and even failures. Following parts of this chapter will introduce several typical world experience of competitive electricity markets which are from UK, Nordic countries and US. 2.6.1. UK

#### 2.6.1.1. The England & Wales Pool

Before privatization, electricity industry in England and Wales was dominated by a vertically integrated nationalised company called Central Electricity Generating Board (CEGB) which was a classic example of a cost-of-service regulated public utility [38]. On 31<sup>st</sup> March 1990, electricity industry in England and Wales started to be restructured with new mechanism [39]. The CEGB with its 75 power stations and one national grid had been split into three generating companies: National Power with 60% of conventional generating capacity (40 power stations with 30 GW capacity), PowerGen with 40% of conventional generating capacity (23 power stations with 20 GW) and Nuclear Electric (12 nuclear stations with 2 GW); and one transmission company: the National Grid Company with the high voltage grid and together with 2 GW of pumped-storage generation [40]. Twelve former area distribution boards had been replaced by 12 regional electricity companies (RECs) which were regarded as local distribution systems [1]. An electricity pool market had been established to arrange and manage electricity trades in England and Wales with wholesale market mechanism. The establishment of electricity pool in England & Wales has been treated as a remarkable innovation in electricity privatization process. The electricity pool mechanism was utilized to break the vertical linkage between generation and transmission and to introduce competition into electricity industry which requires all the generators to sell their electricity in a wholesale market, the pool. Figure 2.3 shows the structure of electricity industry at privatization [3].



Figure 2.3 the Structure of Electricity Industry at Privatization in England & Wales

The England & Wales Pool was a typical mandatory pool market which means every energy trade has to be transacted only inside the pool market. For the generation section, those three main generating companies competed with each other by using generation capacity and biddings. For the transmission section, the National Grid Company was not only just a transmission company who hold transmission network ownership but also an independent system operator who operated the pool market and managed the pool's settlement system on behalf of pool market participants. For the distribution section, 12 private-owned distribution/supply companies accounted for distributing and retailing electricity to customers in different regional areas. It seemed like the former electricity vertically integration was not change a much compared with the pool structure. However, the establishment of pool mechanism had a remarkable significance in electricity industry deregulation history because competition had been brought in so competitive bidding process has been created and facilitated.

In the England & Wales Pool market, by 10 a.m. each day, generation companies should send their information to the system operator including generation output quantity and bid price of electricity provided on the following day [41]. One day ahead, the independent system operator, here is the National Grid Company, already forecasted the power demand quantity in each half hour period [42]. After collected the bid information and output data from generation companies, the system operator will start to accept bids from cheapest price to more expensive price until the forecasted demand is satisfied. Then, the system operator will sort out a bid list which contains names of the generation companies who have been chosen to generate electricity on the following day [43]. Those chosen generation companies were called "in merit" generation companies which means their bids have been accepted by system operator and those whose bids have not been accepted by system operator were called "out of merit" generation companies [44]. Then the independent system operator will utilize the software called "Generating Ordering and Loading" (GOAL) to draw up an unconstrained generation schedule which follows a least cost generation principle [45]. The last job for the system operator is setting prices for all the pool market participants. As known, System Marginal Price (SMP) is defined as the price which is bidden by the most expensive generation company in the bid list [40]. In the pool market, the system operator will pay the Pool Purchase Price (PPP) to all the in merit generation companies. The Pool Purchase Price (PPP) is the System Marginal Price (SMP) plus Capacity Payment (CP) which is defined with Loss of Load Probability (LOLP) and Value of Loss Load (VOLL) [40]. On the other end of pool structure, the suppliers who is willing to purchasing the electricity power from the pool will pay for the price called Pool Selling Price (PSP). The Pool Selling Price (PSP) is the Pool Purchase Price (PPP) plus the Uplifts [1]. The equations which defined the relationships between those prices are shown as following [9]:

$$PPP = SMP + CP \tag{2.1}$$

$$CP = LOLP \times max(0, VOLL - SMP)$$
(2.2)

$$PSP = PPP + Uplifts \tag{2.3}$$

$$Uplifts = \frac{security \ constrained \ cost - unconstrained \ cost}{total \ load \ demand}$$
(2.4)

CP is actually an incentive which is used to reward the generation companies who declare that their capacity is available regardless of whether they are required to guarantee or not. LOLP is the probability that electricity power capacity is unable to support the actual demand. LOLP is predicted in each half hour. VOLL is the price customers are going to pay to avoid electricity power shortage [46]. When expected electricity capacity shortage occurs, CP will climb up and when predicted electricity generation capacity exceeds, CP will drop down. Neglecting power losses and ancillary service, the Uplift can be regarded as the cost of congestion relief. It is claimed that PPP is definitely an ex-ante mechanism. The Uplift is utilized to cover the cost of system services which are important to power system security. Several aspects are taken consideration into Uplift such as transmission constraints, demand forecast errors, ancillary service and other services. The amount of the Uplift will be informed by the end of each trading day so that the Uplift will be regarded as ex-post mechanism. In the pool market, electricity price will change in every 30 minutes depending on generation, demand and generation companies bidding behaviours. Hence, electricity price is not stable. In order to hedge against the price volatility, power sellers and buyers sign a bilateral contract called Contract for Differences (CfD) outside the pool market [47].

Overall, the England & Wales Pool was successful during the period from 1990 to 2001. Since the pool structure was much similar with the former vertically integrated utility, new operation mechanisms and principles were able to easily adapt to the pool system. Operation of power system during this period was almost secure and stable. Nevertheless, the original intention of ideal deregulation was not realized. The aim of introducing competition into electricity industry was to suppress electricity price for

customers but the real electricity price in pool market did not drop following with the reduction of generation cost. The electricity price's drop was not dramatic in the pool market and customers did not benefit from the liberalization of electricity industry. The electricity price in the pool was unable to truly reflect the generation marginal cost. The reason was noticed that in the pool market several dominant generation companies can easily exercise the market power by strategic bidding which will affect the efficiency of the pool market. When the forecasted electricity price is going down, dominant generation companies will use strategic bidding to raise the electricity price in the pool. Another drawback of the pool market is that demand side was lack of competition. In other words, the pool market is the market with inadequate representation of the demand side [40]. Electricity suppliers were all paid at the same pool price rather than on negotiated prices, which inhibited supply side price pressure. Moreover, a centrally forecast estimate of demand limited involvement from the demand side within the pool market. The complexity, opacity and the lack of competition in price setting inhibited the development of derivatives markets and reduced liquidity in the contracts markets. This leads to high margins on the financial contracts struck between generators and suppliers, therefore further raising prices to customers above those that would have prevailed with more competitive arrangements [48]. Electricity customers can only choose the regionally local power supplier for electricity supply and the above phenomenon is unable to satisfy the intention of restructure of electricity market.

#### 2.6.1.2. British Electricity Trading and Transmission Arrangements (BETTA)

The England & Wales Pool had almost successfully operated for around 10 years but its performance was unable to catch up with the steps for further requirement of liberalization of electricity industry. Then the aim is to establish a power industry with whole new mechanisms which would realize the electricity free trading as far as possible like any other commodities [48]. Thus on 27<sup>th</sup> March 2001, in England and Wales, the New Electricity Trading Arrangements (NETA) were introduced to replace the pool market with the concept that market participants have rights to transact electrical power by bilateral trading [49]. On 1<sup>st</sup> April 2005, with the joining of Scottish power system network, the New Electricity Trading Arrangements (NETA) has been extended and reformed as the British Electricity Trading and Transmission Arrangements (BETTA) [50]. The structure and operational principles of new system BETTA have only small difference when compared to NETA. The BETTA can be regarded as the extension of the NETA [51]. The only difference is the BETTA covers the whole British-wide electricity industry which means all the market participants including companies from Scotland can join in this deregulated market. The vertical structure was similar with the former pool structure. However, as the independent system operator, the function of the National Grid Company (NGC) has been changed. Figure 2.4 will show the industry structure of BETTA [3]:



Figure 2.4 the Industry Structure of the BETTA

At this time, NGC has two roles: one is the transmission asset owner and another one is the independent system operator [3]. As the asset owner of transmission network, the NGC should ensure the maintenance, investment and long-term development of transmission system. As the system operator, the NGC needs to keep supply-demand balance all the time and to maintain the real-time security of network operation [52]. BETTA operates the power system differently from the former pool market. Figure 2.5 expresses the diagrammatic representation of BETTA time frame process [53]:



Figure 2.5 the Diagrammatic Representation of BETTA Time Frame Process

In the BETTA, electricity sellers and buyers can transact power trading by bilateral contracts. The bulk of electricity power is traded by Forward/Future contracts and short-term power exchange. The Forward/Future contracts could be signed ahead of days, months or even several years before the actual electric power delivery. The forward contract market is where customers and suppliers can have contracts for electricity directly with generators. In this bilateral market, market participants negotiate with each other directly and agree to trade a quantity of electricity for physical delivery at an agreed location and over a period of time. The future contract market allows derivatives trades in

a regulated market. These are binding agreements between a buyer and a seller to deliver and take delivery on a specified future date a quantity of electricity at a price agreed today [3]. The short-term power exchange is within 24 hours before electricity delivery which offers market participants an opportunity to 'fine adjust' their contract details by a convenient way [48]. To help the system operator access to the power balance condition, market participants are required to notify their predicted physical information in each half hour period, for power seller is the planned generation output and for power buyer is the metered demand, by 11 a.m. one day ahead. This information is called initial physical notifications (IPNs) [3]. Before the gate closes, usually one hour before power delivery which was 3.5 hours prior to 12 June 2002, market participants must submit their final physical notifications (FPNs) to the system operator [54]. During the period between gate closure and real time, SO will run the Balancing Mechanism and match the balance between supply and demand so that system security could be guaranteed. Participants have an opportunity to voluntarily submit their offers and bids to let the SO know how much they are willing to pay or charge for the power quantity difference compared with their FPNs [48]. An offer is a participant's price proposal to increase generation or decrease demand and a bid is a participant's price proposal to decrease generation or increase demand. Offers and bids are requested to submit in pairs and each market participant is allowed to submit maximum of ten pairs [55]. If transmission constraint occurs, imbalance settlement will be done by the SO mandatorily and that participants whose offers have been accepted will be paid at System Sell Price (SSP) while ones whose bids have been accepted will be charged at System Buy Price (SBP) [9]. SSP and SBP are two imbalance prices which are calculated by the SO based on the collected offers and bids [55]. The original contracts position of market participants will be honored [4]. Balancing service that NGC provides is consisted of two types of balancing actions: energy imbalance actions and system imbalance actions [53]. The transmission constraint management is attached to the system imbalance actions. The SO recovers all the cost of balancing service through a Balancing Service Use of System (BSUoS) charge to all the market participants [53]. BSUoS charge is calculated in each half-hour period which divides a whole day into 48 equal intervals. The calculation of transmission congestion cost is

included inside the BSUoS charge calculation [56].

Unlike the former pool structure with system operator centrally dispatching, generators in the BETTA is able to self-dispatch according to the contracts with consumers. Moreover, demand-side in the BETTA is involved into competition mechanism so that suppliers can compete with each other. Some large customers became more active roles in the market. Since more price information is indicated, market liquidity has been improved during the new structure process. By new mechanism of the BETTA, wholesale prices and retail prices have an obvious reduction. However, the BETTA is not perfectly effective due to its dual imbalance settlement prices. Generation companies might deliberately increase electricity output to make their metered output volume larger than contracted volume so that they can escape from the penalty of the SBP or even obtain the reward from the SSP. The suppliers might deliberately lower their predicted demand value shorter than the contracted demand to get rid of the punishment from SBP or even get the advantage from the SSP. Because of above issues, generators operate below the optimum efficiency which will result in primary fuel waste and emission increase and suppliers shorten their predicted contract volume will cause the over-contract market formation hence power system reserve will always stay at a high level which will lead to the whole system operation cost increase. Another disadvantage of the BETTA is the unfairness to renewable power generation especially the wind power. All the market participants are required to have such ability of output accurately prediction, power supply guarantee and short notice flexibility. As known, intermittent is the weakest point of wind energy supply. Unable to predict the future output, wind power is the most likely to face with the penalty by imbalance settlement.

#### 2.6.2. Nordic Countries

On 1<sup>st</sup> Jan 1996, Sweden and Norway joined together and established the Norwegian-Swedish Exchange called as Nord pool ASA which is the initial state of the current Nord pool [57]. Then in 1998, Finland market joined in. Two years later in 2000,

Demark market joined in. In 2002, Nord pool spot AS referred as a separated company organized the Nord pool's spot market. Since then, a new integrated regulatory framework for electricity industry competition across Nordic countries was established which was the first truly multinational electricity market. In 2010, Estonia opened its electricity market and joined in. Then in 2012, Lithuania following the Estonia opened the power market and joined in the pool. In 2013, Latvia opened electricity market and joined in. In 2016, Nord Pool Spot is rebranded to Nord Pool [58]. Since those countries worked together to handle with the power system operation, independent system operators from different countries need to communicate and cooperate very closely. The Nord Pool is owned by the Nordic and Baltic transmission system operators (TSOs) which are Statnett SF (Norway) with 28.2% stakes, Svenska Kraftnät (Sweden) with 28.2% stakes, Fingrid (Finland) with 18.8% stakes and Energinet.dk (Denmark) with 18.8% stakes, Elering (Estonia), Litgrid (Lithuania), AST (Latvia) hold 2% stakes each of them [59]. In addition to owning and operating the high-voltage grid, TSOs are responsible for the security of supply in their countries [60].

The Nord Pool market consists of physical and financial markets which means market participants could not only trade electricity through spot market but also via bilateral contract. According to Nord Pool mechanism, Elspot market is referred as the main physical market which is a day-ahead market arranging hourly bilateral contracts of electricity power delivery [61]. Elspot market will determine the hourly spot price which will be regarded as the reference price both for physical market and financial market [62]. In 2002, approximately 30% of the electricity consumption was traded in the Nord pool and in 2017, this number has been increased to 80%. In 2002, the number of direct market participants in Elspot market was reached at two hundred including generation companies, suppliers and large customers and in 2017, number of companies involved in Nord pool was up to 380 [61, 63]. Another physical market Elbas was introduced as a supplement of Elspot market for the additional balancing services. Elbas as the intraday market offers each market participants an opportunity to adjust their physical power position such as the price and volume of supply or demand one-hour before power delivery [62]. The day-

ahead Elspot market is considered as the spot market and the intraday market Elbas plays the role of balancing mechanism in the Nord pool electricity market. For the financial section, before 2002, there were two financial markets in the Nord Pool: Eltermin and Eloption [61]. Power sellers and buyers could sign a bilateral contract in Eltermin market days, weeks, months and even up to four years ahead for hedging against the price volatility risk. Eloption market was a relatively new financial market compared with Eltermin market. It also provides contract services for market participants to prevent the electricity price risk. Since 2008, Nord pool financial market started to be organized by NASDAQ OMX Commodities exchange which is the dominant marketplace for system price contracts [64]. Besides those financial markets above, the Nord Pool also allows market participants to organise the over the counter (OTC) trading and Contracts for Differences (CfDs) to make bilateral contracts with each other. Figure 2.6 shows the diagrammatic representation of Nord pool time frame process [65].



Figure 2.6 the Diagrammatic Representation of Nord Pool Time Frame Process

In total, Nord Pool structure is quite a successful deregulated electricity market model among the whole global electricity markets. There are four main reasons for Nord Pool success. Firstly, market design is completed and mature. Secondly, significant quantity of hydropower resource enables support of the not great demand in Nordic area. Thirdly, the multiple integrated power system effectively dilutes the market power. Finally, Nord Pool obtained strong support from political side. Even though having above advantages, Nord Pool still has several threats which will affect the normal operation of Nord Pool market. One is power system faults and constraints cannot be completely avoided so that power system operation cannot be fully secured. Then the behavior of exercising market power cannot be stopped so long as competition mechanism develops. The last threat is that hydropower is highly dependent on climatic condition which will easily have influence on electric power supply.

2.6.3. US

#### 2.6.3.1. PJM Interconnection

On 1<sup>st</sup> April 1997, PJM opened its first bid-based energy market and during that year PJM was approved as the first national fully functioning independent system operator (ISO) by the Federal Energy Regulatory Commission (FERC). On that time, PJM operated the transmission network but did not own it in order to open the access of transmission system to non-utility users. Then to develop the advanced competitive wholesale electricity market and multi-areas operation, FERC encouraged the formation of the regional transmission organization (RTO). Hence, PJM became the first national functioning RTO in 2002 [66]. Right now, PJM Interconnection is a regional transmission organization (RTO) in the United States which operates a wholesale electricity market and a power transmission network system which is part of the Eastern Interconnection grid. The PJM market is serving all or part of areas of Delaware, Illinois, Indiana, Kentucky, Maryland, Michigan, New Jersey, North Carolina, Ohio, Pennsylvania, Tennessee, Virginia, West Virginia and the District of Columbia. The head office of PJM is located in

Valley Forge, Pennsylvania [67]. Currently, PJM is operating a competitive electricity wholesale market and a high-voltage grid together serving 65 million customers. So far, PJM market has already 1032 participants with more than 178,563 megawatts of generating capacity, 165,492 megawatts peak demand and 135,252 kilometres (84,042 miles) of transmission line. In 2017, PJM has totally delivered 773,522 gigawatt-hours of power energy [68].

PJM is composed of an energy market which consists of a day-ahead market and a real-time balancing market; a capacity market; a financial transmission right (FTR) auction market and an ancillary service market which consists of a regulation market and a reserve market [69, 70, 71, 72]. In this section, only energy market in the PJM will be discussed. Figure 2.7 displays the tree structure of the PJM market:



Figure 2.7 Tree structure of PJM markets

The day-ahead market is a forward market which will calculate the hourly clearing prices for each hour of the next operating day. These prices are calculated based on

generation companies' offers, demand side bids, virtual supply offers, virtual demand bids and bilateral transaction schedules and those data will be submitted into the day-ahead market by market participants with voluntarily basis [6]. The balancing market is a realtime market which will calculate the clearing price in every 5 minutes period with the principle of actual system operation and security constrained economic dispatch [73]. Different settlement methods are adopted in two types of markets. The day-ahead market settlement is based on the planned hourly power quantities and day-ahead hourly electricity prices. The balancing market settlement is according to the hourly quantity deviations from planned values and real-time hourly electricity prices. Both hourly prices calculation is based on the Locational Marginal Pricing (LMP) [74].

The day-ahead market offers market participants an opportunity to trade electricity power at binding day-head prices. Transmission customers can schedule bilateral transactions at binding day-ahead congestion price which is based on source and sink price difference calculated with LMPs. In the day-ahead market, generation companies who have joined in an installed capacity contract have obligation to hand in their offer schedules even if they are self-scheduled and during outage. Customers should hand in their bids to the system operator to notify whether they are willing to pay congestion charges, or they are willing to be curtailed if transmission congestion occurs in the realtime market. After the day-ahead market bid period closure, the system operator starts to make the dispatch schedule for each hour of the next operating day. The real-time energy market is based on the real-time operation conditions. The generation rebidding period from 4 p.m. to 6 p.m. is introduced for the generation companies who are not selected into day-ahead market schedules to alter their bids for use in the real-time energy market otherwise their original day-ahead bids will come into force in the real-time market [58]. Generation companies who produce more than their scheduled quantity in the day-ahead market will be paid by the real-time LMPs. For customers who consume the power volume more than the planned quantities in day-ahead market will be charged by the real-time LMPs. If market participants who are trading in spot market, their trades will be settled at the real-time LMPs. Besides the day-ahead market and the real-time balancing market, PJM also provides financial instruments such as Financial Transmission Rights (FTRs) for energy sellers and buyers to hedge the prices variation risks.

Overall, the PJM market shows that it is quite a complete and mature competitive power market system after development of so many years. In the PJM, all the market participants have obtained an opportunity with lock in day-ahead prices for financial schedule and power trading. Of course under the functioning market principles, market liquidity and reliability have been dramatically promoted up to a whole new level. Effectively motivation has been transferred to market participants so that they are willing to take part in the day-ahead schedule submission. The PJM also provides incentives to help dispatch the real-time resources. In the PJM market, strengthened financial instrument such as the financial transmission right (FTR) is very effective for prices risk hedging. The long-term investment and development of power system can be ensured by the good market environment [73].

#### 2.6.3.2. California Crisis

The development of deregulated competitive electricity market was not always perfectly successful sometimes even terrible. In 1992, the California Public Utilities Commission (CPUC) planned to restructure the current electricity market which is regarded as the basis of California Assembly Bill AB1890 [27]. The crisis happened in California in the beginning of 21<sup>st</sup> century would be the tough memory during the electricity market deregulation history. On one hand, the crisis was regarded as the results of multiple contributed factors including a drought that reduced the level of hydroelectric power, unexpected outages at nuclear power plants, high natural gas prices, and strong demand for power energy. On the other hand, it was because of the truly mismanagement and wrong rules establishment of market deregulation process.

Wholesale market establishment is the beginning of market deregulation so several wholesale marketplaces were created for this purpose. In order to manage and control these newborn wholesale markets, two new institutions, the Power Exchange (PX) and the California Independent System Operator (CAISO) were set up [75]. However, above two organizations were operated separately rather than one integrated utility which will lead to market inefficiency and market power manipulation. Moreover, retail price control which was created in California market cut off the relationship between consumers and power producers. It is very strange that California has controlled retail electricity prices but volatile wholesale electricity prices [76]. The negative point of applying retail price control is that cost changes in wholesale market cannot be passed to consumers end. In the California market, participants have to sell and buy electricity power only through PX and CAISO and the advance power purchase time is only maximum one day-ahead before power delivery. The current Economists believed that with the higher wholesale electricity prices, incentives will push the investors who are more likely willing to build a groups of new generation plants [27]. Nevertheless, construction of generation plant takes time and especially in California more time is needed for generation plants licensing process. In fact, the failure has been proved that few new generation was brought into the market but only a mass of applications submitted for new generation plants in 1999 and 2000 [27]. In addition, a sudden demand increasing occurred during that period without any prediction [76]. It is indicated that from 1999 to 2000 the power demand growth rate was 4% more than previous years [27]. At that time, due to climatic influence as lack of rainfall, except conventional thermal power, other new generation such as nuclear, gas-fired were already working at full capacity. Worse still, when the winter came, many generation plants needed to be shut down for repair and maintenance. On June of 2000, long-term potential problems finally caused crisis among California area. The electricity price in California market on April 2000 was around 30 dollars per megawatt hour and the price for June increased to the price interval from 250 to 450 dollars per megawatt hour [77]. The first five months of 2001 was full of high wholesale electricity prices, energy emergencies and rolling blackouts. Even worse, on December 2000, the Federal Energy Regulatory Commission (FERC) replaced the retail price control by a soft cap in wholesale market.

The further facts proved that the soft price cap was not only useless but also instigated market participants gaming behaviours which was that generation companies exported electricity from California and then reimported the power at an increased price. Lacking power supply will result in high wholesale prices combining with retail price gap will force power utilities to sell off most of their generation plants and drive retailers quit selling their products. In short, it was a vicious circle that supply shortage caused wholesale price rises that lead supply to further decrease. On June 2001, after government took over, wholesale prices dropped under 50 dollars per megawatt hour, demand declined and new generation plants got in [77]. By such above phenomenon appearance, California crisis had been finally proved was over.

The lessons from California Crisis can be obtained as follows. Firstly, connection and communication between power producers and power suppliers must be ensured. Long-term bilaterally contractual system is significantly necessary in the deregulated electricity market. Any change from wholesale level should have the channel to pass to the retail level. Secondly, strong support and appropriate intervention from governmental side are also regarded as the contributed factor which will make sure the secure and stable operation of electricity market operation.

#### 2.7. Summary

This chapter introduced the contents about the deregulated electricity market. The traditional vertically integrated electricity utility has been presented firstly and coming with the drawbacks which are also the motivation of electricity industry restructure. Then generally understanding of electricity industry deregulation has been described with the history of worldwide electricity deregulation actions, the planned new structure of electricity industry and future expectation of electricity utility deregulation. Three types of deregulated electricity market structure models have been introduced: a pool market model, a bilateral market model and a hybrid market model which can be referred as the combination of the former two models. Independent system operator as the key role of

electricity market deregulation has been recommended with its unique characteristics and important functions. Global experience of deregulated electricity market is the main content of this chapter. The description of the UK deregulated electricity market is the priority among priorities. The old England and Wales Pool as the one of the pioneers achieved a significant step in history of electricity market reform. The British Electricity Trading and Transmission Arrangements as the replacer of pool market came further close to the original intention of competition introduction. Then Nord Pool as the largest multinational electricity market has been discussed with its special features. The US also has two interesting electricity market models but one is quite successful and another one gave us some experimental results. PJM interconnection is definitely a good deregulated electricity market but California crisis reminded that electricity industry reform is a tough and complicated project.

## **Chapter 3**

# Transmission Congestion Management in Deregulated Electricity Market

#### **3.1.** Introduction

In the modern power system, transmission congestion becomes a major issue which could cause power system crisis such as cascade outages, increase in losses, electricity price rises and threats to power system security [78]. Transmission congestion is referred to the situation that operating constraints of transmission line are violated so the line transfer capacity is exceeded. Consequently, power flow going through the transmission line is restricted by its capability [79]. Transportation of electricity is constrained by several physical laws including thermal constraints which is also called overloading, voltage constraints and stability constraints [1, 35]. Congestion is also caused by unexpected events such as generation outages, load demand increase and power system equipment failure or damage [79]. In the competitive environment of deregulated electricity market, congestion occurs generally during the unconstrained dispatch stage when the system operator is not able to accommodate all the desired transactions based on the least cost principle without violating transmission system constraints [78]. At this step, the system operator must take actions to prevent actual congestion happening in the real power delivery. In the old traditional vertically integrated structure, congestion management is simple because system operator only needs to re-dispatch the generation schedule [9]. However, with the deregulation becoming mature, congestion management became more and more complex and frequent, since transmission network is on open access with non-discrimination to all the market participants. Therefore, power flows are determined by power trading rather than controlled by an integrated utility. In fact, the direct way to resolve transmission congestion is to construct new transmission lines to provide adequate power transfer capacity. However, new line construction takes a long time and the investment of transmission network cost are huge. Additionally, there are some technical methods for congestion relieving such as transformer taps, phase shifting transformers and flexible AC transmission system devices (FACTs) [11]. But the above approaches have limited ability on transmission congestion relief. By contrast, market-based methods for congestion management have brighter future because the current congestion is mainly caused by electricity market deregulation. Market-based methods aim to resolve congestion by using the existing transmission network in an economic way which improve the efficiency of the whole power system utilization. It is a better way to go against the congestion problem with market mechanism.

In this chapter, the key content is the congestion management schemes in market environment. Each of them will be illustrated in detail. Section 3.2 includes several major market-based congestion management schemes which are utilized in various electricity market types. Section 3.2.1 explains the pricing schemes for congestion management. In pool and hybrid market, uniform marginal price in section 3.2.1.1, locational marginal price in section 3.2.1.2 and zonal price in section 3.2.1.3 are introduced in details. In bilateral market, transaction curtailments in section 3.2.2 are utilized to relieve transmission congestion. Section 3.2.3 illustrates two financial instruments: contract for difference (CfD) and financial transmission right (FTR) which have been created and developed to hedge the impact of transmission congestion and to stabilize electricity price level. Section 3.3 is a case study of a modified IEEE-14-bus system which is used for a comparison between uniform marginal price and locational marginal price. Finally, section 3.4 is the summary.

#### **3.2.** Market-based Transmission Congestion Management Schemes

Power flow delivery should follow Kirchhoff's laws and a transmission line has its physical limit in aspects of heat, voltage and power angle. If market participants only focus on economic interests maximization rather than transmission network physical limitation, issues of transmission constraints especially transmission congestion will occur inevitably. Hence, in a competitive electricity market, an independent system operator is required to control and manage the power system operation. The independent organization will not be influenced by any other market participant and its function is to balance the conflict between economic interests and power system operation security. A big challenge for the system operator is to keep system operation on a stable and reliable level, whilst at the same time, to allow market participants equal access to the transmission capacity. To accomplish such challenge, power system managers are required to establish a complete system which has specific and reasonable schemes for transmission congestion management.

There are three typical pricing schemes: uniform price, locational marginal price and zonal price. Transaction curtailment is also utilized for congestion management. Besides, several financial approaches are also applied to transmission congestion management. Congestion management methods are categorized as follows in Table 3.1 [9]:

#### **Table 3.1 Congestion Management Schemes in Different Market Structures**



#### 3.2.1. Pricing Schemes

#### **3.2.1.1.** Uniform Marginal Price

As its name implies, the uniform marginal price indicates that electricity trading prices in the market will be identical for all the market participants without considering the location of power source or power sink. The pricing scheme was implemented in the former England & Wales Pool market. The pool system operator adopted this pricing scheme and is referred as "re-dispatch first, compensate later" to respond to the situation when transmission congestion happens. There are two steps the system operator needs to act: firstly unconstrained dispatch and then secondly security-constrained re-dispatch if the congestion has been triggered. In the market dispatch stage, the system operator will collect all the bids from generation companies and rank the bids from cheapest to higher prices until customer demand can be met. The highest price in the bid list will be set as the market clearing price which is also called the system marginal price (SMP) and the generator who bids the most expensive price will become the marginal generator [80]. When transmission congestion is found after the first dispatch, the system operator will launch the second stage: security-constrained re-dispatch. The system operator will reschedule the generator list to remove the possibility of congestion, and at the same time keep the security cost at the minimum.

The algorithm of the first stage market dispatch will be showed as follows:

$$\min \sum_{i=1}^{N_G} C_{G_i}(P_{G_i})$$
(3.1)

Subject to:

$$\sum_{i=1}^{N_G} P_{G_i} = \sum_{j=1}^{N_L} P_{L_j}$$
(3.2)

$$0 \le P_{G_i} \le P_{G_i}^{max} \tag{3.3}$$

Where

$C_{G_i}(P_{G_i})$	: The bid-based generation cost of generator <i>i</i>
$P_{G_i}$	: The active power generated by generator $i$
$P_{L_j}$	: The demand prediction of load <i>j</i>
$N_G$	: The total number of generators
$N_L$	: The total number of loads
$P_{G_i}^{max}$	: The maximum generation capacity of generator <i>i</i>

Equation (3.1) expresses the objective function of the first stage dispatch which is subject to two constraints equation (3.2) which states the power balance between generation and demand and equation (3.3) which states the generation output of each generator should not be beyond the maximum generation capacity. At this stage, the system operator does not take transmission constraints and losses into consideration but only focuses on generation cost minimization. If the first dispatch does not trigger the transmission constraints, the initial generation schedule will be implemented.

Once the market dispatch violates the transmission constraints, an inequality constraint will be added to the dispatch algorithm by the system operator to become a security-constrained dispatch. The additional constraint is expressed as follow:

$$P_{line_k} \le P_{line_k}^{max} \tag{3.4}$$

Where

 $P_{line_k}$ : The power flow on the transmission line k

 $P_{line_k}^{max}$ : The maximum power flow transmission capacity of the transmission line k

Equation (3.4) is the constraint which requires all the transmission lines in the redispatch stage should be operated within their power transmission capacities. A new schedule for re-dispatch generation will be decided by the new algorithm and the generator list is updated because some generators will be required to reduce power generation or even removed from the initial list due to transmission line constraints. The generators which were out-of-merit generators but now are on the new generation list on the second stage are called "constrained on" generators and those who were in-merit generators but now are ordered to generate lower output than their initial plans on the new schedule are called "constrained off" generators [81]. At the time of settlement, all the generation companies who participated in the power delivery are paid by the Pool Purchase Price (PPP) which is mentioned in the section 2.6.1.1 and also compensated by an adjustment payment for the difference between their bid prices and the PPP. The adjustment payments for different type of generators are expressed as follows [9]:

Adjustment for constrained off generators:

 $Adjustment_{off} = (unconstrained amount - constrained amount)$ (3.5)  $\times (PPP - bid \ price_{off})$  Where

*bid price<sub>off</sub>* : The bid price of the "constrained off" generator submitted to the system operator during the unconstrained dispatch

Adjustment for constrained on generators:

 $Adjustment_{on} = (constrained amount - unconstrained amount)$ (3.6)  $\times (bid \ price_{on} - PPP)$ 

Where

*bid price*<sub>on</sub> : The bid price of the "constrained on" generator submitted to the system operator during the unconstrained dispatch

The uniform marginal price scheme is regarded as a simple approach for congestion management because there is only one price set in the whole pool market so the system operator can easily calculate and allocate the cost of congestion. However, there are several shortages indicating that such a pricing scheme needs to be improved or even replaced. The process of dispatch and re-dispatch are not transparent since all the economic and physical information is collected by the system operator so that market participants have to only obey the generation schedule in which the trading price information notified to market participants but it cannot reflect the congestion cost. The congestion cost is allocated based on an equal sharing principle which is unable to provide locational distinction between participants.
#### **3.2.1.2.** Locational Marginal Price

In 1982, MIT Professor Fred Schweppe proposed an hourly spot price in energy marketplace which is regarded as the origin of the locational marginal price (LMP) or nodal price [82, 83]. The definition of the locational marginal price is the minimal marginal cost of supplying the next increment of 1 megawatt hour power at a specific bus [13]. If there is no transmission congestion and transmission losses in the system, the locational marginal price of each node will be the same which is also set as the system marginal price based on the collected bids from market participants. However, in reality, transmission congestion and losses will exist during the system operation. Once congestion or power losses happens, locational marginal price between different nodes will become distinct due to different costs of power supply in each node. At the moment of settlement, generator will be paid based on the LMP at its node and customer will pay for the LMP at its node. Both actions are implemented by the independent system operator who has already run an optimal power flow to calculate the LMP of each node. Locational marginal price is utilized as the primary pricing scheme in the electricity markets of Argentina, Chile, Ireland, New Zealand, Russia, Singapore and several states in the US for electricity price calculation and transmission congestion management because of its capability of providing accurate economic signals by which market operation becomes more efficient and long-term power system investment becomes more attractive [84].

LMP is calculated based on the optimal power flow (OPF) which is the integration of the power flow calculation with the minimization of an economic objective function subject to the equality and inequality constraints of system network operation [85]. Before the calculation of LMP, the system operator will make use of OPF to calculate megawatt dispatch of each generator. There are two types of OPF model including the DC optimal power flow (DCOPF) calculation and the AC optimal power flow (ACOPF) calculation. The ACOPF is built upon an AC power flow which considers both active and reactive power flow at the same time. The ACOPF is known for the accurate simulation results which can be directly used as the benchmark to represent the real power flow. However, because of non-linear characteristics, ACOPF calculation is always complex to simulate large scale power system, requiring large execution time and the high probability of nonconvergence [86]. Compared to ACOPF, DCOPF is simpler and more convenient approach since it is linear and only considers active power flow neglecting voltage, reactive power and transmission loss [87]. DCOPF is often used for the system operator to dispatch generator and calculate LMP because of its simplicity, robustness and higher speed of convergence [86]. The algorithm of DCOPF is illustrated as follows:

In order to adapt to conventional high-voltage transmission system, in DCOPF modeling some assumptions should be formed as follows [88]:

- 1. Voltage angle difference  $\delta_{ij} = \delta_i \delta_j$  between neighboring nodes is small so that  $sin_{\delta_{ij}} = \delta_{ij}$  and  $cos_{\delta_{ij}} \approx 1$
- 2. The voltage profile is flat and the value of p.u. is equal to 1 which means that the voltage amplitude is equal for all nodes
- Transmission line resistance is negligible so that DCOPF can be considered as linear modeling

The DCOPF has the objective function which is to minimize the generation cost with several constraints such as power flow balance, generation limits and line flow limits. Assume there is a power system with N nodes and L lines, the objective function and constraints are shown as follow [89]:

$$min f = \sum_{i=1}^{N} C_i(P_i^g)$$
 (3.7)

Where

$$C_i(P_i^g) = \frac{1}{2}C_{2i}(P_i^g)^2 + C_{1i}P_i^g + C_{0i}$$
(3.8)

Subject to:

$$P_i^g - P_i^d - \sum_{i=1}^N P_{ij} = 0$$
(3.9)

$$P_{i,min}^g < P_i^g < P_{i,max}^g \tag{3.10}$$

$$|P_l| < P_l^{max} \tag{3.11}$$

Where

$P_i^g$	: The real power generated at node <i>i</i>
$C_{0i}$	: The no-load cost coefficient of generator <i>i</i>
$C_{1i}$	: The linear cost coefficient of generator <i>i</i>
$C_{2i}$	: The quadratic cost coefficient of generator <i>i</i>
$P_i^d$	: The real power demand at node <i>i</i>
P <sub>ij</sub>	: The power transmission flow from node <i>i</i> to node <i>j</i>
P <sub>i,min</sub>	: The minimum generation limitation of node <i>i</i>
P <sub>i,max</sub>	: The maximum generation limitation of node <i>i</i>
$P_l$	: The real transmission power flow on the line <i>l</i>
$P_l^{max}$	: The real power maximum transmission capacity of line $l$

Equation (3.7) represents the objective function of the DCOPF which is to minimize the generation cost. Equation (3.8) indicates the bid-based cost function of the generator at node i. Equation (3.9) shows the real power balance constraint at node i.

Equation (3.10) expresses the maximum and minimum output constraints of the node i. Equation (3.11) indicates the capacity constraint of the transmission line l.

For the DCOPF, the calculation of locational marginal pricing is straightforward if line losses are not included. However, the power loss in a power system network may not be negligible so many markets are moving towards marginal loss pricing. As the DCOPF is lossless by definition, it is required to develop a suitable methodology to represent line losses within the DCOPF [86]. The power loss modelling with DCOPF can be found in [87, 90]. It is considered as an equivalent formulation that will not change the optimal solution of the DCOPF. A reduced DCOPF model with losses can be derived with the replacement of real power balance constraint equation at bus *i* by the total real power balance equation considering power loss among all nodes in the system. As known, power injected into all the nodes is equal to power withdrawn out of all the nodes plus transmission losses. Combined with the above concept, equation (3.9) can be rewritten into such new equation as follow [90]:

$$-\sum_{i=1}^{N} P_i^g + \sum_{i=1}^{N} P_i^d + P_{loss} = 0$$
(3.12)

Where

 $P_{loss}$  : The transmission losses in the power system

Based on the equation (3.12), the corresponding Lagrangian equation can be defined as follows [14, 91]:

$$\mathcal{L} = \sum_{i=1}^{N} C_{i} \left( P_{i}^{g} \right) + \lambda_{0} \left( -\sum_{i=1}^{N} P_{i}^{g} + \sum_{i=1}^{N} P_{i}^{d} + P_{loss} \right) + \sum_{l=1}^{L} \mu_{l} \left( |P_{l}| - P_{l}^{max} \right) + \sum_{i=1}^{N} \pi_{i}^{max} \left( P_{i}^{g} - P_{i,max}^{g} \right) + \sum_{i=1}^{N} \pi_{i}^{min} \left( P_{i,min}^{g} - P_{i}^{g} \right)$$
(3.13)

Where

- $\lambda_0$  : The Lagrangian multiplier of the whole system power balance constraint
- N : The total number of nodes
- L : The total number of transmission lines
- $\mu_l$  : The Lagrangian multiplier of transmission line constraint
- $\pi_i^{min}$ : The Lagrangian multipliers of minimum generation capacity of generator *i*
- $\pi_i^{max}$ : The Lagrangian multipliers of maximum generation capacity of generator i

Based on the above Lagrangian equation, equation of the LMP of node *i* can be obtained as follows:

$$LMP_{i} = \frac{\partial \mathcal{L}}{\partial P_{i}^{d}} = \lambda_{0} \left( 1 + \frac{\partial P_{loss}}{\partial P_{i}^{d}} \right) + \sum_{l=1}^{L} \mu_{l} \times \frac{\partial P_{l}}{\partial P_{i}^{d}}$$

$$= \lambda_{0} + \lambda_{0} \times \left( \frac{\partial P_{loss}}{\partial P_{i}^{d}} \right) + \sum_{l=1}^{L} \mu_{l} \times T_{i-1}$$
(3.14)

Where

 $T_{i-1}$  : The sensitivity factor for real power at node *i* with line *l* constraint

The LMP in node *i* can be divided into three components as follows [16]:

$$LMP_{i} = LMP_{i}^{energy} + LMP_{i}^{loss} + LMP_{i}^{congestion}$$
(3.15)

Correspond with equation (3.15), equation of  $LMP_i^{energy}$ ,  $LMP_i^{loss}$  and  $LMP_i^{congestion}$  can be obtained as follows:

$$LMP_i^{energy} = \lambda_0 \tag{3.16}$$

$$LMP_i^{loss} = \lambda_0 \times (\frac{\partial P_{loss}}{\partial P_i^d})$$
(3.17)

$$LMP_{i}^{congestion} = \sum_{l=1}^{L} \mu_{l} \times T_{i-1}$$
(3.18)

Where

 $LMP_i^{energy}$ : The system marginal cost of node *i*  $LMP_i^{congestion}$ : The cost of transmission congestion of node *i*  $LMP_i^{loss}$ : The cost of transmission losses of node *i* 

The LMP based electricity market is designed to enhance power system reliability, provide market price certainty and increase electricity market liquidity [92]. By LMP based pricing approach, economic signals are indicated to market participants so that the exercising market power is effectively prevented. The reason is that if a generator bids higher than the marginal cost, the chance of being selected for dispatch is reduced and the generator might lose the opportunity to generate. If a generator bids lower than the marginal cost, it may not receive sufficient revenue to cover its generation cost [93]. The

influence of transmission congestion and losses are reflected through the variation of LMP in nodes so that the electricity market is transparent. For long-term viewing, the LMP gives effective incentives for investment in generation and transmission assets. Nevertheless, LMP is not a perfect approach which is capable of solving all the problems in power systems. Since bids of generators submitted to the system operator are bid-based rather than cost-based, they still have the chance to enact gaming behaviors. Moreover, it is argued that the LMP only reflect short-term purchase price which is actually the spot market price expectation rather than the long-term average cost of generation. Consequently, the price signals provided by LMP are seemed as subjective, unreliable and arbitrary [94]. The other concern about LMP is that transmission congestion and losses circumstances would result in higher and more volatile nodal prices, even though LMP can be effective and workable, greater revenue collecting from ISO will cause inefficiency for economic operation of electricity market [95]. The last defect of the LMP method is that since power system networks are always big in size, the corresponding LMP specific to each node is complex which requires plenty of time and large degree of coordination.

#### 3.2.1.3. Zonal Price

To deal with the last drawback of the nodal price scheme, some electricity markets are bundling nodes locating in neighboring regions together into a zone in which the electricity price is uniform, this can be regarded as the marginal generation cost of the zone. The pricing scheme which is used to simplify the complex process of the location marginal price is called the zonal price. In many countries, zonal price is adopted as the major approach for electricity pricing and congestion management. Take Nordic countries as a typical example, as mentioned in section 2.6.2 the Nord pool market consists of multiple electricity markets of Nordic and Baltic countries. The local system operator in each country divides the whole nation area into several bidding zones. The number of Norwegian bidding zones nowadays is five. Denmark consists of two bidding zones: eastern Denmark and western Denmark. Finland, Estonia, Lithuania and Latvia constitute one bidding area for each of them. Since 2011, Sweden has four bidding zones [96]. The

system operator splits the whole power system into zones mainly according to the historical data and the economic operation of each node. When there are no transmission constraints, all the electricity prices in each node are identical to the system marginal price which is calculated based on the bids collected from market participants [97]. Once transmission constraints occur, the whole system will be divided into different zones which were defined in advance. Inside each zone, the electricity price is uniform which is calculated according to the bids collected from participants in such single zone. Prices between zones are distinct due to the transmission constraints, for example, congestion, while the constraints inside each zone is considered as infrequent and insignificant [98]. The whole process of zonal split is illustrated in following diagrams Figure 3.1 and Figure 3.2:



Figure 3.1 Meshed network structure of a multi-nodes power system without congestion



Figure 3.2 Multi-nodes power system split into three zones when congestion occurs

From Figure 3.1 and Figure 3.2, it is noted that there is a nine-node power system which is split into three zones once transmission congestion occurs. As predefined, node 1, 2, 3, node 4, 5, 6 and node 7, 8, 9 are composed into three single zones respectively. The transmission line between node 2 and node 4 now becomes the only connection between zone 1 and zone 2 is named as "Tie-line" [9]. In a similar way, the transmission line between node 2 and node 7 and the transmission line between node 6 and node 9 are also set as Tie-lines which are responsible for the connection between zones. It can be observed that zonal price is the combination of the uniform marginal price and the locational marginal price. Inside the zone area, all the nodal prices are identical which is similar with the principle of the uniform marginal price while each different zone can be regarded as a bigger node with its own electricity price which follows the rules of the locational marginal price.

After zone partition is congested, the difference between each zone will appear on the generation capacity and zonal price. For example, assume zone 1 in Figure 3.2 has surplus generation capacity therefore the zonal price in zone 1 will be lower than other zones. Thus zone 1 will be set as the low price zone. If there is a generation shortage occurs in zone 2 then the zonal price in such area is higher than others. Thus zone 2 will be set as the high price zone. Under this circumstance, the system operator will try to figure out an effective way to balance such distinction between those two zones. As a consequence, the Tie-line between zone 1 and zone 2 will be utilized to transport the surplus electric power from zone 1 to zone 2 as long as the Tie-line is always operating under its maximum transmission capacity. Due to this action by the system operator, extra electricity from zone 1 can be sold to zone 2 at a higher price and customers in zone 2 can be benefit from the lower price electricity from zone 1. Same with the principle of the LMP, the system operator will charge the congestion cost from the customers who used the transmission network. The collected revenue due to congestion will be used for power system network investment.

In some cases, the zonal price is regarded as a simpler and lower cost approach for congestion management compared with the locational marginal price. However, the zonal pricing method cannot be simply considered as a simplification of nodal pricing since in real power system it is more complex than theory. Zonal price supposes the intracongestion between nodes inside the zone is infrequent so that the cost of intra-congestion is too small to count. In fact, the intra-zonal congestion is frequent and the corresponding congestion cost is higher than theory [99]. Because price inside one zone is uniform for all nodes while in reality each node has different generation cost so the zonal price is unable to reflect the real cost of generation. As a result, market customers have the burden of unfair costs once intra-congestion occurs. Moreover, it is quite difficult and complicated to define a new zone. Since the boundary of a zone is defined under the assumption that inside a zone there is a low frequency of intra-congestions but the opposite occurs in reality the setting of a new zone usually lags behind the intra-zonal congestion management. The zonal price was considered as a perfect approach for congestion management for a time. However, more electricity markets are transitioning their pricing scheme from the zonal price to the locational marginal price [3].

#### 3.2.2. Transaction Curtailment

In an ideal bilateral electricity market, the power transaction is arranged by market participants themselves who do not have the obligation to submit their contract information to the system operator. However, the unique characteristic of electricity market is that power system has a series of system constraints which must not be violated. To ensure the security of power system operation, sometimes the system operator is forced to curtail the already signed transactions between participants in order to reduce the power flow on the congested transmission branches. Therefore, in bilateral electricity markets, participants still need to report the physical information of their contracts to the system operator and take the risk of transaction rejection in order to give assistance to system operation security [100]. In addition to this, power transactions are signed days, months, or even years ahead of the actual power delivery, transmission network conditions might change during that time so that the initially secure transaction may endanger the system security criterion. Due to above circumstances, the system operator should formulate the schemes of transaction curtailment to maintain the power system under a safe operational condition. By superposition theorem, the power flow in a specific transmission line can be regarded as the sum of power flow from each transaction contribution. So the system operator can relieve an overloading transmission line by power transaction curtailment which is divided into following four types [35, 83, 101, 102]:

#### 3.2.2.1. First Come First Serve

This rule of transaction curtailment is regarded as a simple and straightforward one. Once transmission congestion occurs, the system operator will firstly curtail the transaction which was the last submitted. If the congestion still exists, the system operator will keep rejecting transactions following a reverse order of transaction submission until the congestion is relieved completely. Even though the method is simple, it may not be effective sometimes. The main reason is the system operator curtails the transactions without considering the power flow contribution of each transaction to the congested transmission line. The last submitted transaction might potentially provide the counterflow on the congested transmission line while curtailing it arbitrarily may worsen the congestion problem.

#### **3.2.2.2. Pro Rata (Proportional)**

By this approach, once transmission congestion occurs on a transmission line, the system operator will curtail transactions to relieve the line overload based on the corresponding power flow contribution of each transaction. Based on a linear DC optimal power flow, the sensitivity of each transaction to the power flow of the overloaded transmission line can be calculated relatively accurate so that the flow contribution of each transaction to the congested line can be also found out. The curtailed volume is proportional to the transaction contribution. The power transaction which has a higher contribution to the congested line will be curtailed more power and vice versa. Different from the former method, this method takes the counter-flow into consideration. The transactions that causes counter-flow to the overloaded transmission line is regarded to have zero contribution so it will not be curtailed by the system operator.

#### 3.2.2.3. Minimum-net Curtailment

The only difference between this method and the former one is that this method aims to restrict the totally curtailed net power volume to the minimum. Since different transactions have different power flow sensitivities, curtailment for one certain transaction which has greater contribution to the congested line is more effective than any other transaction curtailments. The system operator focuses on searching such transaction which contributes congestion more than others.

#### **3.2.2.4.** Compensative price

In this approach, if the transaction curtailment is unavoidable, the market participant whose transaction is rejected will receive a compensation payment from the system operator. Each participant is required to submit a compensative price to the system operator in advance to claim at what price they are willing to accept for the action of curtailment. Once the congestion happens, the system operator will start to curtail the transaction with the lowest compensative price until the overloaded line declines to its limit capacity. The final target for the system operator is to minimize the compensation cost while solving the transmission congestion problem. This transaction curtailment scheme requires the system operator to have income so as to compensate the market participants with their curtailed contracts. If the system operator is a nonprofit party, the compensation cost will be allocated to market participants.

#### 3.2.2.5. Willingness to Pay

In this method, the system operator will curtail any power trade if it is necessary to relieve transmission congestion but without paying any compensation to the market participants. For the market participants who does not want their transactions to be rejected have an opportunity to bid a willingness to pay price to the system operator in order to avoid being involved in the curtailment action. If congestion occurs, during the curtailment action, the system operator will firstly curtail the transactions from the participant who did not pay for the willingness to pay price. If all the participants have submitted their willingness to pay prices, the system operator will start to curtail the transaction from the participant who bid at the lowest price until the congestion is removed. Different from the compensation price method, this method will increase income for the system operator. If the system operator is also the owner of the transmission network, this part of income will be utilized for transmission assets maintenance and reinforcement.

#### 3.2.3. Financial Instruments

With the reform of electricity industry, the price of electricity is determined by a series of market conducts rather than controlled by an integrated utility. No matter in what type of electricity market model, the price volatility caused by the transmission congestion is the problem that market participants do not want to face with. For the generators, low price of electricity will raise the difficulty for them to cover the generation cost such as capital investment cost, maintenance cost and use of system charges. For the suppliers, unstable electricity price will have influence on their profit making. For the demand-side consumers, the rapid growth of electricity price will increase their production cost or life cost. Hence, some financial instruments are created to provide market participants a market-based route to hedge the impact of electricity price instability. There are two famous financial instruments: the Contract for Difference and the Financial Transmission Rights.

#### **3.2.3.1.** Contract for Difference

Contract for Difference is a bilateral agreement between market participants to hedge the price volatility risk [103]. The concept of CfD was originally used in the early 1990s in London power market and utilized only in the financial field [104]. Then this financial method was introduced into the electricity market. CfD in the old England & Wales Pool is utilized to hedge the price variation between the pool purchase price and the contractual price. In the Nord pool market, CfD is aim to help power buyers and sellers to reduce the impact of the price difference between zonal price and the system marginal price [9]. The formula of CfD is defined as follows [103]:

$$CfD = Power_{contract} \times (price_{reference} - price_{strike})$$
(3.19)

Where

 $Power_{contract}$ : The power volume signed in the CfD contract  $price_{reference}$ : The reference price which is often the spot market price  $price_{strike}$ : The agreed fixed price signed in the CfD contract

Market spot price tends to be peaky and volatile so generation company and power supplier sign these over-the-counter contracts directly to hedge the price uncertainty. CfD provides a specific amount of electrical power at a fixed price called strike price which is target to compare with a reference price usually the market spot price. If the strike price is higher than the reference price, power buyers need to pay the price difference times the contractual power amount to the power sellers. If the strike price is lower than the reference price, the sellers should pay the difference to the buyers. The majority of CfD contracts in pool market aim to hedge against the pool purchase price volatility. If the pool price is higher than the contractual price, generator will pay for the CfD to supplier. If the pool price is lower than the contractual price, supplier will pay for the CfD to generator. Therefore, electricity price fluctuation can be hedged in the pool market.

#### 3.2.3.2. Financial Transmission Right

To address the financial problem due to transmission congestion, it is a common practice for the independent system operator in the United States electricity wholesale market to issue a financial transmission right (FTR). FTR was introduced in different wholesale markets at different times and named differently. For example, FTR was introduced in PJM market on April 1998 and named Fixed Transmission Right; in New York, FTR was introduced in September 1999 and referred as Transmission Congestion Contract (TCC); FTR was introduces in California since February 2000 and named as the Firm Transmission right; in New England, FTR was referred as the Financial Transmission Right and introduced on March 2003 [105]. Because the market outcome cannot be predicted accurately in advance, market participants are likely to be exposed to congestion charges which may raise the risk of power transaction [106]. FTR is a type of financial instrument which entitle the holders to obtain a compensation for transmission congestion cost when transmission congestion occurs in the day-ahead market. In other words, FTR is an ex ante transmission capacity allocation. The compensation is based on the locational marginal pricing scheme. FTR provides a right to hedge the nodal price distinction between the node of receipt which is also called as source node and the node of delivery which is also called as sink node. FTR gives holders a benefit when the day-ahead congestion happens in the desired direction while asking holders to pay the difference when the day-ahead congestion happens in the opposite direction [106]. Currently there are three types of FTR: right for an exclusive use of the transmission capacity, right to use the transmission capacity and right to collect the congestion rents [106]. The formula of FTR is defined as follow:

$$FTR = Power_{FTR} \times (price_{sink} - price_{source})$$
(3.20)

Where

<i>Power<sub>FTR</sub></i>	: The active power delivered from source node to sink node
price <sub>sink</sub>	: The nodal price of the sink node
price <sub>source</sub>	: The nodal price of the source node

Based on equation (3.20), FTR holders will receive a positive payment according to FTR if the sink LMP is higher than the source LMP. FTR holders will have to pay the liability according to FTR if the source LMP is higher than the sink LMP. Therefore, in short-term, by purchasing FTR, the risk of LMP variation for wholesale electricity participants will be reduced. In long-term, the aim of FTR is to give the right economic signals to market participants to stimulate transmission investments.

#### 3.3. Case Study

In this part of the chapter, the uniform marginal price and the locational marginal price are selected from the above mentioned congestion management schemes as two typical methods for a comparison by performing a case study. A modified IEEE-14 bus system is utilized as the base model to test and compare the performances of the UMP and LMP.

#### 3.3.1. The Modified IEEE-14 Bus System

A modified IEEE-14 bus system is chosen as the test system in this study as shown in Figure 3.3. The 14-bus system consists of 5 generators, 11 loads and 20 transmission lines. Assume all the generators have the same generation capacity. DC optimal power flow (DCOPF) is implemented by software Matpower (version 6.0) to calculate the active power flows [125]. The software Matpower is utilized due to its fast calculating speed and easy convergence, which is because when using DC network modelling, the standard OPF problem can be simplified to a quadratic program, with linear constraints and a quadratic cost function. Here the voltage magnitudes and reactive powers are eliminated from the problem completely and real power flows are modelled as linear functions of the voltage angles [126]. All transmission lines are assumed to be lossless so as to neglect the effect of power losses to active power flow so that transmission congestion is the only point that needs to be focused on. Reactive power source in the system is assumed sufficient hence all the bus voltages are fixed at one per unit. Number of congested transmission lines in this test is restricted to one only in order to reduce the complexity of the test. In the test environment, parameters of power system are set arbitrarily such as bid price of each generator, generation capacity of each generator, active power consumption of each load, and transfer capacity of each transmission line. After running an unconstrained dispatch without considering transmission constraints, it is found that power flow of transmission line 1-2 is over its transfer capacity. So here transmission congestion occurred on transmission line 1-2. The technical considerations such as ancillary service, price elasticity and reactive power effects are neglected.



Figure 3.3 The modified IEEE-14 bus system

Before the test, some parameters of test environment are set as follows:

Bus	Generator	Bid price (£/MWh)	Min(MW)	Max(MW)	Load(MW)
1	Gen1	20	0	100	0
2	Gen2	50	0	100	24
3	Gen3	45	0	100	25
4	-	-	-	-	26
5	-	-	-	-	25
6	Gen6	35	0	100	24
7	-	-	-	-	0
8	Gen8	30	0	100	0
9	-	-	-	-	26
10	-	-	-	-	25
11	-	-	-	-	26
12	-	-	-	-	25
13	-	-	-	-	24
14	-	-	-	-	25
Total					
load					275

Table 3.2 G	<b>Jeneration</b>	and	Load	Data
-------------	-------------------	-----	------	------

Branch	From bus	To bus	Limit(MW)
1	1	2	60
2	1	5	100
3	2	3	100
4	2	4	100
5	2	5	100
6	3	4	100
7	4	5	100
8	4	7	100
9	4	9	100
10	5	6	100
11	6	11	100
12	6	12	100
13	6	13	100
14	7	8	100
15	7	9	100
16	9	10	100
17	9	14	100
18	10	11	100
19	12	13	100
20	13	14	100

Table 5.5 Transmission Capacity of Each Transmission Lin	city of Each Transmission Lin	apacity	I ransmission	1 able 3.3
----------------------------------------------------------	-------------------------------	---------	---------------	------------

In Table 3.2, bid prices, maximum and minimum MW generation outputs of five generators, MW demand of each load and total load are set as above. In Table 3.3, transmission capacity of each transmission line is also set.

#### 3.3.2. Results

Assume the system operator needs to draw up a schedule of generation for a halfhour period. It has received the bid price of each generator in the power system. By implementing an unconstrained dispatch, the system operator found a congestion occurring on the transmission branch 1-2. As a result, it implements a security-constrained dispatch to satisfy the demand eliminating the transmission congestion. Using Matpower to simulate DCOPF, data of unconstrained dispatch and security-constrained dispatch is obtained as follows:

Generator	Bid price(£/MWh)	Unconstrained dispatch (MV)	Security-constrained dispatch(MW)
Gen1	20	100	83.54
Gen2	50	0	0
Gen3	45	0	0
Gen6	35	75	91.46
Gen8	30	100	100
Total (£/h)		7625	7871.91
Congestion cost			
(£/h)			246.91

Table 3.4 Generation	Cost of	Unconstrained	Dispatch	and	Security-Constrained
Dispatch					

Table	3.5	Power	Flows	of	Unconstrained	Dispatch	and	Security-Constrained
Dispat	ch							

From bus	To bus	Unconstrained Security-constrain		Limit(MW)
		dispatch flows(MW)	dispatch flows (MW)	
1	2	70.4	60	60
1	5	29.6	23.5	100
2	3	18.3	16.3	100
2	4	14.0	9.9	100
2	5	14.1	9.8	100
3	4	-6.7	-8.7	100
4	5	-0.5	-1.1	100
4	7	-24.4	-27.8	100
4	9	6.2	4.2	100
5	6	18.3	7.2	100
6	11	17.6	20.8	100
6	12	20.0	20.5	100
6	13	31.7	33.4	100
7	8	-100	-100	100
7	9	75.6	72.2	100
9	10	33.4	30.2	100
9	14	22.3	20.1	100
10	11	8.4	5.2	100
12	13	-5.0	-5.0	100
13	14	2.7	4.9	100

From Table 3.5, it is found that the unconstrained dispatch will cause the transmission line 1-2 congestion because power flow in that branch is 70.4 MW which exceeds the transmission capacity of 60 MW. After a security-constrained dispatch, the power flow of branch 1-2 has been reduced to the limit level. From Table 3.4, it is noted

that generator 1 is ordered to decease its power output while generator 6 is asked to increase its power output. The security-constrained dispatch resulted in a congestion cost of 246.91 £/h. The difference between uniform marginal price and locational marginal price for congestion management is not only on the pricing mechanism but also by way of congestion cost allocation. Assume the congestion cost in both pricing methods is allocated to all the loads. Hence, different performances between the two methods on congestion cost allocation are discovered and stated as follows.

#### **3.3.2.1.** Congestion Cost Allocation by Uniform Marginal Price

The UMP method allocating the congestion cost based on a uniform price which is calculated as follows:

Allocated Congestion cost per 
$$MW = \frac{Total \ congestion \ cost}{Total \ demand}$$
 (3.21)

So the allocation details is found in Table 3.6 as follows:

Load	Demand(MW)	Allocated congestion cost (£/MWh)	Allocated congestion cost (£/h)
L2	24	0.897855	21.54851
L3	25	0.897855	22.44636
L4	26	0.897855	23.34422
L5	25	0.897855	22.44636
L6	24	0.897855	21.54851
L9	26	0.897855	23.34422
L10	25	0.897855	22.44636
L11	26	0.897855	23.34422
L12	25	0.897855	22.44636
L13	24	0.897855	21.54851
L14	25	0.897855	22.44636
Total	275	0.897855	246.91

Table 3.6 Congestion Cost Allocation by Uniform Marginal Price

From Table 3.6, it is found that the allocation of congestion cost is based on a uniform price which is at 0.897855  $\pounds$ /MWh. Even though the final congestion cost allocation of each load is different but it is based on the same price to calculate.

#### **3.3.2.2.** Congestion Cost Allocation by Locational Marginal Price

This part of section will illustrate the congestion cost allocation by LMP method. As mentioned in section 3.2.1.2, locational marginal price is calculated based on the DC optimal power flow. In this case study, the LMP of each node is computed by the software Matpower which follows the DCOPF. Since the testing power system is assumed lossless, the LMPs in security-constrained dispatch only has two components compared with equation (3.15). Here are the system marginal cost component and the congestion component, which is indicated in following equation:

$$LMP_i = LMP_i^{energy} + LMP_i^{congestion}$$
(3.22)

# Table 3.7 Locational Marginal Prices in Unconstrained Dispatch and Security Constrained Dispatch

Bus	LMPs unconstrained (£/MWh)	LMPs security constrained (£/MWh)
1	35.0	20
2	35.0	39.98000457
3	35.0	37.79829949
4	35.0	35.91348352
5	35.0	34.55754368
6	35.0	35
7	35.0	35.67020379
8	35.0	32.9072493
9	35.0	35.53934477
10	35.0	35.44349301
11	35.0	35.22562028
12	35.0	35.04261905
13	35.0	35.07591992
14	35.0	35.33672379

From Table 3.7, it should be noted that in the unconstrained dispatch, the LMP of each node is uniform at  $35 \text{ }\pounds/\text{MWh}$  while in the security constrained dispatch, the LMP of each node differ especially the price at node 2 and node 3 whose nodal prices are obviously higher than others. After obtained the LMP of each node, the load charge in both unconstrained dispatch and security-constrained dispatch can be known. Then the congestion cost allocated to each load can be calculated by the following equation:

The allocated congestion cost in per MW can be obtained by the following equation:

allocated congestion cost of each load in per MW  

$$= \frac{allocated \ congestion \ cost \ of \ each \ load}{load \ demand}$$
(3.24)

Load	Demand(MW)	Load charge unconstrained(£/h)	Load charge security constrained (£/h)	Allocated congestion cost(£/h)	Allocated congestion cost (£/MWh)
L2	24	840	959.5	119.5	5.0
L3	25	875	945.0	70.0	2.8
L4	26	910	933.8	23.8	0.9
L5	25	875	863.9	-11.1	-0.4
L6	24	840	840	0	0
L9	26	910	924.0	14.0	0.5
L10	25	875	886.1	11.1	0.4
L11	26	910	915.9	5.9	0.2
L12	25	875	876.1	1.1	0.04
L13	24	840	841.9	1.8	0.08
L14	25	875	883.4	8.4	0.34
Total				246.9	

 Table 3.8 Congestion Cost Allocation by Locational Marginal Price

From Table 3.8, the allocated congestion cost and the allocated congestion cost in per MW of each load are obtained by LMP method. It is indicated that each load is allocated different congestion cost especially the node 2 and 3 whose congestion cost allocations are more than other loads.

#### 3.3.3. Comparison

Select the fourth column of Table 3.6 and fifth column of Table 3.8, a comparison in congestion cost allocation on each load between two methods is shown in Figure 3.4; Select the third column of Table 3.6 and the sixth column of Table 3.8, a comparison in congestion cost allocation in per MW between two methods is shown in Figure 3.5.



Figure 3.4 Comparison in congestion cost allocation



Figure 3.5 Comparison in congestion cost allocation in per MW

#### 3.3.4. Discussion

From Figure 3.4, it is indicated that LMP considers the location and power flow contribution of each load so that the allocated congestion cost at different nodes is distinct. The cost allocated to Load 2 is larger than others since the transmission congestion occurred on branch 1-2. It should be noted that the congestion cost of Load 5 is a negative value which demonstrates the load does not contribute to the constraint problem, instead, it helps relieving the congestion. From Figure 3.5, it is observed that congestion cost allocation by UMP is based on one non-discriminate price. It does not reflect contribution of each load to the transmission congestion which means every load in the market shares the congestion cost uniformly. To summarise, on one hand, LMP provides economic signals to inform market participants where the congestion occurs frequently so the new generators would know where to build their new power plants. On the other hand, LMP provides a relatively fair way to allocate the congestion cost since the participant who

contributes the congestion more is required to pay for the congestion relief at a higher price.

#### 3.4. Summary

In this chapter, various types of transmission congestion management schemes in electricity market environment are introduced and discussed. In pool market and hybrid market, three types of pricing schemes: uniform marginal price, locational marginal price and zonal price are illustrated in terms of history, mechanism, pros and cons. In bilateral market, five transaction curtailments based on different principles are described. Besides, there are two financial instruments including the contract for difference and the financial transmission right which assist the market participants to hedge the impact of electricity price volatility. Finally, a case study is used for testing the performance of uniform marginal price and locational marginal price in transmission congestion management. A modified IEEE-14 bus system is built for the test. Under same testing conditions, locational marginal price provides stronger economic signals for market participants to inform the congestion frequency and is a fairer rule for congestion cost allocation.

## **Chapter 4**

## **Re-dispatch** Generation in Deregulated Electricity Market with Power Transfer Distribution Factor

#### 4.1. Introduction

Due to electricity market deregulation, transmission network is open access to all the network participants. Nevertheless, the finite transmission capacity is unable to satisfy the profit maximization of all the market participants. The outcome is no doubt the transmission congestion. Re-dispatching generator output is a basic and straightforward way to solve transmission congestion. In this chapter, a generation re-dispatch method with a sensitivity index which is called power transfer distribution factor (PTDF) will be introduced. It reschedules the generation by determining the power flow contribution of each node. This chapter will describe the algorithm of this sensitivity index and the redispatch method mechanism which involves the sensitivity index. Four modified IEEE bus systems are used for testing the performance of this re-dispatch approach.

This chapter is structured as follows. Section 4.2 indicates the difference between conventional re-dispatch generation methods and the proposed re-dispatch method. Section 4.3 represents the algorithm of the sensitivity index PTDF. Then section 4.4 describes the mathematic model of the proposed method including the objective function and constraints which involve the sensitivity index. Section 4.5 is the case study which consists of test results from four modified IEEE bus systems including the IEEE-14 bus system, the IEEE-30 bus system, the IEEE-57 bus system and the IEEE-118 bus system. Section 4.6 is the summary.

### 4.2. Proposed Generation Re-dispatch Method in Deregulated Electricity Market

This section will state the difference and the connection between conventional redispatch methods and the proposed re-dispatch method which relates to the sensitivity index PTDF.

#### 4.2.1. Conventional Re-dispatch Generation in Electricity Market

In the traditional vertically integrated power system, congestion problem can be alleviated by re-dispatching the cheapest available generation capacity because the centralized utility holds all the information to exercise the power to manage the whole power system. However, generation rescheduling will cause the cheaper electricity not to be delivered to the desired load. Hence, the total generation cost will be increased. The extra cost which is called as the security cost is put on the shoulder of the centralized utility or shared amongst consumers.

In pool electricity market, the solution of transmission congestion is similar to the route in the traditional structure of power industry in which the problem is solved by redispatching generator output. But this time, the information of generation cost of each generator is bid-based rather than cost-based which means the system operator reschedule the generation based on a bid-based cost minimization [35]. The security cost due to the transmission congestion is allocated to the demand-side customers by uniform marginal price, locational marginal price or zonal price.

In a bilateral electricity market, the generation re-dispatching is not simple since the system operator is unable to access the price information of the bilateral contract so it has no way to economically control the generator output. The fast way to solve congestion problem is to curtail the bilateral transactions. This approach has been described in section 3.2.2 which introduced five types of transaction curtailment schemes: first come first serve, pro rata, minimum-net curtailment, compensative price and willingness to pay. An alternative way to achieve to relieve the congestion is to provide generators a chance to allow them to submit incremental or decremental bids which indicate the price they are willing to increase or decrease their specific output quantities. As mentioned in the section 2.6.1.2, in practical bilateral market BETTA, the incremental bid is referred as the offer and the decremental bid is referred as the bid. The system operator is allowed to redispatch generator outputs by using these bids as the judgment criteria when it faces the congestion problem. But this method is on a voluntary principle so it is dependent on the submitted bids from the generators. If there are few bids submitted by generators, the system operator may not have sufficient governable power resource to deal with the congestion problem.

#### 4.2.2. Adjustments in the Proposed Re-dispatch Scheme

The inspiration of the proposed re-dispatch method is from the alternative way of congestion management in bilateral market mentioned above. Because this method is more aligned with the market-based principle which allows market participants access to the competition of electricity market in an economic way. This method is also more acceptable to the original intension of electricity industry reform. Generators state their acceptable price for power output variation through a transparent channel to help the system operator reschedule the generation plan to relieve the inevitable congestion in an efficient way, in the meantime, with the minimal security cost.

To make the proposed method more feasible and more effective, some adjustments to the alternative method need to be put forward as follows:

- 1. In the alternative method, the incremental and decremental bid submission for generators is voluntary. But in the proposed method, this bid submission becomes compulsory for generators. The main reason is to guarantee the system operator has sufficient amount of adjustable power resource to solve the transmission congestion issue.
- 2. A generator is permitted to submit a number of pairs of incremental or decremental bids. But the total power volume of its incremental bids should be limited to its generation capacity minus its scheduled power output and the total power volume of its decremental bids should be limited to its scheduled power output.
- 3. The congestion cost or called security cost calculation of the proposed method is based on both costs from incremental bids and decremental bids. Once the congestion occurs, the system operator will accept the incremental and decremental bids from generators to adjust generation plant to relieve congestion. If an incremental bid from a generator is accepted by the system operator, the generator will increase its output to fulfill the promise. Then the generator will get an earning which is equal to the bid price times the incremental power volume. If a decremental bid from a generator is accepted by the system operator will decrease its output. Then the generator will get a compensation which is equal to the bid price times the total cost of incremental power plus the total cost of decremental power.
- 4. In the proposed method, the cost of congestion is also named as the security cost and should be allocated to all the loads. All the generators are required to submit their incremental and decremental bids which is to assist the system operator to resolve congestion problem. Consequently, generators have no reason to shoulder the

congestion cost. The congestion cost allocation will be done according to the power flow contribution of each load, which will be illustrated in the next few sections.

4.2.3. Illustration of the proposed re-dispatch generation scheme

Figure 4.1 shows a diagram to illustrate the proposed re-dispatch method.



Figure 4.1 Illustration of the proposed generation re-dispatch scheme

#### 4.3. The Algorithm of Power Transfer Distribution Factor

Power transfer distribution factor (PTDF) is considered as the sensitivity of power flow of a transmission line when there is a power injection change at a bus [107, 127]. In other words, it indicates the relationship between the active power flow change quantity of a particular transmission line from bus j to bus k and the active power injection change quantity at bus i [35]. PTDF can be defined as follow:

$$PTDF_i^{j-k} = \frac{\Delta P_{j-k}}{\Delta P_i} \tag{4.1}$$

Where

 $PTDF_i^{j-k}$ : The PTDF of the line *j-k* respect to bus *i*  $\Delta P_{j-k}$ : The active power flow variation of line *j-k*  $\Delta P_i$ : The active power injection variation at bus *i* 

To derive the expression of PTDF, DC power flow is introduced to simplify the decoupled power flow which comes from the Newton-Raphson power flow calculation method [108]. The Jacobian matrix is simplified based on assumptions which make DC load flow linear so that power flow problems can be solved relatively quickly and simply [109]. Under rules of DC power flow, the active power flow between bus j and bus k can be expressed as:

$$P_{j-k} = \frac{V_j V_k}{z_{j-k}} \sin(\theta_j - \theta_k)$$
(4.2)

#### Where

- $P_{j-k}$ : The active power flow between bus *j* and bus *k*
- $V_i$ : The voltage of bus j
- $V_k$ : The voltage of bus k
- $\theta_i$ : The phase angle of bus *j*
- $\theta_k$ : The phase angle of bus k
- $z_{j-k}$ : The impedance connected between bus j and bus k

Here these assumptions are adopted such as  $x_{j-k} \gg r_{j-k}$ ,  $V_j = V_k \approx 1$  and  $(\theta_j - \theta_k)$  is very small. So equation (4.2) can be rewrote as:

$$P_{j-k} = \frac{1}{x_{j-k}} \left( \theta_j - \theta_k \right) = b_{j-k} \left( \theta_j - \theta_k \right)$$
(4.3)

Where

 $x_{j-k}$ : The reactance connected between bus *j* and bus *k* 

 $b_{j-k}$ : The susceptance between bus *j* and bus *k* 

Then a matrix form can be expressed associated with equation (4.3):

$$[P] = [B][\theta] \tag{4.4}$$

Where

- [P] : The matrix of active power injection at buses
- $[\theta]$ : The matrix of phase angle of bus voltages
- [*B*] : The susceptance matrix

However, [B] is a singular matrix with rank one deficiency so it is unable to be inverted [110]. Hence a slack bus is selected as a solution to find the inverse of [B]. The whole solution process is shown as:

Firstly, choose a reference bus or called the slack bus. Then eliminate the row and the column which are based on the reference bus so [B] becomes  $[B^*]$ . The eliminated row and column are associated with the selection of slack bus. For example, if the slack bus is assigned as bus number n, the  $n_{th}$  row and  $n_{th}$  column of [B] will be eliminated.  $[B^*]$  is able to be inverted. Get  $[M^*]$  by inverting  $[B^*]$  which is shown in equation (4.5):

$$[B^*]^{-1} = [M^*] \tag{4.5}$$

Where

 $[B^*]$ : The matrix formed eliminating one row and one column of [B]

 $[M^*]$ : The inverse of  $[B^*]$ 

Then add one row and one column of 0 to  $[M^*]$  based on the reference bus. The added row and column are also associated with the selection of slack bus. For example, if the slack bus is assigned as bus number n, the  $n_{th}$  row 0 and  $n_{th}$  column 0 will be added back to  $[M^*]$ . Since [M] is obtained, the relation between  $[\theta]$  and [P] can be expressed as follows:

$$[\theta] = [M][P] \tag{4.6}$$

Where

[M]: The matrix is from  $[M^*]$  by adding one row and one column of 0 according to the corresponding reference bus or slack bus.

Then equation (4.6) can be rewritten as follow:

$$[M] = \frac{[\theta]}{[P]} \tag{4.7}$$

The equation (4.3) can be also expressed to indicate the change quantity of power on line j-k as follow:

$$\Delta P_{j-k} = b_{j-k} \left( \Delta \theta_j - \Delta \theta_k \right) \tag{4.8}$$

Then let both sides of equation divide  $\Delta P_i$ , which leads to the following equation:

$$\frac{\Delta P_{j-k}}{\Delta P_i} = b_{j-k} \left( \frac{\Delta \theta_j}{\Delta P_i} - \frac{\Delta \theta_k}{\Delta P_i} \right)$$
(4.9)

Based on former definition of PTDF in equation (4.1) and the relationship between  $\Delta\theta$  and  $\Delta P$  in equation (4.7), the expression of PTDF can be calculated from equation (4.9) as follows [110]:

$$PTDF_{i}^{j-k} = \frac{\Delta P_{j-k}}{\Delta P_{i}} = b_{j-k} [M_{ji} - M_{ki}]$$
(4.10)

Where

 $M_{ji}$ : The  $j_{th}$  row and  $i_{th}$  column element in the M matrix

 $M_{ki}$ : The  $k_{th}$  row and  $i_{th}$  column element in the M matrix

Since the added row and column are both consisted of 0, the PTDF of slack bus to any congested transmission line is always 0 [110]. Different slack bus selections would cause different PTDF values, which means this method is dependent on the selection of slack bus and might produce different results [108].

### 4.4. Mathematical Model for the Proposed Method with Power Transfer Distribution Factor

Once the sensitivity index PTDF is obtained, the corresponding mathematical model is able to be established. In this chapter, when a single transmission line congestion occurs, the system operator will re-dispatch generation schedules based on the above sensitivity factor and incremental/decremental bids submitted by generation units. The system operator is required to minimize the total cost of rescheduling generation, in the meantime, it should obey the physical rules of power system such as balancing of supply and demand, transmission line constraints and generator capacity [111]. In order to strengthen the achievement of sensitivity factor, all the generators in the system are required to be involved in incremental/decremental bids submission stage [24]. To simplify calculation process, incremental and decremental bids from each generator are assumed at the same price. The objective function and corresponding constraints are shown as:
$$\min \sum_{i=1}^{n_g} C_{G_i} \cdot |\Delta P_{G_i}| \tag{4.11}$$

Subject to:

$$\sum_{i=1}^{n_g} \Delta P_{G_i} = 0 \tag{4.12}$$

$$\sum_{i=1}^{n} PTDF_{G_i}^{j-k} \cdot \Delta P_{G_i} \leq P_{j-k}^{lim} - P_{j-k}^{uncon}$$

$$(4.13)$$

When

$$P_{j-k}^{lim} - P_{j-k}^{uncon} < 0$$

$$\sum_{i=1}^{n} PTDF_{G_i}^{j-k} \cdot \Delta P_{G_i} \ge P_{j-k}^{lim} - P_{j-k}^{uncon}$$
(4.14)

When

$$P_{j-k}^{lim} - P_{j-k}^{uncon} > 0$$
  
$$-P_{G_i} \le \Delta P_{G_i} \le P_{G_i}^{max} - P_{G_i}$$
(4.15)

Where

 $C_{G_i}$ : The incremental/decremental price of generator *i* 

 $\Delta P_{G_i}$ : The incremental/decremental amount of active power injection of generator *i* 

$$PTDF_{G_i}^{j-k}$$
: The PTDF of the line *j*-*k* respect to bus *i*

- $P_{j-k}^{lim}$ : Transmission capacity limit of line *j*-*k*
- $P_{i-k}^{uncon}$ : The active power flow of line *j*-*k* in unconstraint dispatch
- $P_{G_i}$ : The generation amount of generator *i* in unconstraint dispatch
- $P_{G_i}^{max}$ : The generation capacity of generator *i*

Equation (4.11) represents minimization of the total cost of generation redispatching. No matter if the cost is from incremental bids or decremental bids, they are all counted into the cost of congestion. Equation (4.12) indicates that the sum of power output variation from all the bids should be equal to 0 in order to keep the balance of the whole power system. Equation (4.13) and equation (4.14) express that the re-dispatching of generation is based on the sensitivity factor PTDF. The sum of generation variation in re-dispatch should result in the power flow on the congested line reduce below its transmission capacity. Equation (4.15) requests that the total incremental power volume of a generator should be limited to its generation capacity minus its scheduled power output and the total decremental power volume should be limited to its scheduled power output.

# 4.5. Case Studies

The case studies are consisted of the testing results from four modified IEEE bus systems including an IEEE-14 bus system, an IEEE-30 bus system, an IEEE-57 bus system and an IEEE-118 bus system. The test is targeted at the performance of the proposed re-dispatch method associated with the sensitivity factor PTDF. The results are mainly about the generation variations in re-dispatch and the corresponding congestion costs.

# 4.5.1. Test Description

In the case studies, in order to make power flow contribution more distinct and straightforward on transmission network, each bus is considered only has either a generator or a load. All the generators are assumed to have the same generation capacity. DC optimal power flow (DCOPF) is implemented by software Matpower to calculate the active power flows due to its fast calculating speed and easy convergence [35]. All transmission lines are assumed to be lossless so as to neglect the effect of power losses to active power flow so that transmission congestion will be the only consideration that needs to be focused on [112]. Reactive power source in the system is assumed to be sufficient hence all the bus voltages are fixed at one per unit [113]. For demonstration purpose, it is assumed that there is only one bottleneck for transmission in the network. This is the congested line. Number of congested transmission line in each test scenario is restricted to one only in order to highlight the sensitivity method efficiency on one particular line. In the test environment, parameters of power system are set arbitrarily such as bid price of each generator, generation capacity of each generator, active power consumption of each load, and transfer capacity of each transmission line. After running an unconstrained dispatch without considering transmission constraints, it is found that power flow of one particular transmission line is over its transfer capacity. So here transmission congestion occurred on this transmission line. Since different slack bus selections lead to different PTDF values, test associated to PTDF method is aim to find the different results about redispatched generation volume and congestion cost when selecting different slack bus. Each of the generator bus is proposed as the slack bus for each case [114].

Besides, the level of overloading (LO) is introduced in the test scenario to find the limitation of the proposed method with PTDF when transmission congestion become increasingly severe. LO is defined as:

$$LO = \frac{P_{j-k}{}^{uncon} - P_{j-k}{}^{lim}}{P_{j-k}{}^{uncon}} \times 100\%$$
(4.16)

Where

*LO* : The level of the overloading of the congested transmission line  $P_{j-k}^{uncon}$  : The active power of congested line *j-k* in unconstrained dispatch  $P_{j-k}^{lim}$  : The limit value of active power transmission capacity of congested line *j-k* 

LO equals zero means there is no congestion on line *j*-k. When LO is above 0% and less than 100%, it shows the line *j*-k is congested to some degree. If LO reaches 100%, which means the transfer capacity limitation of this particular transmission line reaches zero, the transmission line is totally overloaded.

# 4.5.2. Modified IEEE-14 Bus System

A modified IEEE-14 bus system is chosen as the first test system in this case study as shown in Figure 4.2.



Figure 4.2 The modified IEEE-14 bus system

The 14-bus system consists of 5 generators, 9 loads and 20 transmission lines. Before the test, some parameters of the test environment are set as follows:

Table 4.1 Generator and Load Data in 14 Bus System

Bus	Generator	Incremental Bid price (£/MWh)	Decremental Bid price (£/MWh)	Max power output (MW)	Load(MW)
1	Gen1	30	30	200	0
2	Gen2	30	30	200	0
3	Gen3	40	40	200	0
4	-	-	-	-	50
5	-	-			50
6	Gen6	40	40	200	0
7	-	-	-	-	60
8	Gen8	45	45	200	0
9	-	-	-	-	50
10	-	-	-	-	60
11	-	-	-	-	110
12	-	-	-	-	40
13	-	-	-	-	40
14	-	-	-	-	40
Total load					500

In Table 4.1, incremental/decremental bid prices, maximum MW generation outputs of five generators, MW demand of each load and total load are set in the table.

After a DC optimal power flow without considering the transmission line constraints, the generation schedule for each generator in the unconstrained dispatch is obtained. Then it is found that the power flow of transmission line 3-4 in the unconstrained dispatch is more than its transmission capacity limit. Therefore, congestion occurs on the line 3-4 which is regarded as the congested line in 14-bus case. Such above information is found in Table 4.2.

# Table 4.2 Generation Schedule and the Congested Line Power Flow in theUnconstrained Dispatch in 14 Bus System

Generator	Generation Schedule in the Unconstrained dispatch (MW)
Gen1	153.41
Gen2	26.40
Gen3	160.10
Gen6	160.10
Gen8	0
Transmission line	Unconstrained power flow (MW)
3-4	121.74
	Transmission line limit (MW)
	66.00

Since the power flow of transmission line 3-4 in unconstrained dispatch is 121.74 MW which exceeds its transmission capacity limit 66 MW. According to the equation 4.14, the level of overloading of the congested line 3-4 is 45%. Based on the different slack bus selections, the re-dispatch generation variation of each generator calculated by the proposed method when congested line's LO equals to 45% is obtained and the results are shown in Table 4.3. In order to make it more straightforward, contents of Table 4.3 are visualized in Figure 4.3. Finally, the congestion costs of different slack bus cases are indicated in Figure 4.4.

 Table 4.3 Generator Output Variation in Each Slack Bus Selection Due to Re 

 dispatch in 14-Bus System when LO=45%

Generator	ΔP <sub>G</sub> Slack bus 1 (MW)	ΔP <sub>G</sub> Slack bus 2 (MW)	$\Delta P_G$ Slack bus 3 (MW)	ΔP <sub>G</sub> Slack bus 6 (MW)	ΔP <sub>G</sub> Slack bus 8 (MW)
G1	0	0	0	0	+10.11
G2	0	0	0	0	0
G3	-92.80	-92.80	-92.79	-92.79	-95.16
G6	+38.85	+38.88	+38.84	+38.84	+38.86
G8	+53.94	+53.91	+53.95	+53.95	+46.19



Comparison of generator output variation when LO=45%

Figure 4.3 14-bus system generator output variation in re-dispatch when LO=45%



Congestion cost comparison when LO=45%

Figure 4.4 14-bus system congestion costs of re-dispatch in five slack bus cases when LO =45%

From the Figure 4.3, by using the proposed method with PTDF sensitivity factor to relieve the congestion under the level of overloading of target line at 45%, for different slack bus selection, the different re-dispatch generation variation of each generator is obtained. Generation re-dispatching in the first four slack bus selections are almost the same because generator 3 in these four cases is all asked to decrease its output and generator 6 and 8 are all ordered to increase their initial outputs. Furthermore, their output variation volumes are nearly the same. As a result, from the Figure 4.4, the corresponding congestion costs of these four cases only have slight differences which could be neglected. But when bus 8 is selected as the slack bus, it is found that the increase of generator 1's output leads to generator 3 output being decreased more than the first four cases and generator 8 increasing it the output less than in the other four cases. This is the reason why the congestion cost in slack bus 8 case 7742 £/h is slightly more than other slack bus cases 7693 £/h.

#### 4.5.3. Modified IEEE-30 Bus System

A modified IEEE-30 bus system is the second test system in this case study as shown in Figure 4.5. The 30-bus system consists of 6 generators, 18 loads and 41 transmission lines.



Figure 4.5 The modified IEEE-30 bus system

Before the test, some parameters of the test environment are set as follows:

# Table 4.4 Generator Data in 30 Bus System

Bus	Generator	Incremental Bid price (£/MWh)	Decremental Bid price (£/MWh)	Max power output (MW)
1	Gen1	35	35	300
2	Gen2	35	35	300
5	Gen5	45	45	300
8	Gen8	45	45	300
11	Gen11	45	45	300
13	Gen13	50	50	300

Bus	Load(MW)	Bus	Load(MW)	Bus	Load(MW)
1	-	11	-	21	30
2	-	12	60	22	-
3	30	13	-	23	50
4	40	14	50	24	60
5	-	15	50	25	-
6	-	16	60	26	50
7	50	17	70	27	-
8	-	18	30	28	-
9	-	19	50	29	30
10	60	20	30	30	50
Total load					850

## Table 4.5 Load Data in 30 Bus System

In Table 4.4, incremental/decremental bid prices, maximum MW generation outputs of six generators are set. In Table 4.5, each load MW and total load MW are set as above.

After running a DC optimal power flow without considering the transmission line constraints, the generation schedule for each generator in the unconstrained dispatch is obtained. Then it is shown that the power flow of transmission line 4-12 in the unconstrained dispatch is more than its transmission capacity limit. Therefore, congestion occurs on the line 4-12 which is the congested line in 30-bus case. Such above information is shown in Table 4.6 as follows:

Generator	Generation Schedule in the Unconstrained dispatch (MW)
Gen1	185.28
Gen2	28.48
Gen5	212.08
Gen8	212.08
Gen11	212.08
Gen13	0
Transmission line	Unconstrained power flow (MW)
4-12	224.66
	Transmission line limit (MW)
	107.84

 Table 4.6 Generation Schedule and the Congested line power flow in the

 Unconstrained Dispatch in 30 Bus System

Because the power flow of transmission line 4-12 in unconstrained dispatch is 224.66 MW which is over its transmission capacity limit 107.84 MW. According to the equation 4.14, the level of overloading of the congested line 4-12 reaches to 52%. According to the six slack bus selections, the re-dispatch generation variation of each generator calculated by the proposed method with the LO of congested line equals to 52% is acquired and the results are shown in Table 4.7. To make it more comparable, contents of Table 4.7 is visualized in Figure 4.6. Finally, the corresponding congestion costs of six slack bus cases are indicated in Figure 4.7.

	$\Delta P_{G}$					
Generator	Slack bus 1	Slack bus 2	Slack bus 5	Slack bus 8	Slack bus 11	Slack bus 13
	( <b>MW</b> )					
G1	-185.15	-185.15	-185.15	-185.15	-185.15	-185.15
G2	-5.32	-5.32	-5.32	-5.32	-5.32	-5.32
G5	0	0	0	0	0	0
G8	0	0	0	0	0	0
G11	0	0	0	0	0	0
G13	+190.47	+190.47	+190.47	+190.47	+190.47	+190.47

Table 4.7 Generator Output Variation in Each Slack Bus Selection Due to Redispatch in 30-Bus System when LO=52%

Comparison of generator output variation when LO=52%



Figure 4.6 30-bus system generator output variation in re-dispatch when LO=52%



Congestion cost comparison when LO=52%

Figure 4.7 30-bus system congestion costs of re-dispatch in six slack bus cases when LO=52%

From Figure 4.6, by selecting different buses as slack bus, the generation variation of each generator in re-dispatch stage under the congestion level of the congested line equals to 52% in 30 bus system is calculated by the PTDF method. Generation re-dispatching in all the slack bus selections are identical since in all cases generator 1 and 2 are asked to decrease their outputs and generator 13 is ordered to increase its initial outputs. Furthermore, their output variation volumes in all cases are nearly the same. So in Figure 4.7, the corresponding congestion costs of six cases are nearly identical at 16189 £/h.

# 4.5.4. Modified IEEE-57 Bus System

A modified IEEE-57 bus system is the third test system in this case study as shown in Figure 4.8. The 57-bus system consists of 7 generators, 35 loads and 80 transmission lines.



Figure 4.8 The modified IEEE-57 bus system

Before the test, some parameters of the test environment are set as follows:

Bus	Generator	Incremental Bid price (£/MWh)	Decremental Bid price (£/MWh)	Max power output (MW)
1	Gen1	30	30	300
2	Gen2	35	35	300
3	Gen3	30	30	300
6	Gen6	45	45	300
8	Gen8	45	45	300
9	Gen9	45	45	300
12	Gen12	40	40	300

# Table 4.8 Generator Data in 57 Bus System

Bus	Load(MW)	Bus	Load(MW)	Bus	Load(MW)
1	-	20	60	39	-
2	-	21	-	40	-
3	-	22	-	41	40
4	-	23	30	42	30
5	60	24	-	43	30
6	-	25	50	44	20
7	-	26	-	45	-
8	-	27	30	46	-
9	-	28	20	47	50
10	30	29	50	48	-
11	-	30	30	49	20
12	-	31	30	50	30
13	20	32	20	51	30
14	20	33	30	52	40
15	40	34	-	53	30
16	20	35	50	54	50
17	50	36	-	55	50
18	60	37	-	56	30
19	30	38	20	57	50
Total load					1250

## Table 4.9 Load Data in 57 Bus System

In Table 4.8, incremental/decremental bid prices, maximum MW generation outputs of seven generators are set as above. In Table 4.9, each load demand MW and the total load MW are shown.

After running a DC optimal power flow without considering the transmission line constraints, the generation schedule for each generator in the unconstrained dispatch is gotten. Then power flow results indicate that the power flow of transmission line 7-29 under the unconstrained dispatch exceeds its maximum transmission capacity. Therefore,

in 57-bus case, transmission congestion occurs on the line 7-29. Information about generation schedules and the power flow of the congested line in the unconstrained dispatch are shown in Table 4.10 as follows:

Table	4.10	Generation	Schedule	and	the	Congested	Line	Power	Flow	in	the
Uncon	strain	ed Dispatch	in 57 Bus	Syste	m						

Generator	Generation Schedule in the Unconstrained dispatch (MW)
Gen1	129.50
Gen2	300
Gen3	40.19
Gen6	254.64
Gen8	114.59
Gen9	254.64
Gen12	156.44
Transmission line	Unconstrained power flow (MW)
7-29	213.05
	Transmission line limit (MW)
	196.01

Since the power flow of transmission line 7-29 in unconstrained dispatch is 213.05 MW which is more than its transmission capacity limit 196.01 MW. By using the equation 4.14, the level of overloading of the congested line 7-29 is 8%. According to situations of selecting the different seven generator buses as slack bus, the re-dispatch generation variation of each generator calculated by the proposed method with the LO of congested line equals to 8% is obtained and the results are shown in Table 4.11. To make it more comparable, contents of Table 4.11 is visualized in Figure 4.9. Finally, the corresponding congestion costs of six slack bus cases are indicated in Figure 4.10.

	$\Delta P_{G}$						
Generator	Slack bus						
	1	2	3	6	8	9	12
	( <b>MW</b> )						
G1	+170.47	+170.47	+170.47	+170.47	+170.47	+170.47	+170.47
G2	0	0	0	0	0	0	0
G3	+0.21	0	0	0	0	+0.21	0
G6	-208.98	-208.93	-208.93	-208.93	-208.93	-208.98	-208.93
G8	0	0	0	0	0	0	0
<b>G9</b>	0	0	0	0	0	0	0
G12	+38.31	+38.46	+38.46	+38.46	+38.46	+38.31	+38.46

Table 4.11 Generator Output Variation Due to Re-dispatch in 57-bus System when LO=8%



Comparison of generator output variation when LO=8%

Figure 4.9 57-bus system generator output variation in re-dispatch when LO=8%



Congestion cost comparison when LO=8%

Figure 4.10 57-bus system congestion costs of re-dispatch in seven slack bus cases when LO=8%

From Figure 4.9, in the 57 bus system, by selecting different generator buses as slack bus, generation variation of each generator in the re-dispatch stage under the level of overloading of the congested line 7-29 equals to 8% is calculated by the proposed PTDF sensitivity method. Generation reschedules in all the slack bus cases are all very similar since generator 1 and generator 12 are all asked to increase their outputs and generator 6 is asked to decrease its output in all cases. It is also found that the output variation volumes in all cases are very close. As a consequence, in Figure 4.10, the corresponding congestion costs of seven cases are nearly equal. However, it was found that generator 3 is involved in the re-dispatch when the slack bus was selected at bus 1 and bus 9 (highlighted in Table 4.11). This is the reason why the congestion costs of case 1 and case 9 which are both around 16056 £/h are a little bit more than other cases which are 16054 £/h. But since the involved power volume increase of generator 3 is so small compared with the total power variation volume, which is 0.1%, so it can be neglected by the system operator.

# 4.5.5. Modified IEEE-118 Bus System

A modified IEEE-118 bus system is the last test system in this case study as shown in Figure 4.11. It is noticed that the 118-bus system consists of 12 generators, 104 loads and 186 transmission lines. Before the testing, some parameters of the power system need to be set up. In Table 4.12, incremental/decremental bid prices, maximum MW generation outputs of the twelve generators can be found. In Table 4.13, each load demand MW and the total load MW are displayed.



Figure 4.11 The modified IEEE-118 bus system

Bus	Generator	Incremental Bid price (£/MWh)	Decremental Bid price (£/MWh)	Max power output (MW)
12	Gen12	30	30	300
19	Gen19	30	30	300
24	Gen24	35	35	300
27	Gen27	40	40	300
40	Gen40	30	30	300
49	Gen49	35	35	300
59	Gen59	35	35	300
61	Gen61	40	40	300
70	Gen70	30	30	300
85	Gen85	35	35	300
99	Gen99	35	35	300
110	Gen110	40	40	300

<b>Table 4.12</b>	Generator	Data in	118	Bus	System
-------------------	-----------	---------	-----	-----	--------

# Table 4.13 Load Data in 118 Bus System

Bus	Load(MW)	Bus	Load(MW)	Bus	Load(MW)
1	30	41	30	81	10
2	20	42	60	82	50
3	30	43	20	83	20
4	30	44	15	84	10
5	10	45	50	85	-
6	30	46	20	86	10
7	20	47	30	87	10
8	30	48	20	88	40
9	10	49	-	89	10
10	20	50	15	90	10
11	50	51	15	91	10
12	-	52	15	92	60

13	30	53	20	93	15
14	15	54	10	94	30
15	60	55	60	95	40
16	20	56	80	96	30
17	10	57	15	97	15
18	10	58	15	98	30
19	-	59	-	99	-
20	15	60	70	100	30
21	15	61	-	101	20
22	10	62	70	102	15
23	40	63	10	103	20
24	-	64	20	104	30
25	50	65	10	105	30
26	30	66	30	106	40
27	-	67	20	107	50
28	10	68	10	108	20
29	20	69	-	109	10
30	-	70	-	110	-
31	40	71	10	111	10
32	50	72	10	112	6
33	20	73	10	113	10
34	50	74	60	114	10
35	30	75	50	115	20
36	10	76	60	116	10
37	10	77	60	117	20
38	10	78	70	118	30
39	20	79	30		
40	-	80	10		
Total load					2736

After a DC optimal power flow without considering the transmission line constraints, the initial generation schedule for each generator in the unconstrained dispatch stage is given. Then results of optimal power flow indicate that the power flow of

transmission line 23-24 in the unconstrained dispatch is above its maximum transmission capacity. Therefore, in 118-bus case, transmission congestion occurs on the line 23-24. Information about generation schedules and the power flow of the congested line in the unconstrained dispatch are shown in Table 4.14 as follows:

Table	4.14	Generation	Schedule	and	the	Congested	Line	Power	Flow	in	the
Uncon	strain	ed Dispatch	in 118 Bus	Syst	em						

Generator	Generation Schedule in the Unconstrained dispatch (MW)
Gen12	71.17
Gen19	300
Gen24	300
Gen27	300
Gen40	300
Gen49	119.82
Gen59	91.04
Gen61	53.97
Gen70	300
Gen85	300
Gen99	300
Gen110	300
Transmission line	Unconstrained power flow (MW)
23-24	-211.43
	Transmission line limit (MW)
	2.11

Since the power flow of transmission line 23-24 in the unconstrained dispatch stage is -211.43 MW which is more than its transmission capacity limit 2.11 MW. It should be noted that the negative value means the power flow of the congested line is on an opposite direction. Based on the equation 4.14, the level of overloading of the congested line 23-24 is obtained at 99%. By selecting the different twelve generator buses

as the slack bus, the re-dispatch generation variation of each generator calculated by the proposed method with the LO of congested line equals to 99% is obtained and the results are shown in Table 4.15. To make it more comparable, contents of Table 4.15 is visualized in Figure 4.12. Finally, the corresponding congestion costs of twelve slack bus cases are shown in Figure 4.13.

	$\Delta P_{G}$	$\Delta \boldsymbol{P}_{\boldsymbol{G}}$	$\Delta P_{G}$	$\Delta P_G$	$\Delta \boldsymbol{P}_{\boldsymbol{G}}$	$\Delta P_{G}$	$\Delta P_G$	$\Delta P_{G}$	$\Delta P_G$	$\Delta P_{G}$	$\Delta P_{G}$	$\Delta P_{G}$
	Slack	Slack	Slack	Slack	Slack	Slack	Slack	Slack	Slack	Slack	Slack	Slack
Generator	bus 12	bus 19	bus 24	bus 27	bus 40	bus 49	bus 59	bus 61	bus 70	bus 85	bus 99	bus 110
	( <b>MW</b> )	( <b>MW</b> )	( <b>MW</b> )	( <b>MW</b> )	( <b>MW</b> )	( <b>MW</b> )	( <b>MW</b> )	( <b>MW</b> )	( <b>MW</b> )	( <b>MW</b> )	( <b>MW</b> )	( <b>MW</b> )
Gen12	+228.83	+228.83	+228.83	+228.83	+228.83	+228.83	+228.83	+228.83	+228.83	+228.83	+228.83	+228.83
Gen19	0	0	0	0	0	0	0	0	0	0	0	0
Gen24	-296.55	-296.55	-296.50	-296.50	-296.55	-296.55	-296.50	-296.50	-296.55	-296.50	-296.50	-296.55
Gen27	0	0	0	0	0	0	0	0	0	0	0	0
Gen40	0	0	0	0	0	0	0	0	0	0	0	0
Gen49	+62.65	+62.65	+67.68	+67.68	+62.65	+62.65	+67.68	+67.67	+62.65	+67.68	+67.66	+62.67
Gen59	+5.09	+5.09	0	0	+5.09	+5.09	0	0	+5.09	0	0	+5.09
Gen61	0	0	0	0	0	0	0	0	0	0	0	0
Gen70	0	0	0	0	0	0	0	0	0	0	0	0
Gen85	0	0	0	0	0	0	0	0	0	0	0	0
Gen99	0	0	0	0	0	0	0	0	0	0	0	0
Gen110	0	0	0	0	0	0	0	0	0	0	0	0

Table 4.15 Generator Output Variation in Each Slack Bus Selection Due to Re-dispatch in 118 Bus System at LO=99%



# Generator output variation in re-dispatch when LO=99%

Figure 4.12 118-bus system generator output variation in re-dispatch when LO=99%



# Congestion cost of re-dispatch when LO=99%

Figure 4.13 118-bus system congestion cost of re-dispatch in all slack bus cases when LO=99%

From Figure 4.12, by selecting different generator buses as the slack bus, the redispatch generation variation of each generator when the level of overloading of the congested transmission line 23-24 equals to 99% in 118 bus system is decided by the proposed sensitivity method. As noticed that there are eight generators have already output their maximum capacities in the unconstrained dispatch stage. So there are only four generators left still have extra capacity to allow the system operator to re-dispatch. In all twelve cases, the variation output of generator 12 and 24 are same. The difference is generator 59 is asked to increase its output when the slack bus is selected on bus 12, 19, 40, 49, 70 and 110. Furthermore the increased output variation volume in these cases are nearly the same as 5.09 MW. As a result, the power change volumes of generator 49 in such cases are different from other cases in which generator 59 is not involved in redispatching. So in Figure 4.13, the corresponding congestion costs of the twelve cases are slightly different. The congestion cost of the cases with generator 59 participation is 19615 £/h while the cost of the cases without generator 59 participation is 19611 £/h.

#### 4.5.6. Discussion

The proposed re-dispatch scheme with the sensitivity index PTDF is tested under four different IEEE modified power systems. Overall, the method is capable to re-dispatch the generation schedule based on the objective function with three constraints. The sensitivity factor PTDF is able to find the power flow contribution of each generator to the congested line so the re-dispatching is become more effective and accurate. Based on the rules of the incremental and decremental bids, the corresponding cost due to transmission congestion can be obtained. Additionally, the introduction of the incremental and decremental bids improves the motivation of market participants to be involved in the re-dispatching action so the system operator is allowed to remove the transmission congestion risk with plenty of power resource under a minimum cost. Four modified test system results demonstrate the capability of the proposed PTDF method for generation re-dispatching. From 14-bus test system up to 118-bus test system, the proposed method can all provide a solution to relieve transmission congestion issue by re-dispatching initial generation schedules. With the help of the concept that the level of overloading (LO) of a congested line, the performance of the proposed method is further demonstrated. In the 14-bus case, with the overloading level of the congested line 3-4 reaching 45%, the PTDF method solved the congestion problem properly. In the 30-bus case, the LO of the congested line 4-12 is 52% but the method still works. In the 57-bus case, the LO of the congested line 7-29 is only at 8% and the proposed method fulfilled the rescheduling of the generation plan by asking three generators to change their initial generation outputs. In the 118-bus case, the LO of the congested line 23-24 came up to 99%. It should be noted that this case is very extreme and might be unrealistic but technically it can reflect the proposed method's performance in utmost conditions. The outcome is the PTDF sensitivity method provided a solution to handle this severe congestion problem.

However, the potential weakness is that the PTDF index method is dependent on the slack bus selection. The experimental outcomes proved that the results are indeed different when choosing different generator bus as the slack bus. In the 57-bus case, t a difference in the results existed but the distinction is very small so it can be ignored. In the 14-bus case and 118-bus case, the outcome differences between different slack bus selections are granted. Consequently, it may make the PTDF method become inaccurate if only consider one generator bus as the slack bus. Whereas, if all the generator buses are respectively chosen as the slack bus plus the power system has a number of generators, the result calculation process will become complex. The case of 118-bus system is a difficult example to test it because there are twelve possible choices for the slack bus which need to be considered.

# 4.6. Summary

This chapter introduced a proposed method with the sensitivity index PTDF to redispatch generation when transmission congestion occurs in the power system. Some conventional re-dispatch generation methods are described before the content of the proposed method. The differences between the conventional method and the proposed method are indicated, which makes the latter one more effective and reasonable. The core concept of this proposed method is a sensitivity index PTDF whose algorithm is described with a series of formulae. Then the mathematic model of the proposed method including an objective function and three constraints one of which includes the sensitivity factor are displayed in detail. Case studies consist of four test systems which are modified IEEE systems respectively with 14 buses, 30 buses, 57 buses and 118 buses. Experimental results show that the proposed method is capable under most random circumstances. However, some results proved that it still has drawback. Dependence of slack bus selection lead to the inaccuracy and the complexity of the PTDF based method. This disadvantage should be addressed or another approach will come to replace this method as a solution.

# **Chapter 5**

# **Re-dispatch** Generation in Deregulated Electricity Market with Bus Impedance Matrix Sensitivity

# 5.1. Introduction

In chapter 4, a proposed generation re-dispatch method with the power transfer distribution factor is introduced, described and tested. Through experimental results from four test systems, it is proved that the weakness of the method is the dependence of slack bus selection. Choosing different generator bus as the slack bus leads to the different outcomes of generation rescheduling and the corresponding security costs become more or less distinct as well. Hence the method might be inaccurate on the generation re-dispatching if there is a transmission congestion problem occurs and generators would be in front of an unfair command to be asked to adjust their generation outputs. In this chapter, another sensitivity factor is recommended to apply in the proposed re-dispatching method, which is called the bus impedance matrix sensitivity. As its name implies, the sensitivity index is related to the bus impedance matrix of the power flow system and is able to determine the sensitivity of a particular transmission line power flow with respect to power injection variation at a particular bus. This chapter is aim to test the performance of the re-dispatch scheme with above sensitivity factor and compare it with the PTDF based method under the same test conditions.

In this chapter, the proposed generation re-dispatch method associated with the bus impedance matrix sensitivity will be represented in detail including the methodology of the proposed re-dispatch method in section 5.2. The descriptions of the admittance matrix and the impedance matrix are in the section 5.3. The algorithm of the sensitivity

index is in the section 5.4. Section 5.5 will give the mathematic model of the proposed method including an objective function and three constraints one of which contains the bus impedance matrix sensitivity. Section 5.6 is the case study which consisted of experimental results from four modified IEEE testing systems including 14-bus, 30-bus, 57-bus and 118-bus systems. Comparison between the bus impedance matrix sensitivity method and the power transfer distribution factor method will also be indicated in this section. Section 5.7 is the summary.

# 5.2. Proposed Generation Re-dispatch Method in Deregulated Electricity Market

The mechanism of the proposed generation re-dispatch method in this chapter is as same as the method in Chapter 4. The details of the proposed method principle can be found in section 4.2. The only difference is the proposed method in this chapter adopts two different sensitivity factors to calculate the power flow contribution of each node to the targeted transmission line. This time is the bus impedance matrix sensitivity which will be explained in the following section.

#### **5.3.** The Admittance Matrix and the Impedance Matrix

This section will introduce the admittance matrix and the impedance matrix including their formulations and functions.

#### 5.3.1. The Admittance Matrix

In power engineering, the nodal admittance matrix which is also called admittance matrix, Y matrix or Ybus is regarded as a mathematical model to express the power system network through a simplified calculation [115]. Power system transmission requires optimized management for real and reactive power flows. When a set of load demands, voltages and currents are given, the necessary real and reactive power flows in the power

system needs to be determined. In most realistic power systems, situations are always very complex, which is difficult to deal with. Therefore, computer simulation is required to solve above problems. The admittance matrix is an effective tool which provides a route to simplify a complex power system to a matrix format so the power system problem can be worked out by a computer program.

From load flow equations, a nodal current equation for an n-bus power system is shown as follows [116]:

$$I_i = \sum_{j=1}^n Y_{ij} V_j, i = 1, 2, ..., n$$
(5.1)

Where

 $I_i$ : The nodal current in bus *i* 

 $Y_{ij}$ : The admittance matrix of the n-bus power system

 $V_i$ : The voltage on bus j

Equation (5.1) can be expressed in matrix form as follows:

$$\begin{bmatrix} I_1 \\ I_2 \\ \vdots \\ I_n \end{bmatrix} = \begin{bmatrix} Y_{11} & Y_{12} & \dots & Y_{1n} \\ Y_{21} & Y_{22} & \dots & Y_{2n} \\ \vdots & \vdots & \ddots & \vdots \\ Y_{n1} & Y_{n2} & \dots & Y_{nn} \end{bmatrix} \begin{bmatrix} V_1 \\ V_2 \\ \vdots \\ V_n \end{bmatrix}$$
(5.2)

Hence, in general the format of the Ybus matrix for an n-bus power system is as follows:

$$Y_{bus} = \begin{bmatrix} Y_{11} & Y_{12} & \dots & Y_{1n} \\ Y_{21} & Y_{22} & \dots & Y_{2n} \\ \vdots & \vdots & \ddots & \vdots \\ Y_{n1} & Y_{n2} & \dots & Y_{nn} \end{bmatrix}$$
(5.3)

It should be noted that the admittance matrix is a singular matrix [117]. Consequently, it is necessary to select one of the buses in the system as the slack bus or called reference bus, otherwise, a singular admittance matrix is difficult to be utilized in the power system calculation. By choosing one of the buses as the slack bus, the corresponding row and column in Ybus matrix are deleted so that the reduced admittance matrix turns into a non-singular [116]. Hence, the reduced Ybus matrix can be easily dealt with.

#### 5.3.2. The Impedance Matrix

Unlike the admittance matrix which is used very frequently in power flow study, the impedance matrix was not encouraged initially due to time, memory and computational requirements [118]. However, the impedance matrix becomes an important tool for power system analysis nowadays. If the bus impedance matrix of an n-bus system is available, such a matrix can be utilized for power flow study, short-circuit calculation and fault study [119]. There are a number of algorithms for building up a Zbus matrix including directly building up, non-singular transformation, graph theoretic approach and inversion of bus admittance matrix [118]. In this thesis, Zbus matrix is derived from Ybus matrix inversion. Hence, the Ybus inversion method will be concentrated on in the following content. From load flow equations, a nodal voltage equation for an n-bus power system is shown as follows to express the relation between bus voltage and bus current:

$$V_j = \sum_{i=1}^{n} Z_{ij} I_i, j = 1, 2, \dots, n$$
(5.4)

Where

 $Z_{ij}$ : The impedance matrix of the n-bus power system

Equation (5.4) can be expressed in matrix form as follows:

$$\begin{bmatrix} V_1 \\ V_2 \\ \vdots \\ V_n \end{bmatrix} = \begin{bmatrix} Z_{11} & Z_{12} & \dots & Z_{1n} \\ Z_{21} & Z_{22} & \dots & Z_{2n} \\ \vdots & \vdots & \ddots & \vdots \\ Z_{n1} & Z_{n2} & \dots & Z_{nn} \end{bmatrix} \begin{bmatrix} I_1 \\ I_2 \\ \vdots \\ I_n \end{bmatrix}$$
(5.5)

Hence, in general the formulation of the Zbus matrix for an n-bus power system is as follows:

$$Z_{bus} = \begin{bmatrix} Z_{11} & Z_{12} & \dots & Z_{1n} \\ Z_{21} & Z_{22} & \dots & Z_{2n} \\ \vdots & \vdots & \ddots & \vdots \\ Z_{n1} & Z_{n2} & \dots & Z_{nn} \end{bmatrix}$$
(5.6)

Based on the Ybus inversion algorithm, the relation between Zbus matrix and Ybus matrix should be represented as follows [116]:

$$Z_{bus} = Y_{bus}^{-1} \tag{5.7}$$

However, as mentioned in section 5.3.1, the Ybus matrix is singular and cannot be inverted. Hence the Ybus matrix must be turned into a non-singular matrix by deleting the corresponding row and column of the slack bus. For example, if bus 1 is selected as the slack bus, the corresponding row and column of bus 1 will be deleted so that the Ybus matrix becomes a reduced Ybus matrix.

$$Y_{bus}^{*} = \begin{bmatrix} - & & & & & \\ - & & & & & \\ \vdots & \vdots & & & \vdots \\ - & & & & & \\ Y_{n2} & & & & & \\ Y_{n2} & & & & \\ \end{bmatrix} = \begin{bmatrix} Y_{22} & & & & Y_{2n} \\ \vdots & & & \vdots \\ Y_{n2} & & & & \\ Y_{n2} & & & & \\ \end{bmatrix}$$
(5.8)

Where

 $Y_{bus}^{*}$ : The reduced admittance matrix of the n-bus power system

Now the reduced Ybus matrix has become a non-singular matrix so it is able to be inverted. The inverse of the reduced Ybus matrix, a reduced Zbus matrix can be obtained.

$$Z_{bus}^{\ *} = Y_{bus}^{\ *^{-1}} \tag{5.9}$$
Then by adding one row 0 and one column 0 into the reduced Zbus matrix, the Zbus matrix can be found.

$$Z_{bus} = \begin{bmatrix} 0 & \cdots & 0 \\ \vdots & & \\ 0 & & Z_{bus}^* \end{bmatrix}$$
(5.10)

In conclusion, the Zbus matrix is formulated by inversing the Ybus matrix with selecting a slack bus. It should be noted that the Zbus matrix derived via the inversion of Ybus matrix relies on slack bus selection and elements associated with slack bus are zero.

#### 5.4. The Algorithm of Bus Impedance Matrix Sensitivity

Bus impedance matrix sensitivity (ZS) is the second sensitivity factor to detect the sensitivity of a particular transmission line power flow with respect to a bus power injection. The derivation of this sensitivity factor originated from the expression of the complex power flow of a particular transmission line from bus *j* to bus *k* [24, 112, 120]. A transmission line can be represented by the  $\pi$  equivalent circuit, as shown in Figure 5.1 [121], and the algorithm of the bus impedance matrix sensitivity is displayed as follows:



Figure 5.1  $\pi$  Equivalent Circuit of Transmission Line j-k

The expression of the complex power flow over transmission line j-k from bus j is shown as:

$$S_{j-k} = V_j \cdot I_{j-k}^*$$
(5.11)

Where

 $S_{j-k}$ : The complex power flow of line *j*-*k* calculated at bus *j* 

 $V_i$ : The bus voltage at bus j

 $I_{j-k}^*$ : The conjugate of current through line *j*-*k* 

Using the impedance matrix, the voltage at bus *j* is shown as:

$$V_{j} = \sum_{i=1}^{n} Z_{ji} \cdot I_{i}$$
(5.12)

Where

 $Z_{ji}$ : The  $j_{th}$  row and  $i_{th}$  column element in the impedance matrix

 $I_i$ : The nodal current at bus i

The current through the line j-k is shown as:

$$I_{j-k} = (V_j - V_k) \cdot y_{j-k} + V_j \cdot y_{j-k}^{sh}$$
(5.13)

Where

 $I_{j-k}$ : The current through line *j*-*k* 

 $V_k$ : The bus voltage at bus k

 $y_{j-k}$ : The series admittance of the  $\pi$  equivalent circuit of line *j*-*k* 

 $y_{j-k}^{sh}$ : The shunt admittance of the  $\pi$  equivalent circuit of line *j*-*k* 

Substituting equation (5.12) and equation (5.13) into equation (5.11), the complex power flow over transmission line j-k is rewritten as:

$$S_{j-k} = \sum_{i=1}^{n} V_{j} \cdot \left[ \left( Z_{ji} - Z_{ki} \right) \cdot y_{j-k} + Z_{ji} \cdot y_{j-k}^{sh} \right]^{*} \cdot I_{i}^{*}$$
(5.14)

Where

 $Z_{ki}$ : The  $k_{th}$  row and  $i_{th}$  column element in the impedance matrix

 $I_i^*$ : The conjugate of nodal current at bus *i* 

The active power through line *j*-*k* is the real component of  $S_{j-k}$ :

$$P_{j-k} = \sum_{i=1}^{n} Re\left\{ V_j \left[ \left( Z_{ji} - Z_{ki} \right) \cdot y_{j-k} + Z_{ji} \cdot y_{j-k}^{sh} \right]^* \cdot I_i^* \right\}$$
(5.15)

#### Where

Re: Taking the real component only

Since  $I_i^*$  can be expressed as:

$$I_{i}^{*} = \frac{P_{i} + jQ_{i}}{V_{i}}$$
(5.16)

Where

- $P_i$ : The active power injected at bus i,
- $Q_i$ : The reactive power injected at bus i,
- $V_i$ : The voltage at bus *i*.

So the active power through line *j*-*k* can be rewritten as:

$$P_{j-k} = \sum_{i=1}^{n} Re \left\{ V_j \left[ \left( Z_{ji} - Z_{ki} \right) \cdot y_{j-k} + Z_{ji} \cdot y_{j-k}^{sh} \right]^* \cdot \frac{P_i + jQ_i}{V_i} \right\}$$
(5.17)

Neglecting the contribution of  $Q_i$  to  $P_{j-k}$ , the expression can be further rewritten as:

$$P_{j-k} = \sum_{i=1}^{n} Re \left\{ \frac{V_j}{V_i} \left[ \left( Z_{ji} - Z_{ki} \right) \cdot y_{j-k} + Z_{ji} \cdot y_{j-k}^{sh} \right]^* \cdot P_i \right\}$$
(5.18)

The relation between  $P_{i-k}$  and  $P_i$  has been found and simplified as:

$$P_{j-k} = \sum_{1}^{n} ZS_i^{j-k} \cdot P_i \tag{5.19}$$

$$ZS_{i}^{j-k} = Re\{\frac{V_{j}}{V_{i}}\left[\left(Z_{ji} - Z_{ki}\right) \cdot y_{j-k} + Z_{ji} \cdot y_{j-k}^{sh}\right]^{*}\}$$
(5.20)

Where

 $ZS_i^{j-k}$ : The bus impedance matrix sensitivity of line *j-k* respect to real power injection at bus *i*.

The total active power flow through a particular transmission line can be split into portions and each portion of the active power is associated with one particular bus power injection. The relation is shown as:

$$\Delta P_{j-k}^i = Z S_i^{j-k} \cdot \Delta P_i \tag{5.21}$$

Where

 $\Delta P_{j-k}^i$ : The portion of active power flow change through line *j*-*k* only associated with bus *i* power injection change

 $\Delta P_i$ : The power injection change of bus *i*.

As mentioned in section 5.3.2, the Zbus matrix is dependent on slack bus selection. Hence, the impedance matrix sensitivity method is also dependent on slack bus selection.

### 5.5. Mathematical Model for the Proposed Method with Bus Impedance Matrix Sensitivity

The mathematical model for the proposed method is the same as the one in chapter 4 and the details can be obtained in section 4.4. The only different point is that in the second constraint the PTDF sensitivity factor is replaced by the bus impedance matrix sensitivity. The mathematical model with the bus impedance matrix sensitivity is showed as follows:

$$\min \sum_{i=1}^{n} C_{G_i} \cdot \left| \Delta P_{G_i} \right| \tag{5.22}$$

Subject to:

$$\sum_{i=1}^{n} \Delta P_{G_i} = 0 \tag{5.23}$$

$$\sum_{i=1}^{n} ZS_{G_i}^{j-k} \cdot \Delta P_{G_i} \leq P_{j-k}^{lim} - P_{j-k}^{uncon}$$
(5.24)

When

$$P_{j-k}^{lim} - P_{j-k}^{uncon} < 0$$

$$\sum_{i=1}^{n} ZS_{G_i}^{j-k} \cdot \Delta P_{G_i} \ge P_{j-k}^{lim} - P_{j-k}^{uncon}$$
(5.25)

When

$$P_{j-k}^{lim} - P_{j-k}^{uncon} > 0$$

$$-P_{G_i} \le \Delta P_{G_i} \le P_{G_i}^{max} - P_{G_i} \tag{5.26}$$

Where

 $C_{G_i}$ : The incremental/decremental bid price of generator *i*   $\Delta P_{G_i}$ : The incremental/decremental amount of active power injection of generator *i*   $ZS_{G_i}^{j-k}$ : The bus impedance matrix sensitivity of the line *j-k* respect to bus *i*   $P_{j-k}^{lim}$ : Transmission capacity limit of line *j-k*   $P_{j-k}^{uncon}$ : The active power flow of line *j-k* in unconstrained dispatch  $P_{G_i}$ : The generation amount of generator *i* in unconstrained dispatch  $P_{G_i}^{max}$ : The generation capacity of generator *i* 

Same with the contents of section 4.4, equation (5.22) represents minimization of the total cost of generation re-dispatching. No matter if the cost is from incremental bids or decremental bids, they are all counted into the cost of congestion or called security cost. Equation (5.23) indicates that the sum of power output variations from all the bids should be equal to 0 in order to keep the balance of the whole power system. Equation (5.26) requests that the total incremental power volumes of a generator should be limited to its generation capacity minus its scheduled power output. In this chapter, equation (5.24) and equation (5.25) express that the re-dispatching of generation is based on the sensitivity

factor ZS rather than the PTDF. But the aim is identical that the sum of generation variations in re-dispatch should result in the power flow on the congested line reducing below its transmission capacity.

#### 5.6. Case studies

In a similar way, the case studies also consist of the testing results from four modified IEEE bus systems including an IEEE-14 bus system, an IEEE-30 bus system, an IEEE-57 bus system and an IEEE-118 bus system. These tests are intended to find the performance of the proposed generation re-dispatch method associated with the sensitivity factor ZS. The results are about the generation variations in re-dispatch stage and the corresponding congestion costs or called security costs. This chapter is focus on the comparison between the PTDF based method and the ZS based method under the same test environment.

#### 5.6.1. Test Description

The test condition is same with the chapter 4 and the details can be found in section 4.5.1. The concept the level of overloading (LO) is continued to be used. The content of the test in this chapter is to calculate the rescheduled generation for each generator and the total congestion cost under a same LO condition by using the PTDF method and the ZS method. Compare the results from both methods by tables and graphs. Due to the dependence of slack bus selection, to simplify the comparison process, both methods only consider the first generator bus as the slack bus. Additionally, tests of both the PTDF method and the ZS method under totally five LO scenarios to discover the performances of both two methods when dealing with different congestion situations, especially when the congestion of the transmission line becomes severer. Then compare the security costs caused by both the PTDF method and the ZS method under total to represent the difference between two methods. It

should be noted that in order to simplify the comparison process, the PTDF method and the ZS method in this comparison both only select the first generator bus as the slack bus.

#### 5.6.2. Modified IEEE-14 Bus System

The modified IEEE-14 bus test system is same with the system in chapter 4, which is showed in Figure 5.2 as follow:



Figure 5.2 The modified IEEE-14 bus system

The parameters of the generators and loads in the test environment are set up the same, which are displayed in Table 4.1. The generation schedule in the unconstrained dispatch is showed in Table 4.2.

After a DC optimal power flow without considering the transmission line constraints, the generation schedule for each generator in the unconstrained dispatch is

obtained, which is as same as the section 4.5.2. Then it is found that the power flow of transmission line 3-4 in the unconstrained dispatch is more than its transmission capacity limit and the level of overloading (LO) of the congested line is 33%. Then use two sensitivity methods to provide solutions for this particular congestion problem. The redispatch generation variation of each generator calculated by two sensitivity-based methods when the LO equals to 33% is obtained and the results are shown in Table 5.1. To make it more straightforward, the information of Table 5.1 is visualized in Figure 5.3. The corresponding congestion costs calculated by two methods are indicated in Figure 5.4. Follow the similar way, other four scenarios with the LO of 1%, 7%, 14%, 21% are tested respectively. Finally, the corresponding security costs in five scenarios calculated by two methods are obtained and displayed in Figure 5.5 by two curves to represent the different outcomes by two methods.

# Table 5.1 Generator Output Variation Due to Re-dispatch by Two Methods in 14-Bus System When LO=33%

Generator	ΔP <sub>G</sub> PTDF slack bus 1 (MW)	ΔP <sub>G</sub> ZS slack bus 1 (MW)
G1	0	-11.52
G2	0	+132.72
G3	-67.31	+38.90
G6	+38.90	-160.10
G8	+28.41	0



Figure 5.3 14-bus system generator output variation in re-dispatch by two methods when LO=33%



Congestion cost comparison when LO=33%

Figure 5.4 14-bus system congestion cost of re-dispatch by two methods when LO=33%



Comparison of security cost by two methods in different LO scenarios

Figure 5.5 Comparison of security cost by two methods in five different LO scenarios in 14-bus test system

From Figure 5.3, the rescheduled generation for each generator accomplished by two methods is obtained. It is indicated that the ZS based method asks generator 1, 2, 3, and 6 to adjust their power outputs while only three generators are involved in re-dispatch in the PTDF based method, which are generator 3, 6 and 8. From Figure 5.4, it illustrates that the security cost due to congestion management by the ZS based method, which is £12287/h, is more than twice the PTDF based method, which is £5527/h. The reason can be found in Figure 5.3 that not only the number of generator involvement in re-dispatch by two methods is different, but also the output volume variations of generator 2 and 6 are large plus the decremental bid price of generator 6 is high. From Figure 5.5, it is found that when the level of overloading of the congested line is 1%, 7% and 14%, the security costs due to congestion management by two methods are nearly the same. When the LO is 21%, the security costs by two methods start to be distinct. When the LO condition is at 33%, the distinction of security cost calculation by two methods become larger. It can be concluded that with severer degree of congestion, the security costs caused by two sensitivity-based methods will become more different.

#### 5.6.3. Modified IEEE-30 Bus System

The modified IEEE-30 bus test system is as same as the test system in chapter 4, which is showed in Figure 5.6. The parameters of the generators and loads in the test environment are set up the same, which are displayed in Table 4.4 and Table 4.5. In the unconstrained dispatch, generation schedules of six generators are showed in Table 4.6.



Figure 5.6 The modified IEEE-30 bus system

After a DC optimal power flow without taking the transmission line constraints into consideration, the generation schedule for each generator in the unconstrained dispatch is obtained, which is as same as the section 4.5.3. Then it is found that the power

flow of transmission line 4-12 in the unconstrained dispatch exceeds its transmission capacity limit and the level of overloading (LO) of the congested line is 19%. Then use the two sensitivity-based methods to find solutions for this congestion problem. The redispatch generation variation of each generator based on two methods when LO equals to 19% is obtained and the results are shown in Table 5.2 and visualized in Figure 5.7. The congestion costs calculated by two methods are indicated in Figure 5.8. In the same way, other four scenarios with the LO of 1%, 4%, 8%, 13% are tested respectively. Therefore, the corresponding security costs in totally five scenarios calculated by two methods are obtained and displayed in Figure 5.9 by two curves to represent the different results by two methods.

Table 5.2 Generator Output Variation Due to Re-dispatch by Two Methods in 30-Bus System When LO=19%

Generator	ΔP <sub>G</sub> PTDF slack bus 1 (MW)	ΔP <sub>G</sub> ZS slack bus 1 (MW)
G1	-69.55	0
G2	0	+271.50
G5	0	+73.97
G8	0	-133.39
G11	0	-212.08
G13	+69.58	0



Comparison of generator output variation when LO=19%

Figure 5.7 30-bus system generator output variation in re-dispatch when LO=19%



#### Congestion cost comparison when LO=19%

Figure 5.8 30-bus system congestion cost of re-dispatch when LO=19%



Comparison of security cost by two methods in different LO scenarios

Figure 5.9 Comparison of security cost by two methods in five different LO scenarios in 30-bus test system

From Figure 5.7, it is indicated that the ZS based method asks generator 2, 5, 8, and 11 to adjust their power outputs while only two generators are involved in re-dispatch in the PTDF based method, which are generator 1 and 13. From Figure 5.8, it illustrates that the security cost due to congestion management by the ZS based method, which is £28378/h, is more than five times as the PTDF based method, which is £5914/h. It is because the number of generator involvement in re-dispatch by the ZS method is doubled the PTDF method, and the incremental/decremental volumes of generators involved in the ZS method are larger than the PTDF method plus the incremental/decremental bid prices of generators involved in the ZS method are higher than the ones in the PTDF method. According to the Figure 5.9, it is showed that with higher level of overloading of the congested line, the security costs distinction based on two sensitivity-based methods become larger. It should be noted from Table 5.2 that when dealing with the congestion at LO equals to 19%, the PTDF-based method involves G1 and G13 into re-dispatch stage while both of them are not asked to change their generators to change their outputs while

they are not involved in re-dispatch stage in PTDF-based method. It is very clear to say that under this congestion scenario, two sensitivity-based method have totally different idea to re-dispatch generation output. Consequently, the security cost in ZS-based method is around five times to the PTDF-based method, which can be seen from Figure 5.8. Although, in the first four congestion scenarios, the security cost values in the ZS-based method are all more or less three times to the PTDF-based method, the sudden jump at LO equals to 19% is still obviously observed. The direct reason for this has been explain as the ZS-based method involves more generation power in re-dispatch. To dig in, the deeper reason may be obtained from Appendix Table B. It is found that when selecting bus 1 as the slack bus, two methods calculated totally two different sensitivity factors for six generator buses respectively because two types of sensitivity factors are calculated based on different algorithms. As a result, the ZS-based method under this particular circumstance, not only it may produce more security cost but also the instability.

#### 5.6.4. Modified IEEE-57 Bus System

The modified IEEE-57 bus test system is same with the system in chapter 4, which is showed in Figure 5.10. The parameters of the generators and loads in the test environment are set up the same, which are displayed in Table 4.8 and Table 4.9. In the unconstrained dispatch, generation schedules of seven generators are showed in Table 4.10.



Figure 5.10 The modified IEEE-57 bus system

After a DC optimal power flow without transmission line constraints, the generation schedule for each generator in the unconstrained dispatch is obtained, which is as same as in section 4.5.4. Then it is found that the power flow of transmission line 7-29 in the unconstrained dispatch is above its transmission capacity limit and the level of overloading (LO) of the congested line is 8%. Then the two sensitivity-based methods are utilized to find out solutions for this transmission congestion problem. The generation output variation of each generator in rescheduling based on two sensitivity methods when LO equals to 8% is obtained and the results are shown in Table 5.3 and Figure 5.11. The security costs caused by two methods are indicated in Figure 5.12. Similarly, tests of other four scenarios with the LO of 1%, 3%, 5%, 7% are accomplished respectively. Therefore,

the corresponding security costs in totally five scenarios in two methods are obtained and displayed in Figure 5.13 by two curves to represent the different results by two methods.

# Table 5.3 Generator Output Variation Due to Re-dispatch by Two Methods in 57-Bus System When LO=8%

	$\Delta \boldsymbol{P}_{\boldsymbol{G}}$	$\Delta P_{G}$
Gen	PTDF slack bus 1	ZS slack bus 1
	( <b>MW</b> )	( <b>MW</b> )
G1	+170.47	-129.50
G2	0	-70.36
G3	+0.21	+125.53
G6	-208.98	+45.36
G8	0	+185.41
<b>G9</b>	0	0
G12	+38.31	-156.44



Comparison of generator output variation when LO=8%

Figure 5.11 57-bus system generator output variation in re-dispatch when LO=8%



Congestion cost comparison when LO=8%

Figure 5.12 57-bus system congestion cost of re-dispatch when LO=8%



Comparison of security cost by two methods in different LO scenarios

Figure 5.13 Comparison of security cost by two methods in five different LO scenarios in 57bus test system

From Figure 5.11, it is noted that the ZS based method asks all the generators in the system except generator 9 to adjust their power outputs while only three generators are involved in re-dispatch in the PTDF based method, which are generator 1, 6 and 12. Figure 5.12 illustrates that the security cost due to congestion management by the ZS based method, which is £26755/h, is about £10000/h more than the PTDF based method, which is £16056/h. The reason is the number of generator involved in the re-dispatch by the ZS method is doubled the number in the PTDF method. From Figure 5.13, it is found that when the level of overloading of transmission line 7-29 is 1%, the security costs due to congestion management by two methods are roughly close and in fact the security cost caused by the PTDF based method is slightly higher than the ZS based method. When the LO is 3%, the security costs caused by the two methods start to be distinguished. Based on the results of the LO interval between 3% and 8%, it is showed that with severer degree of transmission congestion occurs on the congested line, the security cost distinction of two sensitivity-based methods becomes larger.

#### 5.6.5. Modified IEEE-118 Bus System

The modified IEEE-118 bus test system is same with the test system in chapter 4, which is showed in Figure 5.14. The parameters of the generators and loads in the test environment are set up as the same in chapter 4, which are displayed in Table 4.12 and Table 4.13. Generation schedule in the unconstrained dispatch is showed in Table 4.14.



Figure 5.14 The modified IEEE-118 bus system

After running a DC optimal power flow without considering transmission line constraints, the generation schedule for each generator in the unconstrained dispatch is obtained, which is as same as the information in section 4.5.5. Then it is found that the power flow of transmission line 23-24 in the unconstrained dispatch is over its transmission capacity limit and the level of overloading (LO) of the congested line is 99%,

which can be regarded as the transmission line is nearly blocked. Then let the two sensitivity-based methods figure out solutions for this serious transmission congestion issue. The generation output variation of each generator in re-dispatch based on two sensitivity methods when LO equals to 99% is obtained and the results are shown in Table 5.4 and Figure 5.15. The security costs caused by two re-dispatch methods are indicated in Figure 5.16. At last, tests of other four scenarios with the LO of 1%, 25%, 50%, 75% are accomplished respectively. Therefore, the corresponding security costs in totally five scenarios of two methods are obtained and displayed in Figure 5.17 by two curves to represent the different results by two methods.

Table 5.4 Generator Output Variation Due to Re-dispatch by Two Methods in 118-Bus System When LO=99%

	$\Delta P_{G}$	$\Delta \boldsymbol{P}_{\boldsymbol{G}}$
Gen	PTDF Slack bus 12	ZS slack bus 12
	( <b>MW</b> )	( <b>MW</b> )
G12	+228.83	-71.17
G19	0	-241.47
G24	-296.55	0
G27	0	-300
G40	0	0
G49	+62.65	+180.18
G59	+5.09	+208.96
G61	0	+223.50
G70	0	0
G85	0	0
G99	0	0
G110	0	0



Figure 5.15 118-bus system generator output variation in re-dispatch when LO=99%



Congestion cost comparison when LO=99%

Figure 5.16 118-bus system congestion cost of re-dispatch when LO=99%



Comparison of security cost by two methods in different LO scenarios

Figure 5.17 Comparison of security cost by two methods in five different LO scenarios in 118bus test system

From Figure 5.15, it is showed that the ZS based method requires generator 12, 19, 27, 49, 59 and 61 to change their power outputs while only four generators which are generator 12, 24, 49 and 59 are involved in re-dispatch when using the PTDF based method. From Figure 5.16, it is illustrated that the security cost due to congestion management by the ZS based method, which is £43939/h, is more than the doubled security cost by the PTDF based method, which is £19615/h. It is due to the involved generators in re-dispatch by the ZS method is more than the ones in the PTDF method. From Figure 5.17, it is indicated that when the level of overloading of the congested line 23-24 is 1%, the security costs due to congestion management by two methods are similarly approximated but in fact, the value in the ZS-based method is doubled the PTDF-based method. From the results of the LO interval between 25% and 99%, it is demonstrated that with greater level of transmission congestion occurs on the congested line 23-24, the security costs distinction of two sensitivity-based methods becomes more distinct, with the values on the ZS-based curve all approximately twice the values on the

PTDF-based curve. The explanation can be seen inside the Appendix Table D. It is found that when selecting bus 12 as the slack bus, the two methods calculated totally different sensitivity factors for twelve generator buses respectively because the two types of sensitivity factors are calculated based on different algorisms. As a result, the ZS-based method may not be selected as the right way to re-dispatch generation under this particular circumstance since it may produce more security cost.

#### 5.6.6. Discussion

In this case study, performances of the two sensitivity-based generation redispatch methods are tested under four different test systems. Experimental results prove that the ZS based method can implement the generation re-dispatching with fulfilling the objective function and the three constraints.

However, the ZS method still has some disadvantages. First of all, the ZS based method is also dependent on the slack bus selection, just like the PTDF based method. Therefore, it will make the ZS method inaccurate if only consider one generator bus as the slack bus. However, if choose each the generator bus in the system as the slack bus respectively, the problem solving process will become time-consuming. Secondly, compared with the PTDF method, generation re-dispatching by the ZS method involves more generators to adjust their initial generation output schedules. In the case of 14-bus system, the comparison of the ratio of involved generators between the ZS method and the PTDF method is 4:3; in the case of 30-bus system, the ratio is 6:4. Due to the number of generator involved in re-dispatching by the ZS method is more than the PTDF method, the corresponding security cost or called congestion cost caused by congestion management in the ZS method is more expensive than the PTDF method. The comparisons of the security costs between two methods in all the test systems are the evidences to prove the above conclusion. In the case of 14-bus system, the security cost

in the ZS method is £12287/h while in the PTDF method the cost is £5526/h. The PTDFbased method only produced nearly 50% security cost of the ZS-based method; in the case of 30-bus system, ZS method causes £28377/h congestion cost and PTDF method causes £5914/h congestion cost, which means the PTDF-based method only results in 21% security cost of the ZS-based method; in the case of 57-bus system, the security cost contrast is £26755/h to £16056/h. The security cost value in the PTDF-based method is only approximately 60% of the ZS-based method; in the case of 118-bus system, the cost contrast is £43939/h to £19615/h, which indicates that the security cost value the PTDFbased method produced is only around 45% of the value in the ZS-based method It is also found that with the increasing level of overloading of the congested line, the distinction of security costs caused by two sensitivity-based methods become more distinguished.

#### 5.7. Summary

In this chapter, the proposed generation re-dispatch method associated with the bus impedance matrix sensitivity is described in details including the methodology of the proposed re-dispatch method with the sensitivity factor and the algorithm of the sensitivity factor. The mathematic model of the proposed method including an objective function and three constraints one of which contains the bus impedance matrix sensitivity are given. Case studies consist of experimental results from four modified IEEE testing systems including a 14-bus, a 30-bus, a 57-bus and a 118-bus systems are implemented. Comparison between the bus impedance matrix sensitivity method and the power transfer distribution factor method indicates that the ZS based method is dependent on slack bus selection as well; number of generator involvement in re-dispatch by the ZS method is higher than the PTDF method which leads to the security cost in the ZS method is higher than the PTDF method when dealing with a same level of overloading congestion problem; with the severer LO situation, the security cost difference between the ZS method and the PTDF method will become larger.

## **Chapter 6**

## Security Cost Allocation in Deregulated Electricity market with Power Transfer Distribution Factor

#### 6.1. Introduction

The extra cost due to transmission congestion management through re-dispatching generation is called the congestion cost or the security cost. It is the responsibility of the system operator to balance the congestion management and the security cost. Since in the deregulated electricity market, the system operator does not have any source of income, the security cost needs to be recovered from market participants. Therefore the security cost allocation become very essential to decentralize the cost balancing problem in the electricity market. As mentioned in section 4.2.2, the security cost of the electricity market is originated from the incremental/decremental bids mechanism. Generators take part in the bid submission so as to help the system operator to relieve the congestion and the system operator will pay for generators the extra cost of the re-dispatches. Since generators have already offered assistance to manage congestion, the security cost will only be allocated to the demand side. The allocation principle is according to the power flow contribution of each load to the congested transmission line. The power flow contribution of each load will be calculated based on the power transfer distribution factor (PTDF) which was introduced in chapter 4. The proposed allocation method will be tested in four test systems under the same test conditions as the case studies of chapter 4 to allocate the already calculated security costs to the loads according to their power flow contributions to the target line.

In this chapter, conventional security cost allocation schemes will be described in section 6.2. Then the mathematic model of the proposed allocation method with PTDF will be given in section 6.3. Section 6.4 is the case study including experimental results of four modified IEEE test systems which are an IEEE-14 bus system, an IEEE-30 bus system, an IEEE-57 bus system and an IEEE-118 bus system. Section 6.5 is the summary.

#### 6.2. Conventional Security Cost Allocation Schemes

In chapter 3, various types of congestion management schemes are discussed in details. In fact, they not only just manage transmission congestion but also intend to balance the extra cost caused by congestion relief. Different electricity market models apply different schemes on security cost allocation.

#### 6.2.1. Security Cost Allocation in Vertically Integrated Power Utility

In the traditional vertically integrated power utility, transmission congestion is resolved by re-dispatching generation. The cost of re-dispatching action which is considered as the security cost will be put on the shoulder of the vertically integrated utility [110].

#### 6.2.2. Security Cost Allocation in Pool Electricity Market

In the pool-based electricity market, transmission congestion is relieved by generation re-dispatching according to market-based rules. The corresponding security cost is allocated to the loads by using uniform marginal price, locational marginal price and zonal price [107, 122]. As discovered in section 3.3, uniform marginal price and locational marginal price is different on the security cost allocation. Uniform marginal price allocates the security cost based on a non-discrimination principle which means all the loads in the system will shoulder the extra cost uniformly. The demand side customers are asked to equally share the security cost without considering their impact on the

congestion problem. By contrast, locational marginal price allocates the security cost with taking load location into consideration. As described in section 3.2.1.2, the locational marginal price is consisted of three components: marginal cost component, congestion cost component and loss cost component. If there is no congestion in the system, the locational marginal price has only the marginal cost component, which means prices on all nodes are identical. Once congestion occurs in the system, the congestion cost component will be added into nodal price calculation, which leads to different locational prices on nodes. Since the loads are charged based on their nodal prices, the security cost is already allocated to the loads through electricity price settlement. The idea of locational marginal price reflects the locational factor of load, in other words, it considers the impact of each load to the congestion. However, this method will trigger merchandising surplus issue which is due to the net income of the system operator [35]. Adopting the locational marginal price as the pricing scheme, the congestion will result in a revenue of the system operator. With more severe congestion situation, the revenue will become get larger. This revenue can be distributed back to the market participants or paid off to the holders of financial instruments [123].

#### 6.2.3. Security Cost Allocation in Bilateral Electricity Market

In the bilateral electricity market, contract between generator and load does not think of congestion and the corresponding security cost. Hence the incurred security cost will be allocated to market participants. Firstly, the total security cost will be divided into generation portion and load portion on a half-half basis [35]. Secondly, each cost portion will be allocated to generators or loads based on their power energy generations/consumptions [107]. The process of the security cost allocation in bilateral market is according to the following equations [35]:

$$SC_{gens} = SC_{loads} = 0.5 \times SC_{total} \tag{6.1}$$

$$SC_{G_i} = \frac{P_{G_i}}{\sum_{i=1}^{N_G} P_{G_i}} \times SC_{gens}$$
(6.2)

$$SC_{L_j} = \frac{P_{L_j}}{\sum_{j=1}^{N_L} P_{L_j}} \times SC_{loads}$$
(6.3)

Where

*SC*total : The total security cost

SC<sub>gens</sub> : The security cost portion for generators

SCloads : The security cost portion for loads

 $SC_{G_i}$ : The security cost allocated to generator *i* 

 $P_{G_i}$ : The MW power generation of generator *i* 

 $SC_{L_i}$ : The security cost allocated to load j

 $P_{L_i}$ : The MW power consumption of load j

 $N_G$ : The number of generators in the system

 $N_L$ : The number of loads in the system

Equation (6.1) expresses the total security cost is divided equally into generator portion and load portion. Equation (6.2) and equation (6.3) show that the security cost allocated to each generator or each load is based on its power energy generation or consumption.

Nevertheless, such security cost allocation method cannot reflect the impact of each market participant to the transmission congestion. This is unfair for the participant especially the one who contributes less to the congestion to shoulder the security cost. A method should be proposed to let the security cost allocate to the participant who is more responsible for the transmission congestion rather than spread equally over all the participants [124]. Therefore the power flow contribution of each market participant to the congestion problem should be calculated in the first place.

### 6.3. Proposed Security Cost Allocation Method with Power Transfer Distribution Factor

The proposed security cost allocation method with PTDF factor will be described in details including the mechanism and the mathematical model of the method.

### 6.3.1. Mechanism of the Proposed Security Cost Allocation Method with Power Transfer Distribution Factor

As mentioned in section 4.3, the power transfer distribution factor can figure out the relationship between the active power flow of a particular transmission line from bus j to bus k and the active power import or export at bus i. With the help of the PTDF factor, the power flow on a particular transmission line can be regarded as the consequence of the combined actions of all the generators and loads in the system. That is to say it is possible to find the contribution of each generator or each load to the power flow of the congested line if there is a congestion situation occurs in the power system.

In the proposed method, generators do not bear the security cost, therefore, all the loads in the system are requested to shoulder all of it. As a result, the contributions of all the generators to the power flow of the target transmission line should only be used to calculate the MW contributions of all the loads to the power flow on the congested line, which is the total power flow MW values on the congested line minus the power flow MW values associated with generators contribution. Once the MW portion on the congested line associated with all the loads is obtained, combined with the power flow specifically associated with each load, the contribution of each load to the power flow on the congested transmission line can be captured. Finally, the produced power flow contribution of each load times the total security cost goes to the allocated security cost to each load.

# 6.3.2. Mathematical Model of the Proposed Security Cost Allocation Method with Power Transfer Distribution Factor

The following equations are the mathematical model which is aim to express the above mechanism of the proposed security cost allocation method with PTDF.

Equation (6.4) expressed the total MW power flow of the congested line is consisted of generators contribution plus loads contribution. Combined with the definition of the PTDF, which is contained in section 4.3, the relationship between the active power flow change quantity of a particular transmission line from bus j to bus k and the active power injection change quantity at bus i, the active power flow quantity of a particular transmission line in the system can be deemed as the result of joint action of all buses no matter they are injecting power into the system or withdrawing power out of the system. The above explanation can be further expressed as that each bus's contribution to the power flow of a particular transmission line can be calculated as its power input or output to the system times its sensitivity factor to the particular transmission line. It should be noted that the contribution of one particular bus to one particular transmission line could be either positive or negative. If the contribution is positive, it means this bus helps to increase the power flow of the transmission line. If the contribution is negative, it means this bus helps to decrease the power flow of the transmission line. Therefore, the power

flow of a particular transmission line j-k consists of such positive contributions and negative contributions from all the buses in the system.

$$P_{j-k} = P_{j-k}^G + P_{j-k}^L$$
(6.4)

Where

 $P_{j-k}$ : The power flow MW value of the target transmission line *j*-*k*  $P_{j-k}^{G}$ : The power flow MW value on line *j*-*k* associated with all the generators  $P_{j-k}^{L}$ : The power flow MW value on line *j*-*k* associated with all the loads

Equation (6.5) indicates that the total MW contribution of generators is the sum of the power flow contribution of each generator.

$$P_{j-k}^{G} = \sum_{i=1}^{N_{G}} PTDF_{G_{i}}^{j-k} \cdot P_{G_{i}}$$
(6.5)

Where

 $PTDF_{G_i}^{j-k}$ : The PTDF of the line *j-k* respect to generator bus *i* 

 $P_{G_i}$ : The MW power energy injected by generator *i* 

Equation (6.6) shows that total MW contribution of load is the sum of the power flow contribution of each load.

$$P_{j-k}^{L} = \sum_{i=1}^{N_{L}} PTDF_{L_{i}}^{j-k} \cdot P_{L_{i}}$$
(6.6)

Where

 $PTDF_{L_i}^{j-k}$ : The PTDF of the line *j*-*k* respect to load bus *i* 

 $P_{L_i}$ : The MW power energy withdrawn by load *i* 

Based on the equation (6.5) and (6.6), equation (6.4) can be further extended as follow:

$$P_{j-k} = \sum_{i=1}^{N_G} PTDF_{G_i}^{j-k} \cdot P_{G_i} + \sum_{i=1}^{N_L} PTDF_{L_i}^{j-k} \cdot P_{L_i}$$
(6.7)

As mentioned before, generators will not be involved in the security cost allocation, so security cost allocation here is only based on the contributions of loads to the congested transmission line. Therefore equation (6.6) is only the equation to be considered into security cost allocation calculation.

Equation (6.8) tells that the MW power flow contribution of each load is equals to the PTDF of the load times the power energy extraction of the load. Such equation is introduced to help to find out each load's contribution percentage to the total power contributions of loads to the congested transmission line.

$$P_{j-k}^{L_i} = PTDF_{L_i}^{j-k} \cdot P_{L_i}$$
(6.8)

Where

 $P_{j-k}^{L_i}$ : The power flow MW value associated with load *i* 

Equation (6.9) is the expression of the contribution percentage of each load.

$$Ctb_{L_{i}} = \frac{P_{j-k}^{L_{i}}}{P_{j-k}^{L}} \times 100\% = \frac{PTDF_{L_{i}}^{j-k} \cdot P_{L_{i}}}{\sum_{i=1}^{N_{L}} PTDF_{L_{i}}^{j-k} \cdot P_{L_{i}}} \times 100\%$$
(6.9)

Where

 $Ctb_{L_i}$ : The power flow contribution percentage of load *i* to the congested line

Equation (6.10) is the calculation of the security cost allocated to each load.

$$CA_{L_i} = Ctb_{L_i} \times SC_{total} \tag{6.10}$$

Where

 $CA_{L_i}$ : The allocated security cost to the load *i* 

 $SC_{total}$ : The total security cost due to congestion management
#### 6.4. Case Studies

The case studies in this chapter follow the rule of the former two chapters, which is using four modified IEEE bus systems including an IEEE-14 bus system, an IEEE-30 bus system, an IEEE-57 bus system and an IEEE-118 bus system to test the performance of the proposed security cost allocation method with PTDF. In order to get a comparison of the method performance, a pro rata allocation method based on power volume will be introduced in the tests.

#### 6.4.1. Test Description

Rules and environment of the test are same with the chapter 4 and the details can be found in section 4.5.1. The test is designed for finding the outcomes of the proposed security cost allocation method with PTDF factor. The security cost will be allocated to each load by the above method and the results will be presented by tables and graphs. To get a comparison associated with the PTDF method, a pro rata based allocation method will be brought to contrast to the PTDF method. Such pro rata method is an evolution of the method mentioned in section 6.2.3. The variation is that the total security cost will be only allocated to the loads according to their power volume consumptions. Equation (6.11) describes the principle of the pro rata method as follows:

$$CA_{L_i} = \frac{P_{L_i}}{\sum_{i=1}^{N_L} P_{L_i}} \times SC_{total}$$
(6.11)

The value of the security cost in each testing case is original from the test results in section 5.5. On one hand, it is to guarantee the integrity of the proposed congestion management scheme because of the proposed method is aim to not only re-dispatch generation but also to balance the security cost attributed to the congestion management. On the other hand, choosing the results from chapter 5 is intended to pave the way for the next chapter which will introduce the impedance matrix based method to allocate the security cost. In this way, it is convenient to compare the outcomes by different security cost allocation schemes.

Since the PTDF based method is dependent on the slack bus selection, selecting different generator bus as slack bus might incur different experimental outcomes. To simplify the experiment, case studies are mainly focus on two comparisons. One is the comparison of different experimental results based on different PTDF values due to different slack bus selections. Another one is the outcome comparison between the pro rata method and the PTDF method when the first generator bus selected as the slack bus. The test description is visualized by Figure 6.1 as follows:



Figure 6.1 Test description for chapter 6

## 6.4.2. Modified IEEE-14 Bus System

The modified IEEE-14 bus test system is exactly same with the system in chapter 4 and 5, which is showed in Figure 6.2 as follow:



Figure 6.2 The modified IEEE-14 bus system

The parameters of the generators and loads are set up the same, which are displayed in Table 4.1. Transmission congestion still occurs on the target branch 3-4. There are 5 generators and 9 loads in the system. The security costs are picked up from the experimental results in section 5.5.2 which are displayed in Figure 6.3.



Security Cost with Different Slack Buses

Figure 6.3 Security cost settings in the 14-bus test system

## 6.4.2.1. Comparison between Security Cost Allocation with Different Slack Bus Selections in the 14-bus Test System

After used the proposed security cost allocation method with PTDF, the security cost allocated to each load in the system is presented in Table 6.1 and Figure 6.4 as follows:

Table 6.1 Security Cost Allocated to I	ach Load with	Different Slack I	Buses in the 14-
bus Test System			

(£/h)	L4	L5	L7	L9	L10	L11	L12	L13	L14
slack 1	642.13	437.46	726.49	585.66	685.43	1184.27	408.55	412.57	444.06
slack 2	626.62	457.46	715.52	579.94	681.58	1189.77	414.38	417.70	443.73
slack 3	572.15	527.59	676.99	559.86	668.05	1209.00	434.83	435.70	442.56
slack 6	1574.05	-762.41	1385.81	929.36	917.03	855.30	58.75	104.66	464.17
slack 8	-384.74	1759.65	0.00	206.95	430.25	1546.82	794.01	751.88	421.91



Security Cost Allocation with Different Slack Buses

Figure 6.4 Security cost allocated to each load with different slack buses in the 14-bus test system



Max, Min and Average Cost Values of Each Load in 14-bus System

#### Figure 6.5 Maximum, minimum and average security cost values of each load in 14-bus system

Figure 6.3 represents the security costs originated from the experimental results in chapter 5. It is indicated that the security costs based on different slack bus selections are very close at around £5526/h.

Figure 6.4 compares the security cost allocation to each load with different slack buses. It is found that when select the generator bus 1, 2 and 3 as the slack bus respectively, the allocation results are similar. However, when bus 6 or bus 8 is regarded as the slack bus, the results of security cost allocation show quite a distinction. In the case of slack bus 6, load 4, 7, 9, 10 and 14 shoulder more security cost compared with the first three cases and the cost allocated to load 11, 12, 13 and 14 is lower than the first three cases. In the case of slack bus 8, load 5, 11, 12 and 13 accept more security cost and load 7, 9 and 10 shoulder less cost compared with the first three cases. It should be noted that load 5 in slack bus 6 and load 4 in slack bus 8 are allocated negative values of security cost, which means such two loads both have negative contributions on power flow of congested line, in other words, they are helping to relieve the congestion. As mentioned in section 6.3.2, the PTDF-based method is trying to find out each bus's real contribution to one particular transmission line's power flow no matter it is positive or negative. Therefore, based on these real contributions of loads, the system operator could take measures to do appropriate security cost allocation to loads. Here, as bus 6 is selected as the slack bus, the PTDF-based method deems load 5 should be allocated negative security cost since it helps to relieve the congestion and as bus 8 is selected as the slack bus, load 4 is considered as the one who helps to relieve the congestion. The system operator should avoid allocating security cost to such loads and even reward them. From the above test results, it should be known that the PTDF based method is indeed suffered the drawbacks of complexity and inaccuracy due to dependence on slack bus selection.

Figure 6.5 shows the maximum, the minimum and the average security cost values of each load in 14-bus system under different slack bus selections. From the figure, it can

be found that load 5 has the biggest distance between the maximum value at £1759/h and the minimum value at -£762/h. The range is £2521/h. Load 14 has the smallest range between the maximum value at £464/h and the minimum value at £421/h. The range is £41/h. The black curve shows the average security cost values of each load under different slack bus selections. From the curve, it can be noted that load 11 has the highest average security cost allocation value at £1197/h and load 12 has the lowest value at £422/h.

## 6.4.2.2. Comparison between Pro Rata Method and PTDF Method in the 14bus Test System

Then the pro rata based method will be implemented to compare with the PTDF method when generator bus one is selected as the slack bus. The comparison will be illustrated in Table 6.2 and Figure 6.6 as follows:

# Table 6.2 Comparison between Two Methods on Security Cost Allocation in the 14-bus Test System

(£/h)	L4	L5	L7	L9	L10	L11	L12	L13	L14
Pro rata	552.66	552.66	663.19	552.66	663.19	1215.86	442.13	442.13	442.13
slack bus 1	642.13	437.46	726.49	585.66	685.43	1184.27	408.55	412.57	444.06



Security Cost Allocation by Two Methods

Figure 6.6 Comparison between two methods on security cost allocation in the 14-bus test system

From Figure 6.6, it is indicated that security cost allocation results by pro rata method and PTDF method seems very similar. Except load 4, 5 and 7 shoulder a little different security cost, the other loads are allocated approximately the same amount of cost. It is unable to indicate the theoretical advantage of the PTDF method compared to the Pro rata method. More discoveries about the comparison between two methods will be obtained in the next few systems.

### 6.4.3. Modified IEEE-30 Bus System

The modified IEEE-30 bus test system is same with the system in chapter 4 and 5, which is shown in Figure 6.7 as follow:



Figure 6.7 The modified IEEE-30 bus system

The parameters of the generators and loads are set up in the same way as those the former two chapters, which are displayed in Table 4.4 and Table 4.5 respectively. There are 6 generators and 18 loads in the system. Transmission congestion are still set on the target branch 4-12. The security costs are taken from the experimental results in section 5.5.3 which are displayed in Figure 6.8.



Security Cost With Different Slack Buses

Figure 6.8 Security cost settings in the 30-bus test system

## 6.4.3.1. Comparison between Security Cost Allocation with Different Slack Bus Selections in the 30-bus Test System

By implementing the proposed PTDF based security cost allocation method, the security cost allocated to each load with different slack buses is presented in Table 6.3 and Figure 6.9 respectively as follows:

## Table 6.3 Security Cost Allocated to Each Load with Different Slack Buses in the 30-bus Test System

(£/h)	L3	L4	L7	L10	L12	L14	L15	L16	L17	L18	L19	L20	L21	L23	L24	L26	L29	L30
slack 1	-10.03	-16.44	30.30	359.03	782.51	601.21	561.51	605.55	506.67	281.85	415.51	232.17	183.93	465.85	405.58	235.65	102.47	170.78
slack 2	-13.75	-21.45	24.90	358.04	788.71	605.52	565.14	608.74	507.00	283.10	416.65	232.57	183.51	467.85	405.38	233.74	100.66	167.76
slack 5	-24.27	-35.63	9.62	355.23	806.27	617.70	575.41	617.79	507.94	286.61	419.91	233.70	182.31	473.53	404.81	228.34	95.55	159.24
slack 8	-37.87	-53.96	-10.12	351.59	828.97	633.45	588.69	629.48	509.16	291.16	424.11	235.15	180.77	480.86	404.07	221.37	88.94	148.23
slack 11	-331.69	-449.83	-436.67	273.11	1319.25	973.69	875.61	882.09	535.52	389.37	514.93	266.63	147.46	639.29	388.11	70.62	-53.79	-89.64
slack 13	458.91	615.39	711.08	484.29	0.00	58.18	103.59	202.37	464.60	125.11	270.56	181.93	237.10	212.99	431.05	476.24	330.26	550.43



## Security Cost Allocation with Different Slack Buses

Figure 6.9 Security cost allocated to each load with different slack buses in the 30-bus test system



Figure 6.10 Maximum, minimum and average security cost values of each load in 30-bus system

Figure 6.8 displays the security costs originated from the experimental results in chapter 5 as the basic security cost settings in this case. It is indicated that the security costs caused by different slack bus selections are all approximated at £5914/h.

Figure 6.9 compares the security cost allocation to each load under different slack bus selections. It is found that selecting the first four generator buses respectively as the slack bus, the allocation results are close. However, when bus 11 or bus 13 is chosen as the slack bus, the outcome of security cost allocation becomes significantly different. In the case of slack bus 11, load 12, 14, 15, 16, 17, 18, 19, 20 and 23 undertake more security cost compared with the first four cases and the cost allocated to load 10, 21, 24 and 26 is lower than the first four cases. In the case of slack bus 13, load 3, 4, 7, 10, 21, 24, 26, 29 and 30 accept more security cost and load 14, 15, 16, 17, 18, 19, 20 and 23 are allocated less cost when compared with the first four cases. It is noticeable that load 3 and 4 in the first five case, load 7 in slack bus 8 and slack bus 13 cases and load 29 and 30 in slack bus 11 are allocated negative values of security cost, which means the loads at here have contributed to relieving congestion. The PTDF-based method detects each bus's real contribution to the congestion no matter it is positive or negative. Here these loads are considered as to help to relieve the congestion. In theory, they should be rewarded by the system operator rather than shoulder the extra cost. From the above results, it is obtained again that the PTDF based method could result in different security costs and is dependent on the selection of slack bus.

Figure 6.10 shows the maximum, the minimum and the average security cost values of each load in 30-bus system under different slack bus selections. From the figure, it can be found that load 7 has the biggest range between the maximum value at  $\pounds$ 711/h and the minimum value at  $\pounds$ 436/h. The range is  $\pounds$ 1147/h. Load 24 has the smallest range between the maximum value at  $\pounds$ 431/h and the minimum value at  $\pounds$ 388/h. The range is  $\pounds$ 43/h. The black curve shows the average security cost values of each load under different

slack bus selections. From the curve, it can be noted that load 12 has the highest average security cost allocation value at  $\pm$ 754/h and load 4 has the lowest value at  $\pm$ 6/h except those zeros.

# 6.4.3.2. Comparison between Pro Rata Method and PTDF Method in the 30bus Test System

The pro rata based method is used to offer a comparison with the PTDF method with selecting generator bus one as the slack bus. The comparison will be represented in Table 6.4 and Figure 6.11 respectively as follows:

Table 6.4 Comparison between Two	o Methods on Security	Cost Allocation in the 30-
bus Test System		

(£/h)	L3	L4	L7	L10	L12	L14
Pro rata	208.73	278.31	347.89	417.46	417.46	347.89
slack bus 1	-10.03	-16.44	30.30	359.03	782.51	601.21
	L15	L16	L17	L18	L19	L20
Pro rata	347.89	417.46	487.04	208.73	347.89	208.73
slack bus 1	561.51	605.55	506.67	281.85	415.51	232.17
	L21	L23	L24	L26	L29	L30
Pro rata	208.73	347.89	417.46	347.89	208.73	347.89
slack bus 1	183.93	465.85	405.58	235.65	102.47	170.78



Security Cost Allocation by Two Methods

Figure 6.11 Comparison between two methods on security cost allocation in the 30-bus test system

Figure 6.11 visualizes the comparison of security cost allocation results between pro rata method and PTDF method. It is found that the interval of two curves from load 17 to load 30 have similar tendency of reaction. The difference between two curves is mainly at load 3, 4, 7, 12, 14, 15, 16, 23, 26, 29 and 30. Loads 3, 4, 7, 26, 29 and 30 in the PTDF method are allocated lower security cost than in pro rata method because PTDF method can be figured out the contribution of these three loads to the transmission congestion is small. Similarly, since the contributions of load 14, 15, 16, 23 and especially 12 to the congestion on the line 4-12 are large, the security cost allocated to those loads are higher than other loads.

## 6.4.4. Modified IEEE-57 Bus System

Test system in this case is used exactly the same system with chapter 4 and 5 respectively, which is shown in Figure 6.12 as follow:



Figure 6.12 The modified IEEE-57 bus system

The parameters of the generators and loads are set up in the same way as in the former two chapters, which are displayed in Table 4.8 and Table 4.9 respectively. There are 7 generators and 35 loads totally in the system. Transmission congestion are still set on the target branch 7-29. The security costs are taken from the experimental results in section 5.5.4 which are displayed in Figure 6.13.



Security Cost With Different Slack Buses

Figure 6.13 Security cost settings in the 57-bus test system

## 6.4.4.1. Comparison between Security Cost Allocation with Different Slack Bus Selections in the 57-bus Test System

By implementing the proposed PTDF based security cost allocation method, the security cost allocated to each load with different slack bus selections is obtained and shown in Table 6.5 and Figure 6.14 respectively as follows:

## Table 6.5 Security Cost Allocated to Each Load with Different Slack Buses in the 57-bus Test System

(£/h)	L5	L10	L13	L14	L15	L16	L17	L18	L19	L20	L23	L25	L27	L28	L29	L30	L31	L32
slack 1	-339.84	21.61	19.71	27.63	23.70	8.60	11.29	-99.07	54.14	239.64	293.39	1215.32	1579.82	1205.49	3247.97	687.52	585.00	286.17
slack 2	-314.51	29.89	25.11	32.85	34.85	14.25	25.65	-79.25	61.67	251.70	295.44	1202.12	1552.42	1183.74	3188.23	680.55	580.38	285.47
slack 3	-244.36	52.87	40.09	47.33	65.81	29.94	65.51	-24.31	82.59	285.25	301.26	1165.93	1476.98	1123.83	3023.66	661.47	567.77	283.62
slack 6	53.12	150.30	103.63	108.75	197.10	96.45	234.52	208.67	171.32	427.51	325.89	1012.43	1157.03	869.74	2325.69	580.54	514.30	275.78
slack 8	4.92	134.51	93.33	98.80	175.83	85.67	207.14	170.92	156.94	404.46	321.90	1037.31	1208.88	910.92	2438.80	593.65	522.97	277.05
slack 9	-305.42	32.89	27.06	34.74	38.89	16.30	30.85	-72.11	64.41	256.11	296.25	1197.59	1542.84	1176.12	3167.30	678.18	578.83	285.27
slack 12	-394.20	3.79	8.09	16.40	-0.31	-3.56	-19.62	-141.66	37.91	213.61	288.86	1243.28	1638.18	1251.84	3375.31	702.26	594.73	287.58

(£/h)	L33	L35	L38	L41	L42	L43	L44	L47	L49	L50	L51	L52	L53	L54	L55	L56	L57	
slack 1	429.25	369.15	133.37	82.73	84.69	28.03	108.18	227.08	78.13	87.66	36.89	2027.93	1295.82	1275.05	411.37	107.46	205.37	
slack 2	428.20	375.32	136.16	92.54	91.53	36.17	111.55	236.50	82.19	94.42	44.82	1993.20	1274.92	1260.48	416.57	113.77	215.29	
slack 3	425.43	392.58	143.97	119.77	110.52	58.74	120.94	262.73	93.49	113.23	66.83	1897.57	1217.42	1220.52	431.16	131.33	242.89	
slack 6	413.67	465.73	177.06	235.24	191.06	154.45	160.78	373.95	141.37	192.97	160.17	1492.00	973.55	1051.02	493.01	205.77	359.92	
slack 8	415.58	453.88	171.70	216.53	178.01	138.94	154.33	355.93	133.61	180.05	145.05	1557.72	1013.07	1078.49	482.99	193.71	340.96	
slack 9	427.91	377.63	137.20	96.10	94.01	39.11	112.79	239.96	83.68	96.89	47.69	1981.06	1267.64	1255.47	418.54	116.08	218.92	
slack 12	431.37	355.74	127.30	61.61	69.96	10.53	100.88	206.73	69.37	73.07	19.82	2101.91	1340.30	1305.93	400.02	93.84	183.96	



## Security Cost Allocation with Different Slack Buses

Figure 6.14 Security cost allocated to each load with different slack buses in the 57-bus test system



#### Max, Min and Average Cost Values of Each Load in 57-bus System

Figure 6.15 Maximum, minimum and average security cost values of each load in 57-bus system

Figure 6.13 displays the security costs which are from the experimental results in chapter 5 as the security cost resources. It is indicated that the security costs calculated using by different slack bus are very similar.

Figure 6.14 gives a comparison of the security cost allocation to each load under each slack bus selection. It is found that when select the generator bus 1, 2, 3, 9 and 12 respectively as the slack bus, the allocation results are close. However, when bus 6 or bus 8 is chosen as the slack bus, the results of security cost allocation become different. In these two slack bus selections, load 25, 27, 28, 29, 30, 31, 52, 53 and 54 are allocated smaller security cost than the other five cases and the costs allocated to load 10, 13, 14, 15, 17, 18, 20, 35, 41, 42, 43, 44, 47, 49, 50, 51, 55, 56 and 57 are more than the other five slack bus selections. It is noticeable that load 5 and 18 in these five slack bus selections are allocated negative security cost, which means such loads have positive impact on congestion relief. The PTDF-based method is used to find out each bus's real contribution to the congestion no matter it is positive or negative. Here these two loads are considered as the ones who are helping to relieve the congestion under these particular slack bus selections. Theoretically, they should be rewarded by the system operator rather than shoulder the security cost. From the above results, it is also obtained that the PTDF based method could result in different allocation of security cost of individual load and is dependent on the selection of slack bus.

Figure 6.15 shows the maximum, the minimum and the average security cost values of each load in 57-bus system under different slack bus selections. From the figure, it can be found that load 29 has the biggest range between the maximum value at £3375/h and the minimum value at £2325/h. The range is £1050/h. Load 32 has the smallest range between the maximum value at £287/h and the minimum value at £275/h. The range is £12/h. The black curve shows the average security cost values of each load under different

slack bus selections. From the curve, it can be noted that load 29 has the highest average security cost allocation value at  $\pounds$ 2966/h and load 5 has the lowest value at  $\pounds$ 220/h.

## 6.4.4.2. Comparison between Pro Rata Method and PTDF Method in the 57bus Test System

The pro rata based method is used to compare with the PTDF method with selecting generator bus one as the slack bus. The comparison will be represented in Table 6.6 and Figure 6.16 as follow respectively:

# Table 6.6 Comparison between Two Methods on Security Cost Allocation in 57-busTest System

(£/h)	L5	L10	L13	L14	L15	L16	L17
Pro rata	770.70	385.35	256.90	256.90	513.80	256.90	642.25
slack bus 1	-339.84	21.61	19.71	27.63	23.70	8.60	11.29
	L18	L19	L20	L23	L25	L27	L28
Pro rata	770.70	385.35	770.70	385.35	642.25	385.35	256.90
slack bus 1	-99.07	54.14	239.64	293.39	1215.32	1579.82	1205.49
	L29	L30	L31	L32	L33	L35	L38
Pro rata	642.25	385.35	385.35	256.90	385.35	642.25	256.90
slack bus 1	3247.97	687.52	585.00	286.17	429.25	369.15	133.37
	L41	L42	L43	L44	L47	L49	L50

Pro rata	513.80	385.35	385.35	256.90	642.25	256.90	385.35
slack bus 1	82.73	84.69	28.03	108.18	227.08	78.13	87.66
	L51	L52	L53	L54	L55	L56	L57
Pro rata	385.35	513.80	385.35	642.25	642.25	385.35	642.25
slack bus 1	36.89	2027.93	1295.82	1275.05	411.37	107.46	205.37





Figure 6.16 Comparison between two methods on security cost allocation in the 57-bus test system

Figure 6.16 indicates the comparison of security cost allocation results between pro rata method and PTDF method back on slack bus 1. The curve of the pro rata method varies between the vertical coordinate intervals between  $\pounds$  0/h to  $\pounds$  1000/h. The slack bus 1 curve is apparently above the pro rata curve only at load 25, 27, 28, 30, 31, 52, 53, 54 and especially 29 and the rest of them are below the pro rata curve. It is explained that the above loads particularly load 29 have higher contribution to the congestion occurs on the line 4-29 so they should bear more security cost than other loads.

#### 6.4.5. Modified IEEE-118 Bus System

The modified IEEE-118 bus test system in this case is same with the system in chapter 4 and 5 respectively, which is shown in Figure 6.17 as follow:



Figure 6.17 The modified IEEE-118 bus system

The parameters of the generators and loads are set up as the same as the former two chapters, which are displayed in Table 4.12 and Table 4.13 respectively. There are 12 generators and 104 loads in the test system. Transmission congestion are still set on the target branch 23-24. The security costs are taken from the experimental results in section 5.5.5 which are displayed in Figure 6.18.



# Security Cost With Different Slack Buses

Figure 6.18 Security cost settings in the 118-bus test system

## 6.4.5.1. Comparison between Security Cost Allocation with Different Slack Bus Selections in the 118-bus Test System

By implementing the proposed PTDF based security cost allocation method, the security cost allocated to each load in the test system with twelve generator buses selected as the slack bus respectively is shown in Table 6.7 and Figure 6.19 respectively as follows:

## Table 6.7 Security Cost Allocated to Each Load with Different Slack Buses in the 118-bus Test System

(£/h)	L1	L2	L3	L4	L5	L6	L7	L8	L9	L10	L11	L13	L14	L15	L16	L17	L18	L20
slack 12	0.74	0.19	0.93	1.91	0.69	1.04	0.43	3.88	1.29	2.58	1.04	0.06	-0.25	-3.70	-3.02	-4.77	-3.52	-32.63
slack 19	7.40	4.64	7.59	8.53	2.89	7.69	4.87	10.44	3.48	6.96	12.15	6.74	3.10	9.78	1.53	-2.40	-1.18	-28.28
slack 24	267.43	178.36	267.39	267.15	89.04	267.36	178.30	266.67	88.89	177.78	445.77	267.60	133.87	536.13	179.15	90.37	90.07	141.78
slack 27	79.40	52.74	79.53	80.14	26.75	79.59	52.90	81.39	27.13	54.26	132.22	78.97	39.31	155.53	50.71	23.29	24.08	18.82
slack 40	-464.03	-310.32	-463.41	-460.32	-153.28	-463.09	-309.56	-454.09	-151.36	-302.73	-773.99	-466.20	-233.97	-944.46	-320.49	-170.58	-166.61	-336.58
slack 49	960.38	641.32	959.71	956.31	318.60	959.35	640.48	949.48	316.49	632.98	1601.30	962.76	482.33	1938.77	652.47	337.58	333.23	594.95
slack 59	878.73	586.77	878.13	875.11	291.55	877.81	586.02	869.02	289.67	579.35	1465.14	880.85	441.27	1773.49	596.70	308.45	304.57	541.56
slack 61	865.30	577.80	864.71	861.75	287.10	864.40	577.06	855.79	285.26	570.53	1442.75	867.38	434.52	1746.32	587.53	303.66	299.86	532.78

slack 70	354.19	236.32	354.06	353.43	117.78	353.99	236.17	352.15	117.38	234.77	590.44	354.63	177.49	711.74	238.41	121.32	120.51	198.51
slack 85	523.80	349.64	523.52	522.11	173.96	523.37	349.29	519.28	173.09	346.18	873.27	524.78	262.78	1055.05	354.26	181.83	180.02	309.44
slack 99	532.17	355.23	531.88	530.43	176.74	531.73	354.87	527.53	175.84	351.68	887.23	533.18	267.00	1072.00	359.98	184.81	182.96	314.92
slack 110	530.16	353.89	529.87	528.44	176.07	529.72	353.53	525.55	175.18	350.37	883.88	531.17	265.99	1067.93	358.61	184.10	182.26	313.60
(£/h)	L21	L22	L23	L25	L26	L28	L29	L31	L32	L33	L34	L35	L36	L37	L38	L39	L41	L42
slack 12	-53.81	-52.01	-313.79	-226.92	-93.25	-37.38	-65.48	-124.44	-214.33	31.83	171.76	103.61	34.48	34.78	35.23	87.75	159.65	388.34
slack 19	-48.80	-48.16	-295.13	-208.73	-83.67	-33.99	-58.99	-111.67	-196.53	35.29	177.56	107.07	35.64	35.93	36.36	89.48	161.37	389.64
slack 24	146.96	101.92	433.53	501.50	290.41	98.34	194.42	387.24	498.42	170.63	404.04	242.29	80.77	80.70	80.59	156.96	228.59	440.29
slack 27	5.42	-6.60	-93.32	-12.03	19.93	2.66	11.19	26.50	-4.07	72.76	240.25	144.50	48.13	48.32	48.60	108.15	179.96	403.61
slack 40	-403.69	-320.23	-1616.05	-1496.27	-761.83	-273.88	-518.37	-1016.12	-1456.37	-210.08	-233.09	-138.11	-46.20	-45.26	-43.84	-32.89	39.46	297.68

slack 49	668.60	501.82	2375.07	2393.99	1287.21	450.95	869.65	1716.66	2350.20	531.30	1007.68	602.69	201.07	200.04	198.49	336.84	407.81	575.54
slack 59	607.14	454.71	2146.35	2171.03	1169.77	409.41	790.10	1560.03	2132.04	488.80	936.52	560.21	186.89	185.97	184.59	315.63	386.67	559.54
slack 61	597.03	446.96	2108.72	2134.36	1150.46	402.58	777.01	1534.27	2096.15	481.81	924.82	553.23	184.56	183.66	182.31	312.15	383.20	556.92
slack 70	212.26	151.98	676.55	738.39	415.19	142.48	278.94	553.66	730.22	215.79	479.64	287.43	95.84	95.65	95.36	179.49	251.05	457.29
slack 85	339.95	249.87	1151.84	1201.65	659.19	228.79	444.23	879.08	1183.51	304.06	627.34	375.61	125.28	124.85	124.21	223.50	294.88	490.30
slack 99	346.25	254.70	1175.30	1224.52	671.23	233.05	452.39	895.14	1205.89	308.42	634.64	379.97	126.73	126.29	125.63	225.67	297.05	491.93
slack 110	344.73	253.53	1169.61	1218.99	668.33	232.03	450.42	891.27	1200.48	307.38	632.92	378.94	126.39	125.95	125.30	225.17	296.56	491.62
(£/h)	L43	L44	L45	L46	L47	L48	L50	L51	L52	L53	L54	L55	L56	L57	L58	L60	L62	L63
slack 12	101.56	112.14	417.83	180.89	287.68	183.83	138.98	139.60	139.75	186.90	93.66	562.56	749.65	139.90	140.01	666.84	666.35	95.33
slack 19	102.86	112.00	415.99	179.72	285.43	182.58	138.00	138.60	138.75	185.55	92.98	558.45	744.18	138.89	139.00	661.71	661.24	94.60

slack 24	153.58	106.39	343.88	134.19	197.29	133.47	99.83	99.68	99.64	132.72	66.31	397.70	530.38	99.61	99.58	461.42	461.54	65.90
slack 27	116.89	110.43	395.96	167.09	260.98	168.95	127.41	127.81	127.90	170.89	85.58	513.86	684.87	127.99	128.06	606.15	605.84	86.64
slack 40	10.86	122.12	546.57	262.20	445.12	271.54	207.16	209.12	209.61	281.27	141.30	849.68	1131.52	210.06	210.42	1024.61	1023.06	146.60
slack 49	288.82	91.53	152.02	12.98	-37.39	2.74	-1.79	-3.95	-4.48	-7.94	-4.70	-30.26	-38.83	-4.98	-5.37	-71.87	-70.17	-10.52
slack 59	272.87	93.27	174.56	27.23	-9.78	18.11	10.16	8.24	7.77	8.60	3.65	20.07	28.12	7.32	6.97	-9.14	-7.63	-1.53
slack 61	270.25	93.55	178.28	29.58	-5.24	20.64	12.13	10.25	9.79	11.33	5.02	28.36	39.15	9.35	9.01	1.19	2.67	-0.05
slack 70	170.53	104.55	319.93	119.05	167.96	117.13	87.13	86.73	86.63	115.14	57.43	344.22	459.24	86.54	86.46	394.76	395.08	56.35
slack 85	203.61	100.89	272.87	89.34	110.45	85.09	62.23	61.33	61.11	80.67	40.03	239.33	319.74	60.90	60.74	264.08	264.78	37.62
slack 99	205.24	100.71	270.55	87.87	107.61	83.51	61.00	60.08	59.85	78.97	39.17	234.16	312.86	59.64	59.48	257.63	258.36	36.70
slack 110	204.87	100.77	271.19	88.26	108.35	83.93	61.32	60.41	60.18	79.41	39.40	235.51	314.66	59.97	59.81	259.31	260.02	36.94

(£/h)	L64	L65	L66	L67	L68	L71	L72	L73	L74	L75	L76	L77	L78	L79	L80	L81	L82	L83
slack 12	191.20	96.33	284.07	189.85	107.38	197.92	277.86	197.92	940.21	744.46	832.70	745.13	865.37	367.55	119.32	111.78	610.07	243.69
slack 19	189.72	95.56	281.93	188.41	106.27	194.00	271.45	194.00	924.37	732.47	820.19	735.35	854.08	362.81	117.84	110.53	602.25	240.57
slack 24	131.67	65.66	198.17	132.00	62.95	40.82	21.28	40.82	305.38	264.03	331.66	353.07	412.88	177.76	60.04	61.88	296.88	118.84
slack 27	173.61	87.27	258.70	172.76	94.26	151.55	202.14	151.55	752.83	602.65	684.80	629.38	731.79	311.52	101.81	97.05	517.61	206.83
slack 40	294.89	149.74	433.69	290.60	184.77	471.63	724.91	471.63	2046.22	1581.45	1705.58	1428.12	1653.64	698.16	222.58	198.71	1155.65	461.18
slack 49	-22.89	-13.96	-24.84	-18.18	-52.40	-367.23	-645.20	-367.23	-1343.44	-983.74	-969.59	-665.09	-762.22	-315.10	-93.91	-67.70	-516.43	-205.39
slack 59	-4.71	-4.60	1.39	-0.52	-38.83	-319.18	-566.71	-319.18	-1149.32	-836.83	-816.41	-545.25	-623.90	-257.09	-75.79	-52.45	-420.70	-167.23
slack 61	-1.71	-3.06	5.71	2.39	-36.59	-311.28	-553.80	-311.28	-1117.37	-812.66	-791.19	-525.52	-601.13	-247.53	-72.80	-49.94	-404.94	-160.94
slack 70	112.35	55.70	170.30	113.23	48.53	-10.23	-62.11	-10.23	99.12	107.94	168.89	225.73	265.91	116.12	40.78	45.67	195.16	78.29

slack 85	74.48	36.19	115.64	76.42	20.27	-110.16	-225.31	-110.16	-304.68	-197.64	-149.81	-23.66	-21.92	-4.60	3.08	13.93	-4.05	-1.13
slack 99	72.61	35.23	112.95	74.61	18.88	-115.09	-233.36	-115.09	-324.60	-212.72	-165.53	-35.96	-36.12	-10.56	1.22	12.37	-13.88	-5.05
slack 110	73.09	35.48	113.65	75.08	19.23	-113.87	-231.38	-113.87	-319.65	-208.96	-161.60	-32.87	-32.55	-9.06	1.68	12.76	-11.41	-4.06
(£/h)	L84	L86	L87	L88	L89	L90	L91	L92	L93	L94	L95	L96	L97	L98	L100	L101	L102	L103
slack 12	121.58	121.45	121.45	484.86	121.05	121.01	120.96	725.35	181.23	362.27	483.29	362.73	180.20	359.24	361.39	241.29	181.22	240.92
slack 19	120.03	119.91	119.91	478.70	119.52	119.48	119.43	716.18	178.94	357.69	477.19	358.14	177.94	354.76	356.84	238.25	178.93	237.89
slack 24	59.48	59.51	59.51	238.29	59.61	59.62	59.63	357.91	89.50	179.05	238.67	178.94	89.75	179.79	179.27	119.42	89.51	119.51
slack 27	103.25	103.17	103.17	412.06	102.91	102.89	102.86	616.87	154.15	308.18	411.07	308.47	153.50	306.26	307.62	205.31	154.14	205.08
slack 40	229.75	229.35	229.35	914.38	228.07	227.95	227.80	1365.46	341.02	681.43	909.44	682.89	337.76	671.85	678.64	453.60	340.98	452.43
slack 49	-101.78	-101.33	-101.33	-402.02	-99.93	-99.80	-99.64	-596.32	-148.70	-296.74	-396.59	-298.34	-145.13	-286.22	-293.67	-197.07	-148.65	-195.78
slack 59	-82.80	-82.40	-82.40	-326.66	-81.15	-81.04	-80.89	-484.01	-120.66	-240.73	-321.82	-242.16	-117.48	-231.37	-238.01	-159.81	-120.62	-158.67
-----------	--------	--------	--------	---------	--------	--------	--------	---------	---------	---------	---------	---------	---------	---------	---------	---------	---------	---------
slack 61	-79.67	-79.28	-79.28	-314.25	-78.06	-77.95	-77.80	-465.51	-116.04	-231.51	-309.51	-232.91	-112.93	-222.34	-228.84	-153.68	-116.01	-152.56
slack 70	39.31	39.40	39.40	158.21	39.66	39.68	39.71	238.56	59.71	119.55	159.22	119.25	60.38	121.51	120.12	79.84	59.72	80.08
slack 85	-0.18	0.00	0.00	1.37	0.58	0.63	0.70	4.83	1.37	3.01	3.62	2.34	2.85	7.36	4.27	2.32	1.38	2.85
slack 99	-2.13	-1.94	-1.94	-6.37	-1.35	-1.29	-1.22	-6.70	-1.51	-2.74	-4.06	-3.42	0.01	1.73	-1.44	-1.51	-1.49	-0.96
slack 110	-1.64	-1.45	-1.45	-4.42	-0.86	-0.81	-0.74	-3.79	-0.79	-1.29	-2.13	-1.97	0.72	3.15	0.00	-0.54	-0.77	0.00
(£/h)	L104	L105	L106	L107	L108	L109	L111	L112	L113	L114	L115	L116	L117	L118				
slack 12	361.39	361.39	481.85	602.31	240.92	120.46	120.46	72.28	-9.69	-42.33	-84.47	107.38	0.00	432.44				
slack 19	356.84	356.84	475.79	594.73	237.89	118.95	118.95	71.37	-7.16	-38.78	-77.39	106.27	4.45	425.69				
slack 24	179.27	179.27	239.03	298.78	119.51	59.76	59.76	35.85	91.57	99.55	199.06	62.95	178.41	161.90				

slack 27	307.62	307.62	410.16	512.70	205.08	102.54	102.54	61.52	20.18	-0.47	-0.83	94.26	52.62	352.58		
slack 40	678.64	678.64	904.85	1131.07	452.43	226.21	226.21	135.73	-186.16	-289.56	-578.55	184.77	-310.92	903.78		
slack 49	-293.67	-293.67	-391.56	-489.45	-195.78	-97.89	-97.89	-58.73	354.68	468.16	935.69	-52.40	641.97	-540.76		
slack 59	-238.01	-238.01	-317.34	-396.68	-158.67	-79.34	-79.34	-47.60	323.68	424.74	848.90	-38.83	587.35	-458.04		
slack 61	-228.84	-228.84	-305.12	-381.40	-152.56	-76.28	-76.28	-45.77	318.58	417.59	834.63	-36.59	578.37	-444.42		
slack 70	120.12	120.12	160.16	200.20	80.08	40.04	40.04	24.02	124.51	145.69	291.27	48.53	236.45	74.00		
slack 85	4.27	4.27	5.70	7.12	2.85	1.42	1.42	0.85	188.91	235.93	471.59	20.27	349.91	-98.09		
slack 99	-1.44	-1.44	-1.92	-2.40	-0.96	-0.48	-0.48	-0.29	192.09	240.38	480.49	18.88	355.51	-106.58		
slack 110	0.00	0.00	0.00	0.00	0.00	0.00	0.00	0.00	191.33	239.30	478.34	19.23	354.17	-104.46		



## Security Cost Allocation with Different Slack Buses

Figure 6.19 Security cost allocated to each load with different slack buses in the 118-bus test system

Since there are totally 104 loads in the system, it is not practical to use only one graph to contain such large density of security cost allocation results. Hence the Figure 6.19 is divided into three sections which are Figure 6.20 contains results from load 1 to load 41, Figure 6.21 contains results from load 42 to load 81 and Figure 6.22 contains results from load 82 to load 118. Those three figures are shown successively as follows:



## Security Cost Allocation with Different Slack Buses from L1 to L41

Figure 6.20 Security cost allocated to each load with different slack buses in the 118-bus test system from L1 to L41



## Security Cost Allocation with Different Slack Buses from L42 to L81

Figure 6.21 Security cost allocated to each load with different slack buses in in the 118-bus test system from L42 to L81



## Security Cost Allocation with Different Slack Buses from L82 to L118





#### Max, Min and Average Cost Values of Each Load in 118-bus System

Figure 6.23 Maximum, minimum and average security cost values of each load in 118-bus system

Figure 6.18 shows the security costs from the experimental results in chapter 5 as the basic security cost. Due to there are 12 generators in the system, slack bus selection have twelve possibilities, which is so say that the whole security cost allocation results should be based on twelve groups of PTDF factors respectively. After observed the contents from Figure 6.20, 6.21 and 6.22 respectively, the following regular patterns are found. Security cost allocations with bus 12, 19 and 27 as the slack bus respectively have similar results. Selecting bus 49, 59 and 61 respectively as the slack bus causes similar allocation outcomes. Selections of slack bus 89, 99 and 110 have similar results for security cost allocation. While selections of slack bus 24, 40 and 70 have their own characteristic respectively to allocate the security cost to the loads. Slack bus 40 as one of the unique cases has very different idea for security cost allocation, which can be observed in the above three graphs. It should be noted that many loads are allocated negative values of security cost, which means such loads have negative contributions on power flow of congested line, in other words, they are helping to relieve the congestion. As mentioned in section 6.3.2, the PTDF-based method is trying to detect each bus's real contribution to one particular transmission line's power flow no matter it is positive or negative. Therefore, based on these real contributions of loads, the system operator could take measures to do appropriate security cost allocation to loads. The system operator should avoid allocating security cost to such loads who have negative contributions to the power flow of the congested line and even reward them. The 118-bus case is a convincing example to illustrate that the PTDF based method could result in different allocation of security cost to load and can be highly dependent on slack bus position.

Figure 6.23 shows the maximum, the minimum and the average security cost values of each load in 118-bus system under different slack bus selections. From the figure, it can be found that load 23 has the biggest range between the maximum value at £2375/h and the minimum value at -£1616/h. The range is £3991/h. Load 44 has the smallest range between the maximum value at £122/h and the minimum value at £91/h. The range is

 $\pounds$ 31/h. The black curve shows the average security cost values of each load under different slack bus selections. From the curve, it can be noted that load 25 has the highest average security cost allocation value at £803/h and load 72 has the lowest value at -£85/h.

## 6.4.5.2. Comparison between Pro Rata Method and PTDF Method in the 118bus Test System

Using the results from the PTDF method of case of slack bus 12 to contrast with the pro rata based method. The comparison is obtained in Table 6.8 and Figure 6.24 respectively as follows:

(£/h)	L1	L2	L3	L4	L5	L6	L7	L8	L9	L10	L11	L13	L14
Pro rata	215.08	143.38	215.08	215.08	71.69	215.08	143.38	215.08	71.69	143.38	358.46	215.08	107.54
slack bus 12	0.74	0.19	0.93	1.91	0.69	1.04	0.43	3.88	1.29	2.58	1.04	0.06	-0.25
	L15	L16	L17	L18	L20	L21	L22	L23	L25	L26	L28	L29	L31
Pro rata	430.15	143.38	71.69	71.69	107.54	107.54	71.69	286.77	358.46	215.08	71.69	143.38	286.77
slack bus 12	-3.70	-3.02	-4.77	-3.52	-32.63	-53.81	-52.01	-313.79	-226.92	-93.25	-37.38	-65.48	-124.44
	L32	L33	L34	L35	L36	L37	L38	L39	L41	L42	L43	L44	L45
Pro rata	358.46	143.38	358.46	215.08	71.69	71.69	71.69	143.38	215.08	430.15	143.38	107.54	358.46
slack bus 12	-214.33	31.83	171.76	103.61	34.48	34.78	35.23	87.75	159.65	388.34	101.56	112.14	417.83
	L46	L47	L48	L50	L51	L52	L53	L54	L55	L56	L57	L58	L60
Pro rata	143.38	215.08	143.38	107.54	107.54	107.54	143.38	71.69	430.15	573.54	107.54	107.54	501.85
slack bus 12	180.89	287.68	183.83	138.98	139.60	139.75	186.90	93.66	562.56	749.65	139.90	140.01	666.84
	L62	L63	L64	L65	L66	L67	L68	L71	L72	L73	L74	L75	L76
Pro rata	501.85	71.69	143.38	71.69	215.08	143.38	71.69	71.69	71.69	71.69	430.15	358.46	430.15
slack bus 12	666.35	95.33	191.20	96.33	284.07	189.85	107.38	197.92	277.86	197.92	940.21	744.46	832.70

#### Table 6.8 Comparison between Two Methods on Security Cost Allocation in the 118-bus Test System

	L77	L78	L79	L80	L81	L82	L83	L84	L86	L87	L88	L89	L90
Pro rata	430.15	501.85	215.08	71.69	71.69	358.46	143.38	71.69	71.69	71.69	286.77	71.69	71.69
slack bus 12	745.13	865.37	367.55	119.32	111.78	610.07	243.69	121.58	121.45	121.45	484.86	121.05	121.01
	L91	L92	L93	L94	L95	L96	L97	L98	L100	L101	L102	L103	L104
Pro rata	71.69	430.15	107.54	215.08	286.77	215.08	107.54	215.08	215.08	143.38	107.54	143.38	215.08
slack bus 12	120.96	725.35	181.23	362.27	483.29	362.73	180.20	359.24	361.39	241.29	181.22	240.92	361.39
	L105	L106	L107	L108	L109	L111	L112	L113	L114	L115	L116	L117	L118
Pro rata	215.08	286.77	358.46	143.38	71.69	71.69	43.02	71.69	71.69	143.38	71.69	143.38	215.08
slack bus 12	361.39	481.85	602.31	240.92	120.46	120.46	72.28	-9.69	-42.33	-84.47	107.38	0.00	432.44



## Security Cost Allocation by Two Methods

Figure 6.24 Comparison between two methods on security cost allocation in the 118-bus test system

Figure 6.24 indicates the comparison of security cost allocation results between pro rata method and PTDF method. The curve of the pro rata method varies between in the vertical coordinate interval between  $\pounds$  0/h to  $\pounds$  600/h. Before load 33, from load 1 to load 18, the PTDF method considers these loads have no contribution to the congested line 23-24, the security cost allocated to them are nearly zero. From load 20 to load 32, the PTDF method treats these loads have negative contributions to the power flow of congested transmission line so their allocated costs are negative values. Based on the same reason, load 113 to load 115 have negative security cost allocations. Beginning from load 33 to load 118, two curves have the similar track but the PTDF curve is higher than the pro rata curve all alone except load 113, 114, 115 and 117. The slack bus 12 curve is apparently higher than the pro rata curve at load 56, 60, 62, 74, 75, 76, 77, 78, 82, 88, 92 and 107. It is because the above loads have higher contribution to the congestion occurs on the line 23-24 hence they shoulder more security cost than other loads. The only difference of judgement on security cost allocation between two method is that the pro rata method is based on the power consumption ratio of loads in the system to allocate the security cost, which is so say that the load with higher active power consumption will be allocated more security cost by the system operator. In the pro rata method, even the load who has a least consumption as long as it withdraws positive volume of power from the system, it will definitely shoulder the cost of congestion. While the PTDF-based method digs in and tries to find out the power flow contribution each load makes to the congested transmission line's power flow. The load with higher power flow contribution to the congested transmission line will be allocated more security cost by the system operator. The negative allocated security costs to the loads in the PTDF-based method is actually to indicate the absolute power flow contribution of load to the congested transmission line, which can inform that such loads help to relieve the congestion. The consumptions of power at such buses are actually decreasing the power flow of the congested transmission line. In theory, negative security cost allocation means that the system operator should reward the loads who helps to relieve the congestion. But it depends on the specific mechanism in each electricity market. The PTDF-based method is only to give a real experimental results to the system operator.

#### 6.4.6. Discussion

By contrasting the PTDF based security cost allocation method with the pro rata based method, no matter in which test systems it is proved that PTDF method is able to calculate the power flow contribution of each load to the congested line. Even though the comparison in 14-bus test system is not distinct, in other three test systems, the comparisons show that the PTDF method allocates higher security cost to the loads who have more contribution to the transmission congestion, at the meantime, allocates lower security cost to the loads who have little impact on the congestion, which seems like fairer and more convincing than the pro rata method.

Nevertheless, through the testing results from four cases, the common drawback of the PTDF method is exposed, which is the complexity and inaccuracy on security cost allocation. It is discovered that the PTDF based method is dependent on slack bus selection, which means choosing different generator bus as the slack bus leads to different outcomes. In each test case, every generator bus is selected as the slack bus respectively and the corresponding experimental results on security cost allocation are obtained. The results indicate that although some of outcomes with different slack buses are similar, there are always one or some loads that have apparently different outcomes on security cost allocation. Consequently, when utilize the PTDF method, the system operator should consider all the slack bus selections in order to guarantee the preciseness of the method, which causes the complexity of the process, for example, in 118-bus case, the system operator should conduct 12 times for calculation to see all the possible results. However, if the system operator only considers one slack bus choice, the result on security cost allocation is only based on the slack bus position.

## 6.5. Summary

In this chapter, conventional security cost allocation schemes are described and discussed. The mathematic model of the proposed allocation method with PTDF factor is given. Experimental results about security cost allocation of four modified IEEE test systems are represented through tables and figures. The comparisons between the pro rata method and the PTDF method by four test systems are obtained. The advantage of the PTDF method is that it can calculate the power flow contribution of each load to the congested transmission line so as to provide a fair way to allocate the security cost. The disadvantage of the PTDF method is that it is dependent on slack bus selection.

## **Chapter 7**

# Security Cost Allocation in Deregulated Electricity Market with Bus Impedance Matrix Sensitivity

#### 7.1. Introduction

In chapter 6, the security cost allocation method associated with the PTDF is demonstrated in four modified IEEE test systems respectively. To provide an alternative security cost allocation method, the bus impedance matrix sensitivity will be introduced, which is indicated and discussed in chapter 5. As another power flow sensitivity factor similar with the PTDF, the ZS factor is also able to seek out the power flow contribution of each load in the system to the congested transmission line. The security cost allocation with bus impedance matrix sensitivity is based on the power flow contribution of each load. To find out the advantages and disadvantages of the ZS based allocation method, the result of security cost allocated to each load in the system by three methods which are the pro rata method, the PTDF method and the ZS method will be compared and discussed in the case studies.

In this chapter, the mathematic model of the proposed security cost allocation method with the ZS will be expressed in section 7.2. Section 7.3 indicates the case studies including experimental results by the ZS method in four modified IEEE test systems including: an IEEE-14 bus system, an IEEE-30 bus system, an IEEE-57 bus system and an IEEE-118 bus system. Then combined with the results from chapter 6, the comparison among three security cost allocation methods which are pro rata method, the PTDF method with the first generator bus as the slack bus and the ZS method with the first generator bus as the slack bus and the SS method with the first generator bus as the slack bus will be obtained. Section 7.4 is the summary.

## 7.2. Proposed Security Cost Allocation Method with Bus Impedance Matrix Sensitivity

The proposed security cost allocation method with the ZS factor will be described in details including the mechanism and the mathematical model of the method.

## 7.2.1. Mechanism of the Proposed Security Cost Allocation Method with Bus Impedance Matrix Sensitivity

The mechanism of the proposed security cost allocation method with the ZS is almost as same as the PTDF based method, which is described in section 6.3.1. But this time it is with the use of the ZS factor to calculate the power flow contribution percentage of each load to the congested transmission line and the security cost allocation is conducted according to the above contribution percentage.

# 7.2.2. Mathematical Model of the Proposed Security Cost Allocation Method with Bus Impedance Matrix Sensitivity

The following equations are the mathematical model of the proposed security cost allocation method with bus impedance matrix sensitivity:

Equation (7.1) expressed the total MW power flow of the congested line is due to generators contribution plus loads contribution.

$$P_{j-k} = P_{j-k}^G + P_{j-k}^L$$
(7.1)

Where

 $P_{j-k}$ : The power flow MW value of the target transmission line *j*-*k* 

 $P_{j-k}^{G}$ : The power flow MW value on line *j*-*k* associated with all the generators

 $P_{j-k}^L$ : The power flow MW value on line *j*-*k* associated with all the loads

Equation (7.2) indicates that the total MW contribution of generators is the sum of the power flow contribution of each generator.

$$P_{j-k}^{G} = \sum_{i=1}^{N_{G}} ZS_{G_{i}}^{j-k} \cdot P_{G_{i}}$$
(7.2)

Where

 $ZS_{G_i}^{j-k}$ : The ZS of the line *j-k* respect to generator bus *i*  $P_{G_i}$ : The MW power energy injected by generator *i*.

Equation (7.3) shows that the total MW contribution of loads is the sum of the power flow contribution of each load.

$$P_{j-k}^{L} = \sum_{i=1}^{N_{L}} ZS_{L_{i}}^{j-k} \cdot P_{L_{i}}$$
(7.3)

Where

$$ZS_{L_i}^{j-k}$$
: The ZS of the line *j*-*k* respect to load bus *i*

 $P_{L_i}$ : The MW power energy withdrawn by load *i*.

Equation (7.4) shows that the MW power flow contribution of each load is equals to the ZS of the load i times the power energy extraction of the load i.

$$P_{j-k}^{L_i} = ZS_{L_i}^{j-k} \cdot P_{L_i} \tag{7.4}$$

Where

 $P_{j-k}^{L_i}$ : The power flow MW value associated with load *i*.

Equation (7.5) calculates of the power flow contribution percentage of each load.

$$Ctb_{L_{i}} = \frac{P_{j-k}^{L_{i}}}{P_{j-k}^{L}} \times 100\% = \frac{ZS_{L_{i}}^{j-k} \cdot P_{L_{i}}}{\sum_{i=1}^{N_{L}} ZS_{L_{i}}^{j-k} \cdot P_{L_{i}}} \times 100\%$$
(7.5)

Where

 $Ctb_{L_i}$ : The power flow contribution percentage of load *i* to the congested line

Equation (7.6) finds out the security cost allocated to each load.

$$CA_{L_i} = Ctb_{L_i} \times SC_{total} \tag{7.6}$$

Where

 $CA_{L_i}$ : The allocated security cost to the load *i* 

*SC<sub>total</sub>* : The total security cost due to congestion management.

#### 7.3. Case Studies

Similar with previous three chapters, the case studies in this chapter use four modified IEEE bus systems including an IEEE-14 bus system, an IEEE-30 bus system, an IEEE-57 bus system and an IEEE-118 bus system to test the performance of the proposed security cost allocation method with the ZS. The comparison between the ZS method and the other two methods including the pro rata method and the PTDF method will be obtained and discussed.

#### 7.3.1. Test Description

Based on the contents in section 5.5, the PTDF based method and ZS based method are both utilized to re-dispatch generation to relieve the transmission congestion. The results indicate that the security costs caused by two congestion management schemes in all four test systems are all apparently different. However, in order to make the comparison of security cost allocation by above two methods more direct viewing, the security cost they are aim to allocate in these case studies should be assumed at a same amount. The security cost of the first slack bus selection in each test system of the section 6.4 will be chosen as the security cost amount in each corresponding test system of these case studies. It should be noted that since the ZS method is also dependent on the slack bus selection, here the ZS method only considers the first generator bus as the slack bus, which is same as the slack bus selection in the PTDF method. The pro rata method which is mentioned in section 6.4.1 will continue to be used to provide the third set of comparable result. Additionally, rule and environment of the test in this chapter are same with the chapter 4 and the details can be found in section 4.5.1. The test description is visualized by Figure 7.1 as follows:



Figure 7.1 Test description for chapter 7

### 7.3.2. Modified IEEE-14 Bus System

The modified IEEE-14 bus test system is exactly same with the system in section 6.4.2, which is showed in Figure 7.2 as follow:



Figure 7.2 The modified IEEE-14 bus system

The parameters of the generators and loads are set up the same, which are displayed in Table 4.1. Transmission congestion occurs on the transmission branch 3-4. There are 5 generators and 9 loads in the system. The security cost amount is picked up from the experimental result shown in Figure 6.3, which is £5526.62/h. Then implement three security cost allocation methods respectively including the pro rata method, the PTDF method with the first slack bus selection and the ZS method with the first slack bus selection to allocate the above security cost to the loads in the system. The comparison will be illustrated in Table 7.1 and Figure 7.3 as follows:

(£/h)	L4	L5	L7	L9	L10	L11	L12	L13	L14
Pro rata	552.66	552.66	663.19	552.66	663.19	1215.86	442.13	442.13	442.13
PTDF slack bus 1	642.13	437.46	726.49	585.66	685.43	1184.27	408.55	412.57	444.06
ZS slack bus 1	665.08	372.62	744.06	605.45	704.43	1175.73	396.82	405.21	457.23

Table 7.1 Comparison between Three Methods on Security Cost Allocation in the 14-bus Test System

Security Cost Allocation by Three Methods



Figure 7.3 Comparison between three methods on security cost allocation in the 14-bus test system

From Figure 7.3, it can be obtained that security cost allocation results by three methods are very close since the tendencies of the above three curves are similar. By contrast with the curve of pro rata method, curves of the PTDF method and the ZS method are even closer. The reason behind this can be found out through the details of the calculation process of load's contributions by three methods. The pro rata method calculates each load's contribution based on their active power consumptions, which uses each load's MW consumption divided by the demand-side total MW. The other two

sensitivity-based methods calculate each load's contribution according to the sensitivitybased contribution of each load, which is calculated by the sensitivity factor of each load times its MW consumption and then divided by the total sensitivity-based contributions of loads. It should be noted that the only difference is the sensitivity factor used, one is the PTDF and the other is the ZS. Take load 14 as an example since the three methods calculated the most similar results on it: It can be found that for the contribution of load 14, the pro rata method calculation is 8%, the PTDF-based method calculation is 8.03% and the ZS-based method calculation is 8.27%.

But they still have differences in details. For example, ZS method allocates the slightly higher cost to load 4 and allocates lower cost to load 5 compared with other two methods. But the difference is so tiny, it is not convincing to conclude anything. More comparisons among three methods will be obtained in the next three test systems.

#### 7.3.3. Modified IEEE-30 Bus System

The modified IEEE-30 bus test system is as same as the system in section 6.4.3, which is showed in Figure 7.4 as follow:



Figure 7.4 The modified IEEE-30 bus system

The parameters of the generators and loads are set up as same as the former two chapters, which are displayed in Table 4.4 and Table 4.5. There are 6 generators and 18 loads in the system. Transmission congestion occurred on the transmission branch 4-12. The security cost amount is picked up from the experimental result in Figure 6.8, which is £5914.08/h. Then implement three security cost allocation methods respectively including the pro rata method, the PTDF method with the first slack bus selection and the ZS method with the first slack bus selection to allocate the above security cost to the loads in the system. The comparison will be illustrated in Table 7.2 and Figure 7.5 as follows:

( <b>£/h</b> )	L3	L4	L7	L10	L12	L14
Pro rata	208.73	278.31	347.89	417.46	417.46	347.89
PTDF slack bus 1	-10.03	-16.44	30.30	359.03	782.51	601.21
ZS slack bus 1	533.95	720.90	835.60	500.32	-149.74	-50.66
	L15	L16	L17	L18	L19	L20
Pro rata	347.89	417.46	487.04	208.73	347.89	208.73
PTDF slack bus 1	561.51	605.55	506.67	281.85	415.51	232.17
ZS slack bus 1	15.60	120.75	450.05	95.07	243.03	171.53
	L21	L23	L24	L26	L29	L30
Pro rata	208.73	347.89	417.46	347.89	208.73	347.89
PTDF slack bus 1	183.93	465.85	405.58	235.65	102.47	170.78
ZS slack bus 1	245.70	165.39	440.80	541.26	386.21	648.33

Table 7.2 Comparison between Three Methods on Security Cost Allocation in the 30-bus Test System



Security Cost Allocation by Three Methods

Figure 7.5 Comparison between three methods on security cost allocation in the 30-bus test system

From Figure 7.5, it can be seen that three curves are apparently different, which means three methods have different ideas for security cost allocation. Among all the loads, except load 10, 17, 20, 21 and 24, the other loads shoulder distinctly different security costs in three allocation schemes. For example, load 3, 4 and 7 in the PTDF method are considered as the low impact loads to the congested line so they are allocated very few security cost respectively while in the ZS method they are considered as the high impact loads so they are allocated more security cost respectively. For load 12 to load 16, the security cost allocated to them is quite higher than other loads in the PTDF method especially the load 12 which is considered as the load who has the most power flow contribution to the congested line so it is allocated the most security cost. However, these above loads in the ZS method have much lower cost allocation especially the load 12 who is regarded as the load with the smallest impact on the congested line. Additionally, load 26, 29 and 30 are allocated more security cost in the ZS method than the PTDF method.

The reason to explain the above results can be found out from the details of the calculation process of load's contributions by three methods. Here two typical examples

are taken, one is load 12, where the most obvious difference among three methods occurs, the other one is load 24, where the smallest difference occurs. It can be found that for the contribution of load 12, the pro rata method calculation is 7.06%, the PTDF-based method calculation is 13.23% and the ZS-based method calculation is -2.5%. For the contribution of load 24, the pro rata method calculation is 7.06%, the PTDF-based method calculation is 6.86% and the ZS-based method calculation is 7.45%.

### 7.3.4. Modified IEEE-57 Bus System

The modified IEEE-57 bus test system is as same as the system in section 6.4.4, which is showed in Figure 7.6 as follow:



Figure 7.6 The modified IEEE-57 bus system

The parameters of the generators and loads are set up the same with the previous two chapters, which are displayed in Table 4.8 and Table 4.9. There are 7 generators and totally 35 loads in the system. Transmission congestion occurred on the transmission branch 7-29. The security cost amount is picked up from the experimental result shown in Figure 6.13, which is £16056.25/h.

Then implement three security cost allocation methods respectively including the pro rata method, the PTDF method with the first slack bus selection and the ZS method with the first slack bus selection to allocate the above security cost to the loads in the system. The comparison between three methods on security cost allocation will be shown in Table 7.3 and Figure 7.7 as follows:

(£/h)	L5	L10	L13	L14	L15	L16	L17
Pro rata	770.70	385.35	256.90	256.90	513.80	256.90	642.25
PTDF slack bus 1	-339.84	21.61	19.71	27.63	23.70	8.60	11.29
ZS slack bus 1	1268.35	551.57	366.19	365.69	739.42	363.85	913.16
	L18	L19	L20	L23	L25	L27	L28
Pro rata	<b>L18</b> 770.70	L19 385.35	<b>L20</b> 770.70	L23 385.35	L25 642.25	L27 385.35	L28 256.90
Pro rata PTDF slack bus 1	L18 770.70 -99.07	L19 385.35 54.14	L20 770.70 239.64	L23 385.35 293.39	L25 642.25 1215.32	L27 385.35 1579.82	L28 256.90 1205.49

Table 7.3 Comparison between Three Methods on Security Cost Allocation in the 57-bus Test System

	L29	L30	L31	L32	L33	L35	L38
Pro rata	642.25	385.35	385.35	256.90	385.35	642.25	256.90
PTDF slack bus 1	3247.97	687.52	585.00	286.17	429.25	369.15	133.37
ZS slack bus 1	-748.84	260.34	322.38	267.65	402.26	831.99	328.22
	L41	L42	L43	L44	L47	L49	L50
Pro rata	513.80	385.35	385.35	256.90	642.25	256.90	385.35
PTDF slack bus 1	82.73	84.69	28.03	108.18	227.08	78.13	87.66
ZS slack bus 1	737.08	560.54	556.39	337.76	866.79	351.10	544.85
	L51	L52	L53	L54	L55	L56	L57
Pro rata	385.35	513.80	385.35	642.25	642.25	385.35	642.25
PTDF slack bus 1	36.89	2027.93	1295.82	1275.05	411.37	107.46	205.37
ZS slack bus 1	550.85	-297.46	-99.03	323.93	766.66	551.05	909.75



## Security Cost Allocation by Three Methods

Figure 7.7 Comparison between three methods on security cost allocation in the 57-bus test system

Figure 7.7 indicates that except the load interval from load 23 to 30 and load interval from load 52 to 54, the curve of the ZS method has the similar tendency with the pro rata curve but a little higher than the pro rata curve. There are two remarkable differences between the ZS method curve and the PTDF method curve, which are the load interval from load 25 to 29 and the load interval from load 52 to 54. In the PTDF method, security cost allocation on load 29 is the first peak point and load 52 is the second peak point. However, the ZS method obtains the opposite results. Security cost allocation on load 52 is the second lowest point. The explain is that in the PTDF method, the power flow contribution percentages of load 29 and 52 respectively to the congested line 7-29 are higher than other loads while in the ZS method, the power flow contribution percentages of load 29 is the lowest point and load 52 is the second lowest point and load 52 is the second lowest point and 52 is method, the power flow contribution percentages of load 29 and 52 respectively to the congested line 7-29 are lower than other loads so security allocated to the load 29 is the lowest point and load 52 is the second lowest point.

To explain the above results, the detailed calculation process of load's contributions by three methods will be discussed. As known, three methods calculate each load's contribution to the congested transmission line based on their own calculation principles respectively. This is the reason why in figure 7.7, three curves are showing differently even though there are still some close points. Here two typical examples are taken to show the details, one is load 29, where the most obvious difference among three methods occurs, the other one is load 32, where the smallest difference occurs. It can be found that for the contribution of load 29, the pro rata method calculation is 4%, the PTDF-based method calculation is 20.2% and the ZS-based method calculation is -4.7%. For the contribution of load 32, the pro rata method calculation is 1.6%, the PTDF-based method calculation is 1.8% and the ZS-based method calculation is 1.7%.

## 7.3.5. Modified IEEE-118 Bus System

The modified IEEE-118 bus test system is as same as the system in section 6.4.5, which is showed in Figure 7.8 as follow:



Figure 7.8 The modified IEEE-118 bus system

The parameters of the generators and loads are set up as the same as the former two chapters, which are displayed in Table 4.12 and Table 4.13. There are 12 generators and 104 loads in the test system. Transmission congestion occurred on the transmission branch 23-24. The security cost amount needs to be allocated is picked up from the experimental result in Figure 6.18, which is £19615.06/h.

Then three security cost allocation methods are implemented respectively including the pro rata method, the PTDF method with the first slack bus selection and the ZS method with the first slack bus selection to allocate such above security cost to the loads in the system. The comparison between three methods on security cost allocation will be shown in Table 7.4 and Figure 7.9 as follows:

(£/h)	L1	L2	L3	L4	L5	L6	L7	L8	L9	L10	L11	L13	L14
Pro rata	215.08	143.38	215.08	215.08	71.69	215.08	143.38	215.08	71.69	143.38	358.46	215.08	107.54
PTDF slack bus 12	0.74	0.19	0.93	1.91	0.69	1.04	0.43	3.88	1.29	2.58	1.04	0.06	-0.25
ZS slack bus 12	-2391.29	-1566.04	-2363.96	-2302.58	-765.03	-2311.52	-1539.97	-2261.45	-764.80	-1497.79	-3859.58	-2322.71	-1139.56
	L15	L16	L17	L18	L20	L21	L22	L23	L25	L26	L28	L29	L31
Pro rata	430.15	143.38	71.69	71.69	107.54	107.54	71.69	286.77	358.46	215.08	71.69	143.38	286.77
PTDF slack bus 12	-3.70	-3.02	-4.77	-3.52	-32.63	-53.81	-52.01	-313.79	-226.92	-93.25	-37.38	-65.48	-124.44
ZS slack bus 12	-4400.37	-1539.53	-755.16	-752.16	-1327.53	-1477.04	-1085.55	-4887.77	-4895.11	-2660.20	-1020.42	-1966.25	-3865.39
	L32	L33	L34	L35	L36	L37	L38	L39	L41	L42	L43	L44	L45
Pro rata													
1101ata	358.46	143.38	358.46	215.08	71.69	71.69	71.69	143.38	215.08	430.15	143.38	107.54	358.46
PTDF slack bus 12	358.46 -214.33	143.38 31.83	358.46 171.76	215.08 103.61	71.69 34.48	71.69 34.78	71.69 35.23	143.38 87.75	215.08 159.65	430.15 388.34	143.38 101.56	107.54 112.14	358.46 417.83
PTDF slack bus 12	358.46 -214.33 -5256.30	143.38 31.83 -1119.64	358.46 171.76 -1810.25	215.08 103.61 -1085.73	71.69 34.48 -362.71	71.69 34.78 -356.46	71.69 35.23 -350.84	143.38 87.75 -517.53	215.08 159.65 -458.06	430.15 388.34 -139.70	143.38 101.56 -367.70	107.54 112.14 124.06	358.46 417.83 896.45
PTDF slack bus 12 ZS slack bus 12	358.46 -214.33 -5256.30 L46	143.38 31.83 -1119.64 L47	358.46 171.76 -1810.25 L48	215.08 103.61 -1085.73 L50	71.69 34.48 -362.71 L51	71.69 34.78 -356.46 L52	71.69 35.23 -350.84 L53	143.38 87.75 -517.53 L54	215.08 159.65 -458.06 L55	430.15 388.34 -139.70 L56	143.38 101.56 -367.70 L57	107.54 112.14 124.06 <b>L58</b>	358.46 417.83 896.45 <b>L60</b>
PTDF slack bus 12 ZS slack bus 12 Pro rata	358.46 -214.33 -5256.30 <b>L46</b> 143.38	143.38         31.83         -1119.64         L47         215.08	358.46 171.76 -1810.25 L48 143.38	215.08 103.61 -1085.73 <b>L50</b> 107.54	71.69 34.48 -362.71 L51 107.54	71.69 34.78 -356.46 L52 107.54	71.69 35.23 -350.84 L53 143.38	143.38       87.75         -517.53       L54         71.69       1000000000000000000000000000000000000	215.08 159.65 -458.06 L55 430.15	430.15 388.34 -139.70 <b>L56</b> 573.54	143.38 101.56 -367.70 L57 107.54	107.54 112.14 124.06 <b>L58</b> 107.54	358.46 417.83 896.45 <b>L60</b> 501.85
PTDF slack bus 12 ZS slack bus 12 Pro rata PTDF slack bus 12	358.46 -214.33 -5256.30 <b>L46</b> 143.38 180.89	143.38         31.83         -1119.64         L47         215.08         287.68	358.46 171.76 -1810.25 <b>L48</b> 143.38 183.83	215.08 103.61 -1085.73 <b>L50</b> 107.54 138.98	71.69 34.48 -362.71 L51 107.54 139.60	71.69 34.78 -356.46 L52 107.54 139.75	71.69 35.23 -350.84 L53 143.38 186.90	143.38         87.75         -517.53         L54         71.69         93.66	215.08 159.65 -458.06 L55 430.15 562.56	430.15 388.34 -139.70 <b>L56</b> 573.54 749.65	143.38 101.56 -367.70 L57 107.54 139.90	107.54 112.14 124.06 <b>L58</b> 107.54 140.01	358.46 417.83 896.45 <b>L60</b> 501.85 666.84

#### Table 7.4 Comparison between Three Methods on Security Cost Allocation in the 118-bus Test System

227
	L62	L63	L64	L65	L66	L67	L68	L71	L72	L73	L74	L75	L76
Pro rata	501.85	71.69	143.38	71.69	215.08	143.38	71.69	71.69	71.69	71.69	430.15	358.46	430.15
PTDF slack bus 12	666.35	95.33	191.20	96.33	284.07	189.85	107.38	197.92	277.86	197.92	940.21	744.46	832.70
ZS slack bus 12	2113.24	304.41	601.61	292.35	839.67	585.55	399.77	1046.69	1535.94	1042.96	4972.80	3845.36	4408.78
	L77	L78	L79	L80	L81	L82	L83	L84	L86	L87	L88	L89	L90
Pro rata	430.15	501.85	215.08	71.69	71.69	358.46	143.38	71.69	71.69	71.69	286.77	71.69	71.69
PTDF slack bus 12	745.13	865.37	367.55	119.32	111.78	610.07	243.69	121.58	121.45	121.45	484.86	121.05	121.01
ZS slack bus 12	3631.07	4236.14	1789.34	553.56	457.26	3214.64	1305.12	658.80	661.90	645.70	2582.30	618.09	660.27
	L91	L92	L93	L94	L95	L96	L97	L98	L100	L101	L102	L103	L104
Pro rata	<b>L91</b> 71.69	<b>L92</b> 430.15	<b>L93</b> 107.54	<b>L94</b> 215.08	<b>L95</b> 286.77	<b>L96</b> 215.08	<b>L97</b> 107.54	<b>L98</b> 215.08	<b>L100</b> 215.08	<b>L101</b> 143.38	L102 107.54	L103 143.38	L104 215.08
Pro rata PTDF slack bus 12	<b>L91</b> 71.69 120.96	L92 430.15 725.35	L93 107.54 181.23	L94 215.08 362.27	L95 286.77 483.29	L96 215.08 362.73	L97 107.54 180.20	L98 215.08 359.24	L100 215.08 361.39	L101 143.38 241.29	L102 107.54 181.22	L103 143.38 240.92	L104 215.08 361.39
Pro rata PTDF slack bus 12 ZS slack bus 12	L91           71.69           120.96           662.21	L92 430.15 725.35 3886.28	L93 107.54 181.23 982.19	L94 215.08 362.27 1953.12	L95           286.77           483.29           2608.39	L96           215.08           362.73           1902.85	L97 107.54 180.20 897.23	L98 215.08 359.24 1785.82	L100 215.08 361.39 1933.55	L101 143.38 241.29 1319.95	L102 107.54 181.22 981.55	L103           143.38           240.92           1354.56	L104           215.08           361.39           2133.30
Pro rata PTDF slack bus 12 ZS slack bus 12	L91           71.69           120.96           662.21           L105	L92 430.15 725.35 3886.28 L106	L93 107.54 181.23 982.19 L107	L94 215.08 362.27 1953.12 L108	L95 286.77 483.29 2608.39 L109	L96           215.08           362.73           1902.85           L111	L97 107.54 180.20 897.23 L112	L98           215.08           359.24           1785.82           L113	L100 215.08 361.39 1933.55 L114	L101 143.38 241.29 1319.95 L115	L102 107.54 181.22 981.55 L116	L103 143.38 240.92 1354.56 L117	L104           215.08           361.39           2133.30           L118
Pro rata PTDF slack bus 12 ZS slack bus 12 Pro rata	L91         71.69         120.96         662.21         L105         215.08	L92 430.15 725.35 3886.28 L106 286.77	L93           107.54           181.23           982.19           L107           358.46	L94 215.08 362.27 1953.12 L108 143.38	L95 286.77 483.29 2608.39 L109 71.69	L96           215.08           362.73           1902.85           L111           71.69	L97 107.54 180.20 897.23 L112 43.02	L98           215.08           359.24           1785.82           L113           71.69	L100           215.08           361.39           1933.55           L114           71.69	L101 143.38 241.29 1319.95 L115 143.38	L102           107.54           181.22           981.55           L116           71.69	L103         143.38         240.92         1354.56         L117         143.38	L104         215.08         361.39         2133.30         L118         215.08
Pro rata PTDF slack bus 12 ZS slack bus 12 Pro rata PTDF slack bus 12	L91         71.69         120.96         662.21         L105         215.08         361.39	L92           430.15           725.35           3886.28           L106           286.77           481.85	L93         107.54         181.23         982.19         L107         358.46         602.31	L94           215.08           362.27           1953.12           L108           143.38           240.92	L95 286.77 483.29 2608.39 L109 71.69 120.46	L96           215.08           362.73           1902.85           L111           71.69           120.46	L97           107.54           180.20           897.23           L112           43.02           72.28	L98           215.08           359.24           1785.82           L113           71.69           -9.69	L100 215.08 361.39 1933.55 L114 71.69 -42.33	L101         143.38         241.29         1319.95         L115         143.38         -84.47	L102 107.54 181.22 981.55 L116 71.69 107.38	L103           143.38           240.92           1354.56           L117           143.38           0.00	L104         215.08         361.39         2133.30         L118         215.08         432.44



Security Cost Allocation by Three Methods

Figure 7.9 Comparison between three methods on security cost allocation in the 118-bus test system

Figure 7.9 indicates that the ZS method gives a distinctly different result on the security cost allocation compared with other two methods. The amounts of the allocated security cost to loads by the ZS method is not at the same order of magnitude with the other two methods. But if observe very carefully, there are still some noticeable details in such above figure. One is that in the PTDF method, allocated security cost on load 23 and load 32 are two negative peaks on the curve. In the ZS method, load 23 and load 32 are also the negative peaks showed on the ZS curve. Another one is from the load interval from load 45 to load 118, the curves of both the PTDF method and the ZS method actually have similar tendencies. The ZS method curve is up and down following the curve of the PTDF method. But the allocated security cost specifically to each load in the ZS method is almost a few times of the values in the PTDF method.

#### 7.3.6. Discussion

In the case studies of this chapter, the bus impendence matrix sensitivity factor is utilized to allocate the security cost due to transmission congestion management. Through four test systems including a 14-bus, a 30-bus, a 57-bus and a 118-bus system, the performance of the ZS based security cost allocation method is found out, which is then compared with the former two methods in chapter 5 under the same testing conditions. Discussions about the four groups of testing results are informed as follows:

In the 14-bus test system case, it is found that plans of security cost allocation through three methods are almost the same. But in the 30-bus test system case, results of security cost allocation by three methods start to be distinct. It can be observed that the ZS method has another idea to consider the power flow contribution percentage of each load to the congested line compared with the PTDF method. Things turn more complicated in the 57-bus test system case because except for the two negative peaks, the ZS method curve is following the tendency of the pro rata method curve. However, the positions of the two negative peaks in the ZS method curve are the two positive peaks of the PTDF method curve. It can be concluded that in the 57-bus test system the ZS method has the opposite logic with the PTDF method on power flow contribution calculation. If it is acceptable for security cost allocation results by the ZS method in the first three test systems, the allocation result of the ZS method in the 118-bus test system seems unreasonable. It is because the power flow contribution percentage of each load to the congestion calculated by the ZS method is so exaggerated no matter if positive values or the negative values, which leads to the values of security cost allocated to load to become exaggerated as well. Since the ZS method considers nearly half of loads have contribution to congestion relief so they should be rewarded. The loads who are considered by the ZS factors as the contributors of the transmission congestion should pay to recover not only the security cost allocated to the loads who have contribution to the congestion looks unreasonably high.

The comparisons between the three security cost allocation methods show the advantages and disadvantages of each method. The pro rata method is simple but not valid because it is unable to find out the contribution of each load to the transmission congestion. The PTDF method can find the power flow contribution percentage of each load to the transmission congestion. Nevertheless, since the dependency of slack bus selection, both the PTDF method and the ZS method will result in different allocation of security cost to the load and can be highly dependent on slack bus position. Furthermore, as the power system becomes large, the security cost allocation results of the ZS method appear unreasonable and unacceptable. Hence the system operator should decide to choose the method to allocate the security cost according to different power system conditions.

### 7.4. Summary

In this chapter, the mechanism and the mathematic model of the proposed allocation method with the bus impedance matrix based sensitivity factor are described. Case studies including experimental results by the ZS method in four modified IEEE test systems: an IEEE-14 bus system, an IEEE-30 bus system, an IEEE-57 bus system and an IEEE-118 bus system are shown. The comparisons among three allocation methods which are pro rata method, PTDF method with first generator bus as the slack bus and the ZS method with first generator bus as the slack bus and the ZS method with first generator bus as the slack bus and the ZS method with first generator bus as the slack bus and the ZS method with first generator bus as the slack bus are obtained and discussed. The conclusion is that each above method has pros and cons on the security cost allocation and there is no a perfect way to solve all the security cost allocation. It should be based on the specific condition when choose the method to allocate security cost due to congestion management.

## **Chapter 8**

## **Conclusion and Future Work**

#### 8.1. Conclusion

This thesis has achieved the goals of proposing a generation re-dispatch method to relieve transmission congestion and developing a security cost allocation method to balance the congestion management and the resulted security cost in deregulated electricity market. The re-dispatch method is actually a minimum cost optimization combined with the incremental and decremental bids submission mechanism for the generators, at the same time, restricted by transmission constraints and generation constraints. Two transmission sensitivity factors, namely the power transfer distribution factor (PTDF) and the bus impedance based sensitivity factor (ZS), are involved in the transmission capacity constraint calculation, which are based on the power flow contribution of each node to the congested transmission line calculated respectively by above two factors. The security cost allocation method allocates the security cost, which is the result of the congestion, to the loads based on the power flow contribution calculated respectively by above two factors as well.

The algorithm of the PTDF has proved that this sensitivity factor is dependent on the location of the slack bus selection. Case studies in chapter 4 attempt to verify the redispatch results are affected by the location of slack bus selection. Results show that the proposed method is capable under each circumstance. However, different results by selecting different generator bus as the slack bus have been obtained including the redispatched MW amount of each generator and the corresponding total security cost due to the congestion management. It is proved that the PTDF-based re-dispatch method has the risk of inaccuracy if there is only one bus chosen as the slack bus and if select all the generators as the slack bus respectively, the complexity of the method will increase. The impedance matrix has been proved to be dependent on the location of slack bus selection. Consequently, the bus impedance based sensitivity factor, which is based on the impedance matrix, is also relying on the location of slack bus selection. Case studies in chapter 5 try to investigate the performance comparison between the ZS-based redispatch method and the PTDF-based re-dispatch method, with selecting the same location of slack bus and facing with the same level of overloading of the congested transmission line. Experimental results indicate that the number of generators involved in the ZS-based re-dispatch method is more than the number in the PTDF-based method, which eventually leads to the total resulted security cost due to the ZS-based re-dispatch method greater than the PTDF-based method. It is also showed that in all test systems, with the intense level of overloading of congestion, the security cost calculated by the ZS-based method is always beyond the PTDF-based method, and the distinction between their resulted security costs becomes larger.

The PTDF-based security cost allocation method also results in different allocation results with selecting different locations of slack bus. Case studies in chapter 6 aim to verify it. A pro rata method is added to the tests to provide a parallel comparison to highlight the advantage of the PTDF-based method. Experimental results indicate that one particular load is allocated different security costs when selecting different locations of slack bus. It is proved that the dependency of location of slack bus selection enhances the risks of inaccuracy and complexity of the security cost allocation by the PTDF-based method. Only considering one slack bus selection is not convincing while taking all the slack buses locations into consideration is time-consuming. Nevertheless, compared with the pro rata method, which is only based on the MW amount percentage, the PTDF-based method is still a fair way to allocate the security cost since it can find out the power flow contribution of each load particular to the congested transmission line.

The ZS-based security cost allocation method also has the dependency of the location of slack bus selection. Case studies in chapter 7 are conducted to provide comparisons among three security cost allocation methods: the pro rata method, the PTDF-based method with the first generator bus as the slack bus and the ZS-based method with the same slack bus. All target security cost amounts are chosen from the costs calculated by the PTDF-based re-dispatch method with the first slack bus selection in chapter 4. Experimental results show that three allocation methods have different ideas to allocate security cost to the loads. It should be noted that in the large power system, the ZS-based method calculates unreasonable allocation results since some of the loads are allocated so much costs while some other loads are allocated too much costs in negative values. The negative values of allocated costs can be regarded as the rewards to the loads by the system operator, but the extreme numbers are unacceptable. It is proved that when implemented in large power systems, the ZS-based security cost allocation method has the risk of providing incomprehensible solutions.

#### 8.2. Future Work

This thesis has developed a proposed congestion management scheme with two different power flow sensitivity factors for both generation re-dispatching and corresponding security cost allocation. However, due to the time constraint, some further issues are not considered in this research and some mechanisms of the proposed method can be further investigated.

• In this thesis, all experimental simulations of power system environment are based on the software package Matpower (version 6.0), which is implemented in the software Matlab. Power flow optimization problems are solved by using it no matter if it is in 14-bus system or 118-bus system. For the experimental purpose, it may be a good choice to use but for the industry standard studies, its limitation might be exposed since it is usually used to solve problems under ideal and steady conditions while in reality, power system is always complex and varying. In the future, some state-of-the-art tools which could meet industrial requirements should be utilized to simulate power system environment and to solve power flow problems. For example, the software CPLEX which is an optimizer implemented in the C programming language to solve mixed-integer programming problems.

- As known, in this thesis, four modified IEEE-bus test systems are used to build as the power system bases for power flow optimizations and congestion management implementations. But these IEEE-bus models have been used for many years and may not model modern power systems with sufficient accuracy. Furthermore, some parameters of power system simulations in this thesis are set arbitrarily such as bid price of each generator, generation capacity of each generator, active power consumption of each load, and transfer capacity of each transmission line. For a more convincing approach, in the future, a realistic power system case will be found and built as an experimental model. Parameters of such power system model will be more close to the real situation, which will also become more logical and real.
- The experiments in this thesis are only based on the DC optimal power flow since its fast calculating speed and easy convergence. All the transmission power flows are considered as lossless since the DC assumptions. In the future work, the AC optimal power flow will be utilized to simulate a more realistic power system environment. The power losses and reactive power will be taken into considerations. Two transmission congestion management schemes which are proposed in this thesis to re-dispatch generation and allocate security cost will be upgraded to adapt to the AC power flow situations.
- In all the test scenarios, there is only one transmission congestion occurs a time. Number of congested transmission line in each test scenario is restricted to one only in order to highlight the sensitivity method efficiency on one particular line. However,

in the reality, there will be multi-congestions occur at the same time during the power system operation. In the future work, both the PTDF-based method and the ZS-based method will be improved to deal with multi-congestion problems including the generation re-dispatching and the corresponding security cost allocation. Under the multi-congestion situations, power flow contribution of each generator and each load will be re-evaluated in order to decide new generation reschedules and new allocation of security cost.

- In the proposed congestion management scheme mechanism, only generators have the opportunity to submit incremental and decremental bids, who will be involved in the congestion relief action because the demand side is assumed to be inelastic. In the future work, loads will be considered as elastic, which means loads in the power system enable to take part in the adjustment bids submission to help the system operator has more options to adjust the power flow on the congested line from both ends of the power system. The corresponding security cost allocation scheme should be designed as, for example, the load who helped the system operator relieve the congestion should not be allocated any security cost but even received rewards from system operator.
- As known, the PTDF method has the drawbacks of inaccuracy and complexity due to the dependency of slack bus selection. But there are still some specific slack bus selections, which are seemed as reasonable solutions on generation re-dispatching and security cost allocation. In the future work, how to correctly choose the generator bus as the slack bus in the PTDF method can be considered as an investigative topic to improve the exist PTDF-based method on congestion management. If the appropriate slack bus can be chosen in a quick way, the PTDF-based method can still be regarded as a suitable approach to manage transmission congestion.

It is found that the ZS-based method always involves more generators into redispatch stage, which leads to corresponding security cost more than the PTDFbased method. In the aspect of security cost allocation, when implemented in large power system, the ZS-based method gives unreasonable outcomes. For the moment, this problems are attributed to the ZS-based sensitivity factors because both generation re-dispatching and security cost allocation in the ZS-based method are based on these sensitivity factors. In the future work, the deeper reason behind this will be researched and discovered so the ZS-based method could be improved to obtain a better performance on congestion management. Since the ZS-based method is also dependent on the location of slack bus selection, if its performance is better than the PTDF-based method, it can become a preferred method for the system operator to solve the transmission congestion issues.

## References

[1] Kwok Lun Lo and Yee Shan Yuen, "Deregulation of Electric Utilities", in *Power System Restructuring and Deregulation*, book, Wiley, pp. 50-74.

[2] David Newbery, "High Level Principles for Guiding GB Transmission Charging and Some of the Practical Problems of Transition to an Enduring Regime", *Paper for Ofgem*, Energy Policy Research Group, University of Cambridge, 27th May, 2011.

[3] G. Simmonds, "Regulation of the UK Electricity Industry", *CRI Industry Brief*, University of Bath, Desktop published by Jan Marchant 2002 Edition.

[4] Perekhodtsev, Dmitri, and Guido Cervigni. "UK Transmission Congestion Problem: Causes and Solutions", *SSRN Scholarly Paper*. Rochester, NY: Social Science Research Network, February 9, 2010

[5] E. S. Amundsen and L. Bergman, "Why has the Nordic electricity market worked so well?", *Utilities Policy*, vol. 14, pp. 148-157, 2006.

[6] J. Tong, "Overview of PJM energy market design, operation and experience", in *Electric Utility Deregulation, Restructuring and Power Technologies*, 2004. (*DRPT* 2004). *Proceedings of the 2004 IEEE International Conference on*, 2004, pp. 24-27.

[7] J. W. M. Cheng, F. D. Galiana, and D. T. McGillis, "Studies of bilateral contracts with respect to steady-state security in a deregulated environment", in *Proceedings of the 20th International Conference on Power Industry Computer Applications*, 1997, pp. 31–36.

[8] H. Glavitsch and F. Alvarado, 'Management of multiple congested conditions in unbundled operation of a power system', in *Proceedings of the 20th International Conference on Power Industry Computer Applications*, 1997, pp. 374–380.

[9] J. Sun, "Demand Side Approaches for Congestion Management in Electricity Market",Ph.D. Thesis, Dept. EEE. Eng., Univ. Strathclyde, 2014.

[10] A. Pillay, S. Prabhakar Karthikeyan, and D. P. Kothari, 'Congestion management in power systems – A review', *International Journal of Electrical Power & Energy Systems*, vol. 70, pp. 83–90, Sep. 2015.

[11] A. Singla, K. Singh, and V. K. Yadav, "Transmission congestion management in deregulated environment: A bibliographical survey", in *International Conference on Recent Advances and Innovations in Engineering (ICRAIE-2014)*, 2014, pp. 1–10.

[12] Luonan Chen, H. Suzuki, T. Wachi, and Y. Shimura, 'Components of nodal prices for electric power systems', *IEEE Transactions on Power Systems*, vol. 17, no. 1, pp. 41–49, Feb. 2002.

[13] G. Hamoud and I. Bradley, 'Assessment of transmission congestion cost and locational marginal pricing in a competitive electricity market', *IEEE Transactions on Power Systems*, vol. 19, no. 2, pp. 769–775, May 2004.

[14] Tong Wu, Z. Alaywan, and A. D. Papalexopoulos, 'Locational marginal price calculations using the distributed-slack power-flow formulation', *IEEE Transactions on Power Systems*, vol. 20, no. 2, pp. 1188–1190, May 2005.

[15] M. BjÃ, rndal, K. JÃ, rnsten, and V. Pignon, 'Congestion management in the Nordic power market - counter purchases and zonal pricing', *Competition and Regulation in Network Industries*, vol. 4, no. 3, pp. 271–293, 2003.

[16] Xingwang Ma, D. I. Sun, and K. W. Cheung, 'Evolution toward standardized market design', *IEEE Transactions on Power Systems*, vol. 18, no. 2, pp. 460–469, May 2003.

[17] M. Shahidehpour, H. Yamin, and Z. Li, *Market Operations in Electric Power Systems: Forecasting, Scheduling, and Risk Management.* John Wiley & Sons, 2003. [18] C. Y. Choo, N. C. Nair, and B. Chakrabarti, 'Impacts of Loop Flow on Electricity Market Design', in *2006 International Conference on Power System Technology*, 2006, pp. 1–8.

[19] I. Androcec and I. Wangensteen, 'Different Methods for Congestion Management and Risk Management', in 2006 International Conference on Probabilistic Methods Applied to Power Systems, 2006, pp. 1–6.

[20] R. Gnanadass, N. Prasad Padhy, and T. G Palanivelu, 'A new method for the transmission congestion management in the restructured power market', *Journal of Electrical Engineering, Electrika*, vol. 9, Jan. 2007.

[21] T. K. P. Medicherla, R. Billinton, and M. S. Sachdev, 'Generation Rescheduling and Load Shedding to Alleviate Line Overloads-Analysis', *IEEE Transactions on Power Apparatus and Systems*, vol. PAS-98, no. 6, pp. 1876–1884, Nov. 1979.

[22] T. Niimura and Y. Niu, 'Transmission congestion relief by economic load management', in *IEEE Power Engineering Society Summer Meeting*, 2002, vol. 3, pp. 1645–1649 vol.3.

[23] C. Barbulescu, S. Kilyeni, D. P. Cristian, and D. Jigoria-Oprea, 'Congestion management using open power market environment electricity trading', in *45th International Universities Power Engineering Conference UPEC2010*, 2010, pp. 1–6.

[24] K. Singh, N. P. Padhy, and J. D. Sharma, 'Bus impedance matrix based approach for congestion management in deregulated environment', in *IEEE PES T D 2010*, 2010, pp. 1–6.

[25] Kopsakangas Savolainen, M. and Svento, R. "Real-Time Pricing in the Nordic Power markets", *Energy Economics 34*, 1131–1142, 2012.

[26] Y. Yuen, "Congestion Management and its Implementation Using Information Technologies", Ph.D. Thesis, Dept. EEE. Eng., Univ. Strathclyde, 2001.

[27] James Sweeney, "The California Electricity Crisis: Lessons for the Future", National Academy of Engineering, *the Bridge* 32, no. 2, 2002.

[28] M. H. J. B. K. Bhattacharya, J. E. Daalder, *Operation of Restructured Power Systems*.Boston: Kluwer Academic Publishers, 2001.

[29] Kopsakangas-Savolainen, Maria, and Rauli, Svento. "Real-Time Pricing: An Application to the Nordic Power Markets", *Modern Energy Markets*, 29–44. Green Energy and Technology. Springer London, 2012.

[30] M. Pollitt, "Electricity reform in Chile-lessons for developing countries", *J. Network Ind.*, vol. 5, p. 221, 2004.

[31] Bolle, F and Breitmoser, Y, "On the allocative efficiency of ownership unbundling",Department of Business Administration and Economics Discussion Paper, No. 255.European University Viadrina Frankfurt (Oder), January, 2006.

[32] C. Zhao, "Allocation of Wheeling Charges and Congestion Management for Cross-Border Trading", Ph.D. Thesis, Dept. EEE. Eng., Univ. Strathclyde, 2012.

[33] K. Yang, "Optimal Procurement and Pricing of Reactive Power Ancillary Services in Competitive Electricity Market", Ph.D. Thesis, Dept. EEE. Eng., Univ. Strathclyde, 2010.

[34] M.P. Abdullah, "Congestion Management and Security Cost Allocation in Electricity Market", Ph.D. Thesis, Dept. EEE. Eng., Univ. Strathclyde, 2008

[35] M. G. Pollitt, "Lessons from the History of Independent System Operators in the Energy Sector, with applications to the Water Sector," *Faculty of Economics, Working Paper*, Aug. 2011.

[36] United States of America Federal Energy Regulatory Commission. (1999, Accessed on 31<sup>st</sup> January 2011). *Regional Transmission Organizations, RTO Final Rule Part 1*. Available: https://www.ferc.gov/legal/maj-ord-reg/land-docs/RM99-2A.pdf [37] Foreign & Commonwealth Office. (2014, Accessed on 26 March 2014). UK's Electricity Market Reform and Contract for Difference system. Available: https://assets.publishing.service.gov.uk/government/uploads/system/uploads/attachment \_data/file/296341/EMR\_CfD\_Report\_.pdf

[38] D. M. Newbery and M. G. Pollitt, "The restructuring and privatization of the U.K. electricity supply - was it worth it?", The World Bank, 17080, Sep. 1997.

[39] Horrocks, S and Lean, T, "An oral history of the electricity supply industry: Scoping study for proposed National Life Stories Project", British Library, 2011.

[40] Newbery, D.M, "The UK experience: privatization with market power", CEPR monitoring European deregulation: Electricity, 8 February 1999.

[41] Newbery, D.M, "Pool Reform and Competition in Electricity", in M. Beesley (ed.)Regulating Utilities: Understanding the Issues, London: Institute of Economic Affairs, 1998.

[42] D. S. Kirschen, "Market power in the Electricity Pool of England and Wales", in *IEEE Power Engineering Society Winter Meeting. Conference Proceedings*, pp. 36–40 vol.1, 2001.

 [43] NERA. (2002, Accessed on 27/06/2002). A Comparative Study of the Electricity Markets in UK, Spain and Nord Pool. Available: https://www.nera.com/content/dam/nera/publications/archive1/5566.pdf

[44] P. Cramton, "Electricity market design", *Oxf Rev Econ Policy*, vol. 33, no. 4, pp. 589–612, Nov. 2017.

[45] M. A. Crew, *Regulation under Increasing Competition*: Kluwer Academic Publishers, 1999.

[46] J. Zhao, K. L. Lo, and J. Lu, "Variously worldwide types of deregulated electricity markets and their respective transmission congestion management schemes", in *2016 51st International Universities Power Engineering Conference (UPEC)*, 2016, pp. 1–6.

[47] L. Bergman *et al.*, *A European Market for Electricity?* Centre for Economic Policy Research, 1999.

[48] S. Hesmondhalgh, "Is NETA the blueprint for wholesale electricity trading arrangements of the future?", *Power Systems, IEEE Transactions on*, vol. 18, pp. 548-554, 2003.

[49] R. Dettmer, "Living with NETA [New Electricity Trading Arrangements]", *IEE Review*, vol. 48, pp. 32-36, 2002.

[50] Ofgem. (2005, Accessed on 04/02/2005). *BETTA User Guide: A summary of the new British Electricity Trading and Transmission Arrangements (BETTA) and a high-level guide to the key activities required to implement the new arrangements and run-off the pre-BETTA arrangements.* Available: https://www.ofgem.gov.uk/sites/default/files/docs/2005/02/9549-2605.pdf

[51] ELEXON. (2018, Accessed on 25/09/2018). Glossary Term: British Electricity Trading and Transmission Arrangements. Available: https://www.elexon.co.uk/glossary/british-electricity-trading-and-transmissionarrangements/

[52] R. Green, "Are the British electricity trading and transmission arrangements futureproof?", *Utilities Policy*, vol. 18, no. 4, pp. 186–194, Dec. 2010.

[53] National Audit Office. (2014, Accessed on May 2014). *Electricity Balancing Service*.
 Available: https://www.nao.org.uk/wp-content/uploads/2014/05/Electricity-Balancing-Services.pdf

[54] Tovey, N. K, (2005 Accessed on 06/11/2006). Developments in the Electricity Markets in the UK: the move towards BETTA. Available: https://archive.uea.ac.uk/~e680/energy/energy\_links/neta/nktovey\_moscow\_2005.pdf

[55] ELEXON. (2016, Accessed on 24 October 2016). *Imbalance Pricing Guidance*. Available:https://www.elexon.co.uk/wpcontent/uploads/2016/10/Imbalance\_Pricing\_gui dance\_v11.0.pdf

[56] National Grid. (2017, Accessed on 17 Jan 2017). November 2016 *Monthly Balancing Service Summary 2016/2017*. Available: http://www2.nationalgrid.com/UK/Industry-information/Electricity-transmission-operational-data/Report-explorer/Services-Reports/

[57] O. Gjerde, "The deregulated Nordic electricity market-10 years of experience", presented at the Transmission and Distribution Conference and Exhibition 2002: Asia Pacific. IEEE/PES, 2002

[58] Nord Pool. (2018, Accessed on 26/09/2018). *History*. Available: https://www.nordpoolgroup.com/About-us/History/

[59] Nord Pool. (2016, Accessed on 14/03/2016). Europe's Leading Power Market.
Available: http://www.ephzhb.ba/wp-content/uploads/02-HENRIETTE-LOLAND-Nord-Pool-Group.pdf

[60] Nord Pool. (2018, Accessed on 26/09/2018). *The Nordic Electricity Exchange and the Nordic Model for a Liberalized Electricity Market*. Available: https://www.nordpoolgroup.com/globalassets/download-center/rules-andregulations/the-nordic-electricity-exchange-and-the-nordic-model-for-a-liberalizedelectricity-market.pdf

[61] L. Bergman, "The Nordic electricity market-continued success or emerging problems?", *Swedish economic policy review*, vol. 9, pp. 51-88, 2002.

[62] J. Peljo, "Futures Pricing in the Nordic Electricity Market", MA thesis. Finland: Dept. of Finance, School of Business, Aalto University, 2013.

[63] Nord Pool. (2018, Accessed on 26/09/2018). *Annual report 2017*. Available: https://www.nordpoolgroup.com/globalassets/download-center/annual-report/annual-report-nord-pool\_2017.pdf

[64] Swedish Energy Markets Inspectorate. (2016, Accessed on Nov 2016). *Area price hedging and the Nordic market model*. Available: https://www.ei.se/Documents/Publikationer/rapporter\_och\_pm/Rapporter%202016/Ei\_R 2016\_11.pdf

[65] Mundaca, L. T., C. Dalhammar, D. Harnesk, "The Integrated NORDIC Power Market and the Deployment of Renewable Energy Technologies: Key Lessons and Potential Implications for the Future ASEAN Integrated Power Market" in Kimura, S., H. Phoumin and B. Jacobs (eds.), *Energy Market Integration in East Asia: Renewable Energy and its Deployment into the Power System*, ERIA Research Project Report, 2012-26, Jakarta: ERIA. pp.25-97.

[66] PJM. (2018, Accessed on 27/09/2018). *PJM History*. Available: https://www.pjm.com/about-pjm/who-we-are/pjm-history.aspx

[67] PJM. (2018, Accessed on 27/09/2018). *Who We Are*. Available: https://www.pjm.com/about-pjm/who-we-are.aspx

[68] PJM. (2018, Accessed on 27/09/2018). *Annual Report 2017*. Available: https://www.pjm.com/-/media/about-pjm/newsroom/annual-reports/2017-annual-report.ashx?la=en

[69] PJM. (2018, Accessed on 27/09/2018). *Market for Electricity*. Available: https://learn.pjm.com/electricity-basics/market-for-electricity.aspx

[70] PJM. (2018, Accessed on 27/09/2018). *Ancillary Services Market*. Available: https://learn.pjm.com/three-priorities/buying-and-selling-energy/ancillary-services-market.aspx

[71] PJM. (2018, Accessed on 27/09/2018). *Financial Transmission Rights FAQs*. Available: https://learn.pjm.com/three-priorities/buying-and-selling-energy/ftr-faqs.aspx#faq-box-text0

 [72] PJM. (2018, Accessed on 27/09/2018). PJM Markets. Available: https://learn.pjm.com/three-priorities/buying-and-sellingenergy/~/media/3B551D5617594B159FD0EFB898CBB336.ashx

[73] Z. Fan, T. Horger, J. Bastian, "Current and emerging challenges in PJM energy market", *Proc. Transmission and Distribution Conf. Expo. (PES)*, pp. 1-7, 2010.

[74] A. L. Ott, "Experience with PJM market operation, system design, and implementation", *Power Systems, IEEE Transactions on*, vol. 18, pp. 528-534, 2003.

[75] C. Pechman and P. P. I. o. California, *California's Electricity Market: A Post-crisis Progress Report*: Public Policy Institute of California, 2006.

[76] C.-K. Woo, "What went wrong in California's electricity market?", *Energy*, vol. 26, no. 8, pp. 747–758, Aug. 2001.

[77] C. Weare, "The California Electricity Crisis: Causes and Policy Options", Public Policy Institute of California, 2003

[78] A. Kumar, S. C. Srivastava, and S. N. Singh, "Congestion management in competitive power market: A bibliographical survey", *Electric Power Systems Research*, vol. 76, no. 1, pp. 153–164, Sep. 2005.

[79] N. I. Yusoff, A. A. M. Zin, and A. B. Khairuddin, "Congestion management in power system: A review", in 2017 3rd International Conference on Power Generation Systems and Renewable Energy Technologies (PGSRET), 2017, pp. 22–27.

[80] A. Pillay, S. Prabhakar Karthikeyan, and D. P. Kothari, "Congestion management in power systems – A review", *International Journal of Electrical Power & Energy Systems*, vol. 70, pp. 83–90, Sep. 2015.

[81] E. Bompard, P. Correia, G. Gross, and M. Amelin, "Congestion-management schemes: a comparative analysis under a unified framework", *Power Systems, IEEE Transactions on*, vol. 18, pp. 346-352, 2003.

[82] F. C. Schweppe, M. C. Caramanis, R. D. Tabors, and R. E. Bohn, *Spot Pricing of Electricity*: Springer, 1988.

[83] M. C. Caramanis, R. E. Bohn, and F. C. Schweppe, "Optimal Spot Pricing: Practice and Theory", *IEEE Transactions on Power Apparatus and Systems*, vol. PAS-101, no. 9, pp. 3234–3245, Sep. 1982.

[84] P. Holmberg, E. Lazarczyk, "Congestion management in electricity networks: Nodal zonal and discriminatory pricing", *EPRG Workshop Paper Series*, Apr. 2012.

[85] S. Frank, S. Rebennack, "An introduction to optimal power flow: Theory formulation and examples", *IIE Trans.*, vol. 48, no. 12, pp. 1172-1197, 2016.

[86] V. Sarkar and S. Khaparde, "DCOPF-based marginal loss pricing with enhanced power flow accuracy by using matrix loss distribution", *Power Systems, IEEE Transactions on*, vol. 24, pp. 1435-1445, 2009.

[87] L. Fangxing and B. Rui, "DCOPF-Based LMP Simulation: Algorithm, Comparison with ACOPF, and Sensitivity", *Power Systems, IEEE Transactions on*, vol. 22, pp. 1475-1485, 2007.

[88] K. Van den Bergh, E. Delarue, and W. D'haeseleer, DC Power Flow in Unit Commitment Models, TME Working Paper-Energy and Environment, Tech. Rep., 2014.

[89] Murali Matcha, Sailaja Kumari, Matam, Sydulu, and Maheswarapu, "LMP calculation with distributed loss using GA based DCOPF", *J. Electrical Systems* vol. 8, no. 3 pp. 292-303, 2012.

[90] E. Litvinov, T. Zheng, G. Rosenwald, and P. Shamsollahi, "Marginal loss modeling in LMP calculation", *Power Systems, IEEE Transactions on*, vol. 19, pp. 880-888, 2004.

[91] T. Orfanogianni and G. Gross, "A general formulation for LMP evaluation", *Power Systems, IEEE Transactions on*, vol. 22, pp. 1163-1173, 2007.

[92] W. E. C. Council, "Energy Imbalance Market Functional Specification", 2011.

[93] E. Hausman, A. P. P. Association, and S. E. Economics, *LMP Electricity Markets: Market Operations, Market Power, and Value for Consumers*: American Power Association, 2007.

[94] H. Li and L. Tesfatsion, "ISO net surplus collection and allocation in wholesale power markets under LMP", *Power Systems, IEEE Transactions on*, vol. 26, pp. 627-641, 2011.

[95] Nord Pool. (2018, Accessed on 08/10/2018). *Bidding Areas*. Available: https://www.nordpoolgroup.com/the-power-market/Bidding-areas/

[96] Mette Bjorndal, Kurt Jornsten, and Virginie Pignon, "Congestion management in the Nordic power market: counter purchases and zonal pricing", *Journal of Network Industries*, vol. 4, pp. 271-292, 2003.

[97] Yong T. Yoon, Santosh G. Raikar, and Marija D. Ilic, "Congestion management for large electric power systems", Energy Laboratory Publication, MIT, May 2000.

[98] Z. Alaywan, T. Wu, and A. D. Papalexopoulos, "Transitioning the California market from a zonal to a nodal framework: an operational perspective", in *Power Systems Conference and Exposition, 2004. IEEE PES*, 2004, pp. 862-867.

[99] S. M. Harvey and W. W. Hogan, "Nodal and zonal congestion management and the exercise of market power", Harvard University Working Paper, 2000.

[100] R. S. Fang and A. K. David, "Transmission congestion management in an electricity market", *IEEE Transactions on Power Systems*, vol. 14, no. 3, pp. 877–883, Aug. 1999.

[101] J. Fu and J. W. Lamont, "A combined framework for service identification and congestion management", *IEEE Transactions on Power Systems*, vol. 16, no. 1, pp. 56–61, Feb. 2001.

[102] F. Sousa, F. Lopes, and J. Santana, "Multi-agent Electricity Markets: A Case Study on Contracts for Difference", in 2015 26th International Workshop on Database and Expert Systems Applications (DEXA), 2015, pp. 86–90.

[103] Z. Nie, F. Gao, J. Wu, X. Guan, and K. Liu, "Contract for difference (CfD) energy decomposition model for maximizing social benefit in electricity market", in *2016 12th World Congress on Intelligent Control and Automation (WCICA)*, 2016, pp. 2449–2454.

[104] J. Sun, "US financial transmission rights: Theory and practice," Economics Department, Iowa State University2006.

[105] V. G. Umale and S. B. Warkad, "Locational marginal pricing based financial transmission right", in 2017 IEEE International Conference on Circuits and Systems (ICCS), 2017, pp. 459–462.

[106] V. Sarkar and S. A. Khaparde, "A Comprehensive Assessment of the Evolution of Financial Transmission Rights", *IEEE Transactions on Power Systems*, vol. 23, no. 4, pp. 1783–1795, Nov. 2008.

[107] M. P. Abdullah, M. Y. Hassan, and F. Hussin, "Assessment of contribution-based congestion cost allocation using AC and DC for bilateral market", in *2010 IEEE International Conference on Power and Energy*, 2010, pp. 897–901.

[108] Y. S. C. Yuen, K. L. Lo, "Simulations of bilateral energy markets using MATLAB", *Int. J. Comput. Math. Electr. Electron. Eng*, vol. 22, no. 2, pp. 424-443, 2003.

[109] A. Kumar, S. C. Srivastava, and S. N. Singh, "A zonal congestion management approach using real and reactive power rescheduling", *IEEE Transactions on Power Systems*, vol. 19, no. 1, pp. 554–562, Feb. 2004.

[110] S. Dutta and S. P. Singh, "Optimal Rescheduling of Generators for Congestion Management Based on Particle Swarm Optimization", *IEEE Transactions on Power Systems*, vol. 23, no. 4, pp. 1560–1569, Nov. 2008.

[111] J. O. F. Silva and P. Cuervo, "Uplift Generation Cost Allocation in Pay-as-Bid Markets with Congested Transmission Networks", in *2006 IEEE PES Power Systems Conference and Exposition*, 2006, pp. 711–717.

[112] K. Singh, V. K. Yadav, N. P. Padhy, and J. Sharma, "Congestion Management Considering Optimal Placement of Distributed Generator in Deregulated Power System Networks", *Electric Power Components and Systems*, vol. 42, no. 1, pp. 13–22, Jan. 2014.

[113] X. Cheng and T. J. Overbye, "PTDF-based power system equivalents", *IEEE Transactions on Power Systems*, vol. 20, no. 4, pp. 1868–1876, Nov. 2005.

[114] A. G. Exposito, J. L. M. Ramos, and J. R. Santos, "Slack bus selection to minimize the system power imbalance in load-flow studies", *IEEE Transactions on Power Systems*, vol. 19, no. 2, pp. 987–995, May 2004.

[115] L. Jianfeng, "A New Method for Harmonic Penetration Study in Power Networks with Renewable Generation", Ph.D. Thesis, Dept. EEE. Eng., Univ. Strathclyde, 2014.

[116] C. L. Wadhwa, *Electrical Power Systems*, UK: New Academic Science Ltd, 2009, pp. 592-596.

[117] M. Bazrafshan and N. Gatsis, "Comprehensive Modeling of Three-Phase Distribution Systems via the Bus Admittance Matrix", *IEEE Transactions on Power Systems*, vol. 33, no. 2, pp. 2015–2029, Mar. 2018.

[118] A. K. Saxena and D. Anand Rao, "A new approach to bus impedance matrix building", *Computers & Electrical Engineering*, vol. 29, no. 1, pp. 55–65, Jan. 2003.

[119] I. Kim and R. G. Harley, "A study on power-flow and short-circuit algorithms capable of analyzing the effect of load current on fault current using the bus impedance matrix", *in 2016 IEEE Electrical Power and Energy Conference (EPEC)*, 2016, pp. 1–3.

[120] A. J. Conejo, J. Contreras, D. A. Lima, and A. Padilha-Feltrin, "Zbus Transmission Network Cost Allocation", *IEEE Transactions on Power Systems*, vol. 22, no. 1, pp. 342–349, Feb. 2007.

[121] J. D. Glover and M. S. Sarma, Power System Analysis and Design, 5th ed., Pacific Grove, CA: Brooks/Cole, 2012.

[122] M. P. Abdullah, M. Y. Hassan, and F. Hussin, "Congestion cost allocation in a poolbased electricity market", in *2008 IEEE 2nd International Power and Energy Conference*, 2008, pp. 1033–1037.

[123] M. E. Baran, V. Banunarayanan, and K. E. Garren, "Equitable allocation of congestion relief cost to transactions", *IEEE Transactions on Power Systems*, vol. 15, no. 2, pp. 579–585, May 2000.

[124] S.-C. Hsieh, C.-C. Chu, and H.-M. Wang, "Congestion cost allocation and congestion indices for a competitive electricity market", in *2002 IEEE International Conference on Industrial Technology*, 2002, vol. 2, pp. 854–859 vol.2.

[125] R. D. Zimmerman, C. E. Murillo-Sanchez. MATPOWER User's Manual, Version6.0. 2016. [Online]. Available: https://matpower.org/docs/MATPOWER-manual-6.0.pdf

[126] R. D. Zimmerman, C. E. Murillo-Sanchez, and R. J. Thomas, "MATPOWER: Steady-State Operations, Planning and Analysis Tools for Power Systems Research and Education," Power Systems, IEEE Transactions on, vol. 26, no. 1, pp. 12–19, Feb. 2011.

[127] H. H. Yan, "PTDF and TLR from a power marketer's perspective," 1999 IEEEPower Engineering Society Summer Meeting. Conference Proceedings (Cat. No.99CH36364), Edmonton, Alta., Canada, 1999, pp. 156-161 vol.1.

# Appendix

## A. PTDF and ZS Sensitivity Factors in Modified IEEE-14 Bus System

# Table A. PTDF and ZS Sensitivity Factors Associated with Congested TransmissionLine 3-4 in Modified IEEE-14 Bus System

Bus	<u>1</u>	2	<u>3</u>	4	5	<u>6</u>	7
PTDF slack 1	0	0.02735	0.467992	-0.151329	-0.103095	-0.118834	-0.142675
Bus	<u>8</u>	9	10	11	12	13	14
PTDF slack 1	-0.142675	-0.1380197	-0.13461	-0.12686	-0.12035	-0.12153	-0.13081
Bus	1	2	<u>3</u>	4	5	<u>6</u>	7
PTDF slack 2	-0.02735	0	0.440642	-0.178679	-0.130445	-0.146184	-0.170025
Bus	<u>8</u>	9	10	11	12	13	14
PTDF slack 2	-0.170025	-0.1653696	-0.16196	-0.15421	-0.1477	-0.14888	-0.15816
	· · · · · · · · · · · · · · · · · · ·						
Bus	1	2	<u>3</u>	4	5	<u>6</u>	7
PTDF slack 3	-0.46799	-0.44064	0	-0.619321	-0.571087	-0.586826	-0.610667
Bus	<u>8</u>	9	10	11	12	13	14
PTDF slack 3	-0.610667	-0.6060119	-0.602602	-0.594852	-0.588342	-0.58953	-0.5988
	·	·		·			
Bus	<u>1</u>	2	<u>3</u>	4	5	<u>6</u>	7
PTDF slack 6	0.118834	0.146184	0.586826	-0.032495	0.0157391	0	-0.023841

Bus	<u>8</u>	9	10	11	12	13	14
PTDF slack 6	-0.023841	-0.0191856	-0.015776	-0.008026	-0.001516	-0.0027	-0.01198
		·	·	·	·	·	
Bus	<u>1</u>	<u>2</u>	<u>3</u>	4	5	<u>6</u>	7
PTDF slack 8	0.142675	0.170025	0.610667	-0.008654	0.0395796	0.0238405	0
Bus	<u>8</u>	9	10	11	12	13	14
PTDF slack 8	0	0.00465493	0.0080646	0.0158148	0.0223245	0.02114	0.011863
	<u></u>						
Bus	1	2	3	4	5	<u>6</u>	7
ZS slack 1	0	-0.065096	-0.502803	0.1088657	0.0609936	0.078889	0.1014944
Bus	<u>8</u>	9	10	11	12	13	14
ZS slack 1	0.0988872	0.09910399	0.0960888	0.0874781	0.0811924	0.082909	0.093553

## **B.** PTDF and ZS Sensitivity Factors in Modified IEEE-30 Bus System

Bus	<u>1</u>	<u>2</u>	3	4	<u>5</u>	6	7
PTDF slack 1	0	-0.005474107	0.01572735	0.019335496	-0.0200072	-0.034518376	-0.028508711
Bus	<u>8</u>	9	10	11	12	<u>13</u>	14
PTDF slack 1	-0.036955563	-0.194812341	-0.281490099	-0.194812341	-0.613507156	-0.613507156	-0.565639667
Bus	15	16	17	18	19	20	21
PTDF slack 1	-0.528284695	-0.4747652	-0.340492078	-0.44196387	-0.390922	-0.364057844	-0.288409
Bus	22	23	24	25	26	27	28
PTDF slack 1	-0.290589	-0.438285424	-0.31798937	-0.22171	-0.22171	-0.16067	-0.04856
Bus	29	30		1	1	1	1
PTDF slack 1	-0.16067	-0.16067					
Bus	1	2	3	4	<u>5</u>	6	7
PTDF slack 2	0.005474107	0	0.02120145	0.024809603	-0.0145331	-0.029044269	-0.023034605
Bus	<u>8</u>	9	10	<u>11</u>	12	<u>13</u>	14
PTDF slack 2	-0.031481456	-0.189338234	-0.276015992	-0.189338234	-0.608033049	-0.608033049	-0.560165561
Bus	15	16	17	18	19	20	21
PTDF slack 2	-0.522810588	-0.4692911	-0.335017972	-0.436489763	-0.3854479	-0.358583737	-0.2829349
Bus	22	23	24	25	26	27	28
PTDF slack 2	-0.2851149	-0.432811318	-0.312515263	-0.21624	-0.21624	-0.1552	-0.04309
Bus	29	30		·	·	<u> </u>	·
PTDF slack 2	-0.1552	-0.1552					
Bus	1	2	3	4	5	6	7

# Table B. PTDF and ZS Sensitivity Factors Associated with Congested TransmissionLine 4-12 in Modified IEEE-30 Bus System

PTDF slack 5	0.020007234	0.014533128	0.03573458	0.039342731	0	-0.014511141	-0.008501477
Bus	<u>8</u>	9	10	<u>11</u>	12	<u>13</u>	14
PTDF slack 5	-0.016948328	-0.174805106	-0.261482864	-0.174805106	-0.593499921	-0.593499921	-0.545632433
Bus	15	16	17	18	19	20	21
PTDF slack 5	-0.50827746	-0.454758	-0.320484844	-0.421956635	-0.3709148	-0.34405061	-0.2684017
Bus	22	23	24	25	26	27	28
PTDF slack 5	-0.2705818	-0.41827819	-0.297982136	-0.2017	-0.2017	-0.14066	-0.02855
Bus	29	30		1	1		1
PTDF slack 5	-0.14066	-0.14066					
	·	·	·				
Bus	<u>1</u>	<u>2</u>	3	4	<u>5</u>	6	7
PTDF slack 8	0.036955563	0.031481456	0.05268291	0.056291059	0.0169483	0.002437187	0.008446852
Bus	<u>8</u>	9	10	<u>11</u>	12	<u>13</u>	14
PTDF slack 8	0	-0.157856778	-0.244534536	-0.157856778	-0.576551593	-0.576551593	-0.528684104
Bus	15	16	17	18	19	20	21
PTDF slack 8	-0.491329132	-0.4378097	-0.303536515	-0.405008307	-0.3539664	-0.327102281	-0.2514534
Bus	22	23	24	25	26	27	28
PTDF slack 8	-0.2536334	-0.401329862	-0.281033807	-0.18475	-0.18475	-0.12372	-0.01161
Bus	29	30				·	
PTDF slack 8	-0.12372	-0.12372					
	·	·	·				
Bus	1	2	3	4	<u>5</u>	6	7
PTDF slack 11	0.194812341	0.189338234	0.21053969	0.214147837	0.1748051	0.160293965	0.16630363
Bus	<u>8</u>	9	10	<u>11</u>	12	<u>13</u>	14
PTDF slack 11	0.157856778	0	-0.086677758	0	-0.418694815	-0.418694815	-0.370827326
Bus	15	16	17	18	19	20	21
PTDF slack 11	-0.333472354	-0.2799529	-0.145679737	-0.247151529	-0.1961096	-0.169245503	-0.0935966
Bus	22	23	24	25	26	27	28

PTDF slack 11	-0.0957767	-0.243473083	-0.123177029	-0.0269	-0.0269	0.034141	0.146251
	20	20					
Bus	29						
PTDF slack 11	0.034141	0.034141					
Bus	1	2	3	4	<u>5</u>	6	7
PTDF slack 13	0.613507156	0.608033049	0.6292345	0.632842652	0.5934999	0.57898878	0.584998444
Bus	<u>8</u>	9	10	<u>11</u>	12	<u>13</u>	14
PTDF slack 13	0.576551593	0.418694815	0.332017057	0.418694815	0	0	0.047867488
Bus	15	16	17	18	19	20	21
PTDF slack 13	0.085222461	0.1387419	0.273015077	0.171543286	0.2225852	0.249449312	0.3250982
Bus	22	23	24	25	26	27	28
PTDF slack 13	0.3229181	0.175221731	0.295517786	0.391798	0.391798	0.452835	0.564946
Bus	29	30					
PTDF slack 13	0.452835	0.452835					
	·		·				
Bus	<u>1</u>	<u>2</u>	3	4	<u>5</u>	6	7
ZS slack 1	0	-0.518185671	-0.5471995	-0.55408986	-0.5191326	-0.502721889	-0.513800764
Bus	<u>8</u>	9	10	<u>11</u>	12	<u>13</u>	14
ZS slack 1	-0.500350201	-0.338042623	-0.256367645	-0.328463384	0.076725495	0.075841548	0.031153106
Bus	15	16	17	18	19	20	21
ZS slack 1	-0.009589601	-0.0618713	-0.197662755	-0.097427353	-0.1494352	-0.175784547	-0.2517981
Bus	22	23	24	25	26	27	28
ZS slack 1	-0.2499219	-0.101699336	-0.225869378	-0.32787	-0.33282	-0.39007	-0.49044
Bus	29	30					
ZS slack 1	-0.39579	-0.39865					

## C. PTDF and ZS Sensitivity Factors in Modified IEEE-57 Bus System

Bus	1	2	<u>3</u>	4	5	<u>6</u>	7
PTDF slack 1	0	0.003417	0.013791	0.032453	0.064671	0.080317	0.146625
Bus	<u>8</u>	<u>9</u>	10	11	<u>12</u>	13	14
PTDF slack 1	0.066028	0.004692	-0.00822	-0.00606	-0.00685	-0.01125	-0.01578
Bus	15	16	17	18	19	20	21
PTDF slack 1	-0.00677	-0.00491	-0.002577	0.018853	-0.020605	-0.045604	-0.092267
Bus	22	23	24	25	26	27	28
PTDF slack 1	-0.09901	-0.11166	-0.32485	-0.27753	-0.36981	-0.60128	-0.68822
Bus	29	30	31	32	33	34	35
PTDF slack 1	-0.74171	-0.26167	-0.22265	-0.16337	-0.16337	-0.09042	-0.0843
Bus	36	37	38	39	40	41	42
PTDF slack 1	-0.08008	-0.07841	-0.07614	-0.07754	-0.07855	-0.02362	-0.03223
Bus	43	44	45	46	47	48	49
PTDF slack 1	-0.01067	-0.061758	-0.031227	-0.033567	-0.05186	-0.05812	-0.04461
Bus	50	51	52	53	54	55	56
PTDF slack 1	-0.03336	-0.01404	-0.57887	-0.49319	-0.29117	-0.09394	-0.0409
Bus	57						
PTDF slack 1	-0.0469						
	<u>~</u>						
Bus	1	<u>2</u>	<u>3</u>	4	5	<u>6</u>	7
PTDF slack 2	-0.00342	0	0.010374	0.029035	0.061254	0.0769	0.143208
Bus	<u>8</u>	<u>9</u>	10	11	<u>12</u>	13	14
PTDF slack 2	0.062611	0.001275	-0.01164	-0.00948	-0.01027	-0.01467	-0.01919
Bus	15	16	17	18	19	20	21

# Table C. PTDF and ZS Sensitivity Factors Associated with Congested TransmissionLine 7-29 in Modified IEEE-57 Bus System

	1						
PTDF slack 2	-0.01018	-0.00833	-0.005994	0.015435	-0.024022	-0.049021	-0.095684
Bus	22	23	24	25	26	27	28
PTDF slack 2	-0.10242	-0.11508	-0.32827	-0.28095	-0.37323	-0.6047	-0.69163
Bus	29	30	31	32	33	34	35
PTDF slack 2	-0.74513	-0.26509	-0.22607	-0.16679	-0.16679	-0.09384	-0.08772
Bus	36	37	38	39	40	41	42
PTDF slack 2	-0.0835	-0.08183	-0.07956	-0.08096	-0.08197	-0.02703	-0.03565
Bus	43	44	45	46	47	48	49
PTDF slack 2	-0.01409	-0.065175	-0.034644	-0.036985	-0.05527	-0.06154	-0.04802
Bus	50	51	52	53	54	55	56
PTDF slack 2	-0.03678	-0.01746	-0.58229	-0.49661	-0.29459	-0.09736	-0.04432
Bus	57						
PTDF slack 2	-0.05032	-					
Bus	<u>1</u>	2	<u>3</u>	4	5	<u>6</u>	7
Bus PTDF slack 3	<u>1</u> -0.01379	<u>2</u> -0.01037	<u>3</u> 0	<b>4</b> 0.018662	5	<u>6</u> 0.066526	7 0.132834
Bus PTDF slack 3 Bus	<u>1</u> -0.01379 <u>8</u>	2 -0.01037 <u>9</u>	3 0 10	4 0.018662 11	5 0.05088 <u>12</u>	<u>6</u> 0.066526 13	7 0.132834 14
Bus PTDF slack 3 Bus PTDF slack 3	1 -0.01379 <u>8</u> 0.052237	2 -0.01037 2 -0.0091	3 0 10 -0.02202	4 0.018662 11 -0.01985	5 0.05088 <u>12</u> -0.02064	<u>6</u> 0.066526 13 -0.02504	7 0.132834 14 -0.02957
Bus PTDF slack 3 Bus PTDF slack 3 Bus	1 -0.01379 <u>8</u> 0.052237 15	2 -0.01037 2 -0.0091 16	3 0 10 -0.02202 17	4 0.018662 11 -0.01985 18	5 0.05088 <u>12</u> -0.02064 19	6 0.066526 13 -0.02504 20	7 0.132834 14 -0.02957 21
Bus PTDF slack 3 Bus PTDF slack 3 Bus PTDF slack 3	1 -0.01379 <u>8</u> 0.052237 15 -0.02056	2 -0.01037 2 -0.0091 16 -0.0187	3 0 10 -0.02202 17 -0.016368	4 0.018662 11 -0.01985 18 0.005062	5 0.05088 <u>12</u> -0.02064 19 -0.034395	6 0.066526 13 -0.02504 20 -0.059394	7 0.132834 14 -0.02957 21 -0.106057
Bus PTDF slack 3 Bus PTDF slack 3 Bus PTDF slack 3 Bus	1 -0.01379 <u>8</u> 0.052237 15 -0.02056 22	2 -0.01037 2 -0.0091 16 -0.0187 23	3 0 10 -0.02202 17 -0.016368 24	4 0.018662 11 -0.01985 18 0.005062 25	5 0.05088 12 -0.02064 19 -0.034395 26	6         0.066526         13         -0.02504         20         -0.059394         27	7 0.132834 14 -0.02957 21 -0.106057 28
Bus PTDF slack 3 Bus PTDF slack 3 Bus PTDF slack 3 Bus PTDF slack 3	1           -0.01379           8           0.052237           15           -0.02056           22           -0.1128	2 -0.01037 2 -0.0091 16 -0.0187 23 -0.12545	3 0 10 -0.02202 17 -0.016368 24 -0.33865	4 0.018662 11 -0.01985 18 0.005062 25 -0.29132	5 0.05088 12 -0.02064 19 -0.034395 26 -0.3836	6           0.066526           13           -0.02504           20           -0.059394           27           -0.61507	7 0.132834 14 -0.02957 21 -0.106057 28 -0.70201
Bus PTDF slack 3 Bus PTDF slack 3 Bus PTDF slack 3 Bus PTDF slack 3 Bus	1           -0.01379           8           0.052237           15           -0.02056           22           -0.1128           29	2 -0.01037 2 -0.0091 16 -0.0187 23 -0.12545 30	3 0 10 -0.02202 17 -0.016368 24 -0.33865 31	4 0.018662 11 -0.01985 18 0.005062 25 -0.29132 32	5 0.05088 12 -0.02064 19 -0.034395 26 -0.3836 33	6         0.066526         13         -0.02504         20         -0.059394         27         -0.61507         34	7 0.132834 14 -0.02957 21 -0.106057 28 -0.70201 35
Bus PTDF slack 3 Bus PTDF slack 3 Bus PTDF slack 3 Bus PTDF slack 3 Bus PTDF slack 3	1           -0.01379           8           0.052237           15           -0.02056           22           -0.1128           29           -0.7555	2 -0.01037 2 -0.0091 16 -0.0187 23 -0.12545 30 -0.27546	3         0         10         -0.02202         17         -0.016368         24         -0.33865         31         -0.23644	4 0.018662 11 -0.01985 18 0.005062 25 -0.29132 32 -0.17717	5 0.05088 12 -0.02064 19 -0.034395 26 -0.3836 33 -0.17717	6         0.066526         13         -0.02504         20         -0.059394         27         -0.61507         34         -0.10421	7 0.132834 14 -0.02957 21 -0.106057 28 -0.70201 35 -0.09809
Bus PTDF slack 3 Bus PTDF slack 3 PTDF slack 3 Bus PTDF slack 3 Bus PTDF slack 3 Bus Bus	1           -0.01379           8           0.052237           15           -0.02056           22           -0.1128           29           -0.7555           36	2 -0.01037 9 -0.0091 16 -0.0187 23 -0.12545 30 -0.27546 37	3         0         10         -0.02202         17         -0.016368         24         -0.33865         31         -0.23644         38	4 0.018662 11 -0.01985 18 0.005062 25 -0.29132 32 -0.17717 39	5 0.05088 12 -0.02064 19 -0.034395 26 -0.3836 33 -0.17717 40	6           0.066526           13           -0.02504           20           -0.059394           27           -0.61507           34           -0.10421           41	7 0.132834 14 -0.02957 21 -0.106057 28 -0.70201 35 -0.09809 42
Bus PTDF slack 3 Bus PTDF slack 3 Bus PTDF slack 3 Bus PTDF slack 3 Bus PTDF slack 3 Bus PTDF slack 3	1           -0.01379           8           0.052237           15           -0.02056           22           -0.1128           29           -0.7555           36           -0.09387	2 -0.01037 2 -0.0091 16 -0.0187 23 -0.12545 30 -0.27546 37 -0.0922	3         0         10         -0.02202         17         -0.016368         24         -0.33865         31         -0.23644         38         -0.08993	4 0.018662 11 -0.01985 18 0.005062 25 -0.29132 32 -0.17717 39 -0.09133	5 0.05088 12 -0.02064 19 -0.034395 26 -0.3836 33 -0.17717 40 -0.09234	6           0.066526           13           -0.02504           20           -0.059394           -0.059394           27           -0.61507           34           -0.10421           41	7 0.132834 14 -0.02957 21 -0.106057 28 -0.70201 35 -0.09809 42 -0.04602
Bus PTDF slack 3 Bus PTDF slack 3 Bus PTDF slack 3 Bus PTDF slack 3 Bus PTDF slack 3 Bus PTDF slack 3 Bus	1           -0.01379           8           0.052237           15           -0.02056           22           -0.1128           29           -0.7555           36           -0.09387           43	2 -0.01037 2 -0.0091 16 -0.0187 23 -0.12545 30 -0.27546 37 -0.0922 44	3         0         10         -0.02202         17         -0.016368         24         -0.33865         31         -0.23644         38         -0.08993         45	4 0.018662 11 -0.01985 18 0.005062 25 -0.29132 32 -0.17717 39 -0.09133 46	5 0.05088 12 -0.02064 19 -0.034395 26 -0.3836 33 -0.17717 40 -0.09234 47	6         0.066526         13         -0.02504         20         -0.059394         27         -0.61507         34         -0.10421         41         -0.03741         48	7 0.132834 14 -0.02957 21 -0.106057 28 -0.70201 35 -0.09809 42 -0.04602 49

Bus	50	51	52	53	54	55	56
PTDF slack 3	-0.04715	-0.02783	-0.59266	-0.50698	-0.30496	-0.10773	-0.05469
Bus	57			1	1	1	1
PTDF slack 3	-0.06069						
Bus	1	2	<u>3</u>	4	5	<u>6</u>	7
PTDF slack 6	-0.08032	-0.0769	-0.06653	-0.04786	-0.01565	0	0.066308
Bus	<u>8</u>	<u>9</u>	10	11	<u>12</u>	13	14
PTDF slack 6	-0.01429	-0.07562	-0.08854	-0.08638	-0.08717	-0.09157	-0.09609
Bus	15	16	17	18	19	20	21
PTDF slack 6	-0.08708	-0.08523	-0.082894	-0.06146	-0.100921	-0.12592	-0.172583
Bus	22	23	24	25	26	27	28
PTDF slack 6	-0.17932	-0.19198	-0.40517	-0.35785	-0.45013	-0.6816	-0.76853
Bus	29	30	31	32	33	34	35
PTDF slack 6	-0.82203	-0.34199	-0.30297	-0.24369	-0.24369	-0.17074	-0.16462
Bus	36	37	38	39	40	41	42
PTDF slack 6	-0.1604	-0.15873	-0.15645	-0.15786	-0.15887	-0.10393	-0.11255
Bus	43	44	45	46	47	48	49
PTDF slack 6	-0.09099	-0.142074	-0.111544	-0.113884	-0.13217	-0.13844	-0.12492
Bus	50	51	52	53	54	55	56
PTDF slack 6	-0.11368	-0.09436	-0.65919	-0.57351	-0.37149	-0.17426	-0.12122
Bus	57			1	1	1	
PTDF slack 6	-0.12722						
Bus	1	<u>2</u>	<u>3</u>	4	5	<u>6</u>	7
PTDF slack 8	-0.06603	-0.06261	-0.05224	-0.03358	-0.00136	0.014289	0.080597
Bus	<u>8</u>	<u>9</u>	10	11	<u>12</u>	13	14
PTDF slack 8	0	-0.06134	-0.07425	-0.07209	-0.07288	-0.07728	-0.0818

Bus	15	16	17	18	19	20	21
PTDF slack 8	-0.07279	-0.07094	-0.068605	-0.04718	-0.086632	-0.111632	-0.158294
Bus	22	23	24	25	26	27	28
PTDF slack 8	-0.16503	-0.17769	-0.39088	-0.34356	-0.43584	-0.66731	-0.75424
Bus	29	30	31	32	33	34	35
PTDF slack 8	-0.80774	-0.3277	-0.28868	-0.2294	-0.2294	-0.15645	-0.15033
Bus	36	37	38	39	40	41	42
PTDF slack 8	-0.14611	-0.14444	-0.14217	-0.14357	-0.14458	-0.08964	-0.09826
Bus	43	44	45	46	47	48	49
PTDF slack 8	-0.0767	-0.127786	-0.097255	-0.099595	-0.11788	-0.12415	-0.11063
Bus	50	51	52	53	54	55	56
PTDF slack 8	-0.09939	-0.08007	-0.6449	-0.55922	-0.3572	-0.15997	-0.10693
Bus	57		1	1	1		1
PTDF slack 8	-0.11293						
Bus	<u>1</u>	<u>2</u>	<u>3</u>	4	5	<u>6</u>	7
PTDF slack 9	-0.00469	-0.00127	0.009099	0.027761	0.059979	0.075625	0.141933
Bus	<u>8</u>	<u>9</u>	10	11	<u>12</u>	13	14
PTDF slack 9	0.061336	0	-0.01292	-0.01076	-0.01154	-0.01594	-0.02047
Bus	15	16	17	18	19	20	21
PTDF slack 9	-0.01146	-0.0096	-0.007269	0.014161	-0.025297	-0.050296	-0.096959
Bus	22	23	24	25	26	27	28
PTDF slack 9	-0.1037	-0.11636	-0.32955	-0.28222	-0.3745	-0.60597	-0.69291
Bus	29	30	31	32	33	34	35
PTDF slack 9	-0.7464	-0.26636	-0.22734	-0.16807	-0.16807	-0.09512	-0.08899
Bus	36	37	38	39	40	41	42
PTDF slack 9	-0.08478	-0.08311	-0.08083	-0.08223	-0.08324	-0.02831	-0.03693
Bus	43	44	45	46	47	48	49

PTDF slack 9	-0.01536	-0.06645	-0.035919	-0.038259	-0.05655	-0.06282	-0.0493
Bus	50	51	52	53	54	55	56
PTDF slack 9	-0.03806	-0.01873	-0.58357	-0.49788	-0.29586	-0.09863	-0.04559
Bus	57		1	1	1	1	1
PTDF slack 9	-0.05159						
Bus	1	2	<u>3</u>	4	5	<u>6</u>	7
PTDF slack 12	0.006849	0.010266	0.020639	0.039301	0.07152	0.087165	0.153474
Bus	<u>8</u>	<u>9</u>	10	11	<u>12</u>	13	14
PTDF slack 12	0.072877	0.011541	-0.00138	0.000785	0	-0.0044	-0.00893
Bus	15	16	17	18	19	20	21
PTDF slack 12	8.37E-05	0.001938	0.004271	0.025701	-0.013756	-0.038755	-0.085418
Bus	22	23	24	25	26	27	28
PTDF slack 12	-0.09216	-0.10482	-0.31801	-0.27068	-0.36296	-0.59443	-0.68137
Bus	29	30	31	32	33	34	35
PTDF slack 12	-0.73486	-0.25482	-0.2158	-0.15653	-0.15653	-0.08357	-0.07745
Bus	36	37	38	39	40	41	42
PTDF slack 12	-0.07323	-0.07156	-0.06929	-0.07069	-0.0717	-0.01677	-0.02538
Bus	43	44	45	46	47	48	49
PTDF slack 12	-0.00382	-0.054909	-0.024378	-0.026719	-0.04501	-0.05127	-0.03776
Bus	50	51	52	53	54	55	56
PTDF slack 12	-0.02651	-0.00719	-0.57203	-0.48634	-0.28432	-0.08709	-0.03405
Bus	57		1	1	1	1	1
PTDF slack 12	-0.04005						
Bus	1	2	<u>3</u>	4	5	<u>6</u>	7
ZS slack 1	0	-0.43762	-0.45725	-0.47381	-0.49952	-0.50805	-0.54726
Bus	<u>8</u>	<u>9</u>	10	11	<u>12</u>	13	14

ZS slack 1	-0.47657	-0.44238	-0.43445	-0.43838	-0.42475	-0.43265	-0.43207
Bus	15	16	17	18	19	20	21
ZS slack 1	-4.37E-01	-0.42989	-0.431562	-0.47493	-0.442649	-0.415818	-0.372272
Bus	22	23	24	25	26	27	28
ZS slack 1	-0.36433	-0.35139	-0.12318	-0.1839	-0.08028	0.194211	0.293751
Bus	29	30	31	32	33	34	35
ZS slack 1	0.353904	-0.20506	-0.25393	-0.31623	-0.31685	-0.38876	-0.3932
Bus	36	37	38	39	40	41	42
ZS slack 1	-0.39421	-0.39329	-0.3878	-0.39478	-0.39631	-0.43543	-0.44152
Bus	43	44	45	46	47	48	49
ZS slack 1	-0.43825	-0.399062	-0.418448	-0.423371	-0.40965	-0.404	-0.41483
Bus	50	51	52	53	54	55	56
ZS slack 1	-0.42916	-0.43388	0.175723	0.078001	-0.15309	-0.36233	-0.43404
Bus	57		1				
ZS slack 1	-0.42995						
## **D. PTDF and ZS Sensitivity Factors in Modified IEEE-118 Bus System**

Bus	1	2	3	4	5	6	7
PTDF slack 12	-0.000506564	-0.00019322	-0.00063956	-0.00130797	-0.00141	-0.00071019	-0.00044063
Bus	8	9	10	11	<u>12</u>	13	14
PTDF slack 12	-0.00265335	-0.0026533	-0.00265335	-0.00042824	0	-3.84E-05	0.00033664
Bus	15	16	17	18	<u>19</u>	20	21
PTDF slack 12	0.001265137	0.00310089	0.00979718	0.007227716	0.004719312	0.044668464	0.073657207
Bus	22	23	<u>24</u>	25	26	<u>27</u>	28
PTDF slack 12	0.106777444	0.161067317	-0.74920996	0.093181956	0.063820318	0.085372375	0.076739505
Bus	29	30	31	32	33	34	35
PTDF slack 12	0.067218105	-0.00503605	0.06387602	0.08801095	-0.0326731	-0.0705309	-0.0709073
Bus	36	37	38	39	<u>40</u>	41	42
PTDF slack 12	-0.0708035	-0.0714129	-0.0723287	-0.0900837	-0.1007402	-0.1092636	-0.132891
Bus	43	44	45	46	47	48	<u>49</u>
PTDF slack 12	-0.1042598	-0.1534989	-0.1715773	-0.1856969	-0.1968885	-0.18872184	-0.18953008
Bus	50	51	52	53	54	55	56
PTDF slack 12	-0.1902348	-0.1910837	-0.19129264	-0.19187355	-0.192307	-0.1925082	-0.1923959
Bus	57	58	<u>59</u>	60	<u>61</u>	62	63
PTDF slack 12	-0.1914906	-0.1916436	-0.1947263	-0.19559232	-0.19570744	-0.19544898	-0.19573972
Bus	64	65	66	67	68	69	<u>70</u>
PTDF slack 12	-0.19628671	-0.19777535	-0.19442001	-0.194898	-0.22047231	-0.2540757	-0.3739976
Bus	71	72	73	74	75	76	77
PTDF slack 12	-0.40636704	-0.57049397	-0.406367	-0.321739	-0.305702	-0.28494842	-0.2549823
Bus	78	79	80	81	82	83	84
PTDF slack 12	-0.253825	-0.2515479	-0.24497793	-0.2295062	-0.25051721	-0.25016723	-0.24962509
Bus	<u>85</u>	86	87	88	89	90	91
PTDF slack 12	-0.2493618	-0.24936182	-0.24936182	-0.24887509	-0.2485353	-0.24845956	-0.24836232
Bus	92	93	94	95	96	97	98
PTDF slack 12	-0.24821437	-0.24806378	-0.2479338	-0.24807325	-0.248249	-0.2466575	-0.2458634
Bus	<u>99</u>	100	101	102	103	104	105

## Table D. PTDF and ZS Sensitivity Factors Associated with Congested TransmissionLine 23-24 in Modified IEEE-118 Bus System

PTDF slack 12	-0.2466652	-0.24733107	-0.2477101	-0.2480465	-0.24733107	-0.24733107	-0.24733
Bus	106	107	108	109	<u>110</u>	111	112
PTDF slack 12	-0.24733	-0.24733	-0.24733	-0.24733	-0.24733	-0.24733	-0.24733
Bus	113	114	115	116	117	118	
PTDF slack 12	0.019897	0.086903	0.086714	-0.22047	0	-0.29596	
	1	1	1	1	1		
Bus	1	2	3	4	5	6	7
PTDF slack 19	-0.005225877	-0.00491253	-0.00535887	-0.00602729	-0.0061293	-0.0054295	-0.00515994
Bus	8	9	10	11	<u>12</u>	13	14
PTDF slack 19	-0.00737266	-0.0073727	-0.00737266	-0.00514755	-0.00471931	-4.76E-03	-0.00438267
Bus	15	16	17	18	<u>19</u>	20	21
PTDF slack 19	-0.00345418	-0.0016184	0.00507786	0.002508403	0	0.039949152	0.068937895
Bus	22	23	<u>24</u>	25	26	<u>27</u>	28
PTDF slack 19	0.102058132	0.156348005	-0.75392927	0.088462644	0.059101006	0.080653063	0.072020193
Bus	29	30	31	32	33	34	35
PTDF slack 19	0.062498793	-0.00975537	0.05915671	0.08329164	-0.0373924	-0.0752502	-0.0756266
Bus	36	37	38	39	<u>40</u>	41	42
PTDF slack 19	-0.0755229	-0.0761323	-0.077048	-0.094803	-0.1054595	-0.1139829	-0.1376103
Bus	43	44	45	46	47	48	<u>49</u>
PTDF slack 19	-0.1089791	-0.1582182	-0.1762966	-0.1904162	-0.2016078	-0.19344115	-0.1942494
Bus	50	51	52	53	54	55	56
PTDF slack 19	-0.1949541	-0.195803	-0.19601195	-0.19659287	-0.1970263	-0.1972275	-0.1971152
Bus	57	58	<u>59</u>	60	<u>61</u>	62	63
PTDF slack 19	-0.1962099	-0.196363	-0.1994456	-0.20031163	-0.20042676	-0.20016829	-0.20045903
Bus	64	65	66	67	68	69	<u>70</u>
PTDF slack 19	-0.20100602	-0.20249466	-0.19913932	-0.19961731	-0.22519162	-0.258795	-0.3787169
Bus	71	72	73	74	75	76	77
PTDF slack 19	-0.41108635	-0.57521328	-0.4110864	-0.3264583	-0.3104213	-0.28966773	-0.2597016
Bus	78	79	80	81	82	83	84
PTDF slack 19	-0.2585444	-0.2562673	-0.24969725	-0.2342255	-0.25523652	-0.25488654	-0.2543444
Bus	<u>85</u>	86	87	88	89	90	91
PTDF slack 19	-0.2540811	-0.25408113	-0.25408113	-0.2535944	-0.2532547	-0.25317887	-0.25308163
Bus	92	93	94	95	96	97	98
PTDF slack 19	-0.25293368	-0.2527831	-0.25265311	-0.25279256	-0.2529683	-0.2513768	-0.2505827
Bus	<u>99</u>	100	101	102	103	104	105

PTDF slack 19	-0.2513845	-0.25205038	-0.25242941	-0.2527658	-0.25205038	-0.25205038	-0.25205
Bus	106	107	108	109	<u>110</u>	111	112
PTDF slack 19	-0.25205	-0.25205	-0.25205	-0.25205	-0.25205	-0.25205	-0.25205
Bus	113	114	115	116	117	118	
PTDF slack 19	0.015178	0.082183	0.081995	-0.22519	-0.00472	-0.30068	
Bus	1	2	3	4	5	6	7
PTDF slack 24	0.748703396	0.749016745	0.748570404	0.747901986	0.74779995	0.74849977	0.748769332
Bus	8	9	10	11	<u>12</u>	13	14
PTDF slack 24	0.746556614	0.74655661	0.746556614	0.748781722	0.749209961	7.49E-01	0.749546601
Bus	15	16	17	18	<u>19</u>	20	21
PTDF slack 24	0.750475097	0.75231085	0.75900714	0.756437677	0.753929273	0.793878425	0.822867168
Bus	22	23	<u>24</u>	25	26	<u>27</u>	28
PTDF slack 24	0.855987405	0.910277278	0	0.842391917	0.813030279	0.834582336	0.825949466
Bus	29	30	31	32	33	34	35
PTDF slack 24	0.816428066	0.744173908	0.81308598	0.83722091	0.71653684	0.67867907	0.67830265
Bus	36	37	38	39	<u>40</u>	41	42
PTDF slack 24	0.67840642	0.67779702	0.6768813	0.6591262	0.6484698	0.6399464	0.6163189
Bus	43	44	45	46	47	48	<u>49</u>
PTDF slack 24	0.6449501	0.5957111	0.5776327	0.563513	0.5523215	0.56048812	0.55967988
Bus	50	51	52	53	54	55	56
PTDF slack 24	0.5589751	0.5581262	0.55791732	0.55733641	0.5569029	0.5567018	0.556814
Bus	57	58	<u>59</u>	60	<u>61</u>	62	63
PTDF slack 24	0.5577193	0.5575663	0.5544837	0.55361764	0.55350252	0.55376099	0.55347024
Bus	64	65	66	67	68	69	<u>70</u>
PTDF slack 24	0.55292326	0.55143461	0.55478995	0.55431196	0.52873765	0.4951343	0.3752124
Bus	71	72	73	74	75	76	77
PTDF slack 24	0.34284292	0.17871599	0.3428429	0.4274709	0.443508	0.46426154	0.4942277
Bus	78	79	80	81	82	83	84
PTDF slack 24	0.4953849	0.497662	0.50423203	0.5197037	0.49869275	0.49904273	0.49958487
Bus	<u>85</u>	86	87	88	89	90	91
PTDF slack 24	0.4998481	0.49984814	0.49984814	0.50033487	0.5006746	0.5007504	0.50084764
Bus	92	93	94	95	96	97	98
PTDF slack 24	0.50099559	0.50114618	0.50127616	0.50113671	0.500961	0.5025524	0.5033465
Bus	<u>99</u>	100	101	102	103	104	105

PTDF slack 24	0.50254478	0.501878894	0.501499864	0.5011635	0.501878894	0.50187889	0.501879
Bus	106	107	108	109	<u>110</u>	111	112
PTDF slack 24	0.501879	0.501879	0.501879	0.501879	0.501879	0.501879	0.501879
Bus	113	114	115	116	117	118	
PTDF slack 24	0.769107	0.836113	0.835924	0.528738	0.74921	0.453247	
		-					-
Bus	1	2	3	4	5	6	7
PTDF slack 27	-0.08587894	-0.08556559	-0.08601193	-0.08668035	-0.0867824	-0.08608257	-0.085813
Bus	8	9	10	11	<u>12</u>	13	14
PTDF slack 27	-0.08802572	-0.0880257	-0.08802572	-0.08580061	-0.08537238	-8.54E-02	-0.08503574
Bus	15	16	17	18	<u>19</u>	20	21
PTDF slack 27	-0.08410724	-0.0822715	-0.0755752	-0.07814466	-0.08065306	-0.04070391	-0.01171517
Bus	22	23	<u>24</u>	25	26	<u>27</u>	28
PTDF slack 27	0.021405069	0.075694942	-0.83458234	0.007809581	-0.02155206	0	-0.00863287
Bus	29	30	31	32	33	34	35
PTDF slack 27	-0.01815427	-0.09040843	-0.0214964	0.00263858	-0.1180455	-0.1559033	-0.1562797
Bus	36	37	38	39	<u>40</u>	41	42
PTDF slack 27	-0.1561759	-0.1567853	-0.1577011	-0.1754561	-0.1861125	-0.1946359	-0.2182634
Bus	43	44	45	46	47	48	<u>49</u>
PTDF slack 27	-0.1896322	-0.2388712	-0.2569496	-0.2710693	-0.2822609	-0.27409422	-0.27490246
Bus	50	51	52	53	54	55	56
PTDF slack 27	-0.2756072	-0.2764561	-0.27666502	-0.27724593	-0.2776794	-0.2778805	-0.2777683
Bus	57	58	<u>59</u>	60	<u>61</u>	62	63
PTDF slack 27	-0.276863	-0.277016	-0.2800986	-0.28096469	-0.28107982	-0.28082135	-0.28111209
Bus	64	65	66	67	68	69	<u>70</u>
PTDF slack 27	-0.28165908	-0.28314772	-0.27979238	-0.28027037	-0.30584469	-0.3394481	-0.4593699
Bus	71	72	73	74	75	76	77
PTDF slack 27	-0.49173941	-0.65586634	-0.4917394	-0.4071114	-0.3910744	-0.37032079	-0.3403546
Bus	78	79	80	81	82	83	84
PTDF slack 27	-0.3391974	-0.3369203	-0.33035031	-0.3148786	-0.33588958	-0.33553961	-0.33499746
Bus	<u>85</u>	86	87	88	89	90	91
PTDF slack 27	-0.3347342	-0.33473419	-0.33473419	-0.33424747	-0.3339077	-0.33383193	-0.3337347
Bus	92	93	94	95	96	97	98
PTDF slack 27	-0.33358674	-0.33343616	-0.33330617	-0.33344563	-0.3336214	-0.3320299	-0.3312358
Bus	<u>99</u>	100	101	102	103	104	105

PTDF clock 27	0.0000	-0 33270244	-0 33308242	-0 333/190	-0 33270244	-0 33270244	-0 3327
D	-0.3320376	107	100	100	110	111	110
Bus	106	107	108	109	<u>110</u>	111	112
PTDF slack 27	-0.3327	-0.3327	-0.3327	-0.3327	-0.3327	-0.3327	-0.3327
Bus	113	114	115	116	117	118	
PTDF slack 27	-0.06548	0.00153	0.001342	-0.30584	-0.08537	-0.38134	
Bus	1	2	3	4	5	6	7
PTDF slack 40	0.100233594	0.100546943	0.100100601	0.099432184	0.09933015	0.100029968	0.100299529
Bus	8	9	10	11	<u>12</u>	13	14
PTDF slack 40	0.098086812	0.09808681	0.098086812	0.10031192	0.100740159	1.01E-01	0.101076798
Bus	15	16	17	18	<u>19</u>	20	21
PTDF slack 40	0.102005295	0.10384105	0.11053733	0.107967874	0.105459471	0.145408623	0.174397366
Bus	22	23	<u>24</u>	25	26	<u>27</u>	28
PTDF slack 40	0.207517603	0.261807476	-0.6484698	0.193922114	0.164560477	0.186112534	0.177479664
Bus	29	30	31	32	33	34	35
PTDF slack 40	0.167958264	0.095704106	0.16461618	0.18875111	0.06806703	0.03020927	0.02983284
Bus	36	37	38	39	<u>40</u>	41	42
PTDF slack 40	0.02993661	0.02932722	0.0284115	0.0106564	0	-0.0085234	-0.0321509
Bus	43	44	45	46	47	48	<u>49</u>
PTDF slack 40	-0.0035197	-0.0527587	-0.0708371	-0.0849568	-0.0961483	-0.08798168	-0.08878993
Bus	50	51	52	53	54	55	56
PTDF slack 40	-0.0894947	-0.0903436	-0.09055248	-0.0911334	-0.0915669	-0.091768	-0.0916558
Bus	57	58	<u>59</u>	60	<u>61</u>	62	63
PTDF slack 40	-0.0907505	-0.0909035	-0.0939861	-0.09485216	-0.09496729	-0.09470882	-0.09499956
Bus	64	65	66	67	68	69	<u>70</u>
PTDF slack 40	-0.09554655	-0.09703519	-0.09367985	-0.09415784	-0.11973215	-0.1533355	-0.2732574
Bus	71	72	73	74	75	76	77
PTDF slack 40	-0.30562688	-0.46975381	-0.3056269	-0.2209989	-0.2049618	-0.18420826	-0.1542421
Bus	78	79	80	81	82	83	84
PTDF slack 40	-0.1530849	-0.1508078	-0.14423777	-0.1287661	-0.14977705	-0.14942707	-0.14888493
Bus	<u>85</u>	86	87	88	89	90	91
PTDF slack 40	-0.1486217	-0.14862166	-0.14862166	-0.14813493	-0.1477952	-0.1477194	-0.14762216
Bus	92	93	94	95	96	97	98
PTDF slack 40	-0.14747421	-0.14732363	-0.14719364	-0.14733309	-0.1475089	-0.1459174	-0.1451233
Bus	<u>99</u>	100	101	102	103	104	105

PTDF slack 40	-0.145925	-0.14659091	-0.14696994	-0.1473063	-0.14659091	-0.14659091	-0.14659
Bus	106	107	108	109	<u>110</u>	111	112
PTDF slack 40	-0.14659	-0.14659	-0.14659	-0.14659	-0.14659	-0.14659	-0.14659
Bus	113	114	115	116	117	118	
PTDF slack 40	0.120637	0.187643	0.187454	-0.11973	0.10074	-0.19522	
Bus	1	2	3	4	5	6	7
PTDF slack 49	0.18902352	0.189336868	0.188890527	0.18822211	0.18812007	0.188819893	0.189089455
Bus	8	9	10	11	<u>12</u>	13	14
PTDF slack 49	0.186876737	0.18687674	0.186876737	0.189101845	0.189530084	1.89E-01	0.189866724
Bus	15	16	17	18	<u>19</u>	20	21
PTDF slack 49	0.190795221	0.19263097	0.19932726	0.1967578	0.194249396	0.234198548	0.263187291
Bus	22	23	<u>24</u>	25	26	<u>27</u>	28
PTDF slack 49	0.296307528	0.350597401	-0.55967988	0.28271204	0.253350402	0.274902459	0.266269589
Bus	29	30	31	32	33	34	35
PTDF slack 49	0.256748189	0.184494031	0.25340611	0.27754104	0.15685696	0.11899919	0.11862277
Bus	36	37	38	39	<u>40</u>	41	42
PTDF slack 49	0.11872654	0.11811714	0.1172014	0.0994464	0.0887899	0.0802665	0.0566391
Bus	43	44	45	46	47	48	<u>49</u>
PTDF slack 49	0.0852703	0.0360312	0.0179528	0.0038332	-0.0073584	0.00080824	0
Bus	50	51	52	53	54	55	56
PTDF slack 49	-0.0007047	-0.0015536	-0.00176256	-0.00234347	-0.0027769	-0.0029781	-0.0028658
Bus	57	58	<u>59</u>	60	<u>61</u>	62	63
PTDF slack 49	-0.0019605	-0.0021136	-0.0051962	-0.00606223	-0.00617736	-0.00591889	-0.00620963
Bus	64	65	66	67	68	69	<u>70</u>
PTDF slack 49	-0.00675662	-0.00824526	-0.00488993	-0.00536791	-0.03094223	-0.0645456	-0.1844675
Bus	71	72	73	74	75	76	77
PTDF slack 49	-0.21683695	-0.38096389	-0.216837	-0.1322089	-0.1161719	-0.09541834	-0.0654522
Bus	78	79	80	81	82	83	84
PTDF slack 49	-0.064295	-0.0620179	-0.05544785	-0.0399761	-0.06098712	-0.06063715	-0.060095
Bus	<u>85</u>	86	87	88	89	90	91
PTDF slack 49	-0.0598317	-0.05983173	-0.05983173	-0.05934501	-0.0590053	-0.05892948	-0.05883224
Bus	92	93	94	95	96	97	98
PTDF slack 49	-0.05868428	-0.0585337	-0.05840372	-0.05854317	-0.0587189	-0.0571274	-0.0563333
Bus	<u>99</u>	100	101	102	103	104	105

PTDF slack 49	-0.0571351	-0.05780098	-0.05818001	-0.0585164	-0.05780098	-0.05780098	-0.0578
Bus	106	107	108	109	<u>110</u>	111	112
PTDF slack 49	-0.0578	-0.0578	-0.0578	-0.0578	-0.0578	-0.0578	-0.0578
Bus	113	114	115	116	117	118	
PTDF slack 49	0.209427	0.276433	0.276244	-0.03094	0.18953	-0.10643	
Bus	1	2	3	4	5	6	7
PTDF slack 59	0.19421969	0.194533039	0.194086698	0.19341828	0.19331624	0.194016064	0.194285626
Bus	8	9	10	11	<u>12</u>	13	14
PTDF slack 59	0.192072908	0.19207291	0.192072908	0.194298016	0.194726255	1.95E-01	0.195062894
Bus	15	16	17	18	<u>19</u>	20	21
PTDF slack 59	0.195991391	0.19782714	0.20452343	0.20195397	0.199445567	0.239394719	0.268383462
Bus	22	23	<u>24</u>	25	26	<u>27</u>	28
PTDF slack 59	0.301503699	0.355793572	-0.55448371	0.287908211	0.258546573	0.28009863	0.27146576
Bus	29	30	31	32	33	34	35
PTDF slack 59	0.26194436	0.189690202	0.25860228	0.28273721	0.16205313	0.12419536	0.12381894
Bus	36	37	38	39	<u>40</u>	41	42
PTDF slack 59	0.12392271	0.12331331	0.1223975	0.1046425	0.0939861	0.0854627	0.0618352
Bus	43	44	45	46	47	48	<u>49</u>
PTDF slack 59	0.0904664	0.0412274	0.023149	0.0090293	-0.0021623	0.00600441	0.00519617
Bus	50	51	52	53	54	55	56
PTDF slack 59	0.0044914	0.0036425	0.00343361	0.0028527	0.0024192	0.0022181	0.0023303
Bus	57	58	<u>59</u>	60	<u>61</u>	62	63
PTDF slack 59	0.0032356	0.0030826	0	-0.00086606	-0.00098119	-0.00072272	-0.00101346
Bus	64	65	66	67	68	69	<u>70</u>
PTDF slack 59	-0.00156045	-0.00304909	0.00030625	-0.00017174	-0.02574606	-0.0593494	-0.1792713
Bus	71	72	73	74	75	76	77
PTDF slack 59	-0.21164078	-0.37576771	-0.2116408	-0.1270128	-0.1109757	-0.09022216	-0.060256
Bus	78	79	80	81	82	83	84
PTDF slack 59	-0.0590988	-0.0568217	-0.05025168	-0.03478	-0.05579095	-0.05544098	-0.05489883
Bus	<u>85</u>	86	87	88	89	90	91
PTDF slack 59	-0.0546356	-0.05463556	-0.05463556	-0.05414884	-0.0538091	-0.0537333	-0.05363607
Bus	92	93	94	95	96	97	98
PTDF slack 59	-0.05348811	-0.05333753	-0.05320754	-0.053347	-0.0535228	-0.0519313	-0.0511372
Bus	<u>99</u>	100	101	102	103	104	105

PTDF slack 59	-0.0519389	-0.05260481	-0.05298384	-0.0533202	-0.05260481	-0.05260481	-0.0526
Bus	106	107	108	109	<u>110</u>	111	112
PTDF slack 59	-0.0526	-0.0526	-0.0526	-0.0526	-0.0526	-0.0526	-0.0526
Bus	113	114	115	116	117	118	
PTDF slack 59	0.214623	0.281629	0.281441	-0.02575	0.194726	-0.10124	
						<u>.</u>	
Bus	1	2	3	4	5	6	7
PTDF slack 61	0.19520088	0.195514228	0.195067887	0.19439947	0.19429743	0.194997254	0.195266815
Bus	8	9	10	11	<u>12</u>	13	14
PTDF slack 61	0.193054098	0.1930541	0.193054098	0.195279205	0.195707444	1.96E-01	0.196044084
Bus	15	16	17	18	<u>19</u>	20	21
PTDF slack 61	0.196972581	0.19880833	0.20550462	0.20293516	0.200426756	0.240375908	0.269364652
Bus	22	23	<u>24</u>	25	26	<u>27</u>	28
PTDF slack 61	0.302484888	0.356774761	-0.55350252	0.2888894	0.259527762	0.281079819	0.272446949
Bus	29	30	31	32	33	34	35
PTDF slack 61	0.262925549	0.190671391	0.25958347	0.2837184	0.16303432	0.12517655	0.12480013
Bus	36	37	38	39	<u>40</u>	41	42
PTDF slack 61	0.1249039	0.1242945	0.1233787	0.1056237	0.0949673	0.0864439	0.0628164
Bus	43	44	45	46	47	48	<u>49</u>
PTDF slack 61	0.0914476	0.0422086	0.0241302	0.0100105	-0.0011811	0.0069856	0.00617736
Bus	50	51	52	53	54	55	56
PTDF slack 61	0.0054726	0.0046237	0.0044148	0.00383389	0.0034004	0.0031993	0.0033115
Bus	57	58	<u>59</u>	60	<u>61</u>	62	63
PTDF slack 61	0.0042168	0.0040638	0.0009812	0.00011513	0	0.00025847	-3.23E-05
Bus	64	65	66	67	68	69	<u>70</u>
PTDF slack 61	-0.00057926	-0.0020679	0.00128743	0.00080945	-0.02476487	-0.0583683	-0.1782901
Bus	71	72	73	74	75	76	77
PTDF slack 61	-0.21065959	-0.37478653	-0.2106596	-0.1260316	-0.1099946	-0.08924098	-0.0592748
Bus	78	79	80	81	82	83	84
PTDF slack 61	-0.0581176	-0.0558405	-0.04927049	-0.0337988	-0.05480976	-0.05445979	-0.05391764
Bus	<u>85</u>	86	87	88	89	90	91
PTDF slack 61	-0.0536544	-0.05365437	-0.05365437	-0.05316765	-0.0528279	-0.05275212	-0.05265488
Bus	92	93	94	95	96	97	98
PTDF slack 61	-0.05250692	-0.05235634	-0.05222635	-0.05236581	-0.0525416	-0.0509501	-0.050156
Bus	<u>99</u>	100	101	102	103	104	105

PTDF slack 61	-0.0509577	-0.05162362	-0.05200265	-0.052339	-0.05162362	-0.05162362	-0.05162
Bus	106	107	108	109	<u>110</u>	111	112
PTDF slack 61	-0.05162	-0.05162	-0.05162	-0.05162	-0.05162	-0.05162	-0.05162
Bus	113	114	115	116	117	118	
PTDF slack 61	0.215604	0.28261	0.282422	-0.02476	0.195707	-0.10026	-
Bus	1	2	3	4	5	6	7
PTDF slack 70	0.373490996	0.373804345	0.373358003	0.372689586	0.37258755	0.37328737	0.373556931
Bus	8	9	10	11	<u>12</u>	13	14
PTDF slack 70	0.371344214	0.37134421	0.371344214	0.373569322	0.37399756	3.74E-01	0.3743342
Bus	15	16	17	18	<u>19</u>	20	21
PTDF slack 70	0.375262697	0.37709845	0.38379474	0.381225276	0.378716873	0.418666024	0.447654768
Bus	22	23	<u>24</u>	25	26	<u>27</u>	28
PTDF slack 70	0.480775005	0.535064877	-0.3752124	0.467179516	0.437817879	0.459369936	0.450737065
Bus	29	30	31	32	33	34	35
PTDF slack 70	0.441215666	0.368961507	0.43787358	0.46200851	0.34132443	0.30346667	0.30309025
Bus	36	37	38	39	<u>40</u>	41	42
PTDF slack 70	0.30319402	0.30258462	0.3016689	0.2839138	0.2732574	0.264734	0.2411065
Bus	43	44	45	46	47	48	<u>49</u>
PTDF slack 70	0.2697377	0.2204987	0.2024203	0.1883006	0.1771091	0.18527572	0.18446748
Bus	50	51	52	53	54	55	56
PTDF slack 70	0.1837627	0.1829138	0.18270492	0.18212401	0.1816905	0.1814894	0.1816016
Bus	57	58	<u>59</u>	60	<u>61</u>	62	63
PTDF slack 70	0.1825069	0.1823539	0.1792713	0.17840524	0.17829012	0.17854858	1.78E-01
Bus	64	65	66	67	68	69	<u>70</u>
PTDF slack 70	0.17771086	0.17622221	0.17957755	0.17909956	0.15352525	0.1199219	0
Bus	71	72	73	74	75	76	77
PTDF slack 70	-0.03236948	-0.19649641	-0.0323695	0.0522585	0.0682956	0.08904914	0.1190153
Bus	78	79	80	81	82	83	84
PTDF slack 70	0.1201725	0.1224496	0.12901963	0.1444913	0.12348035	0.12383033	0.12437247
Bus	<u>85</u>	86	87	88	89	90	91
PTDF slack 70	0.1246357	0.12463574	0.12463574	0.12512247	0.1254622	0.125538	0.12563524
Bus	92	93	94	95	96	97	98
PTDF slack 70	0.12578319	0.12593378	0.12606376	0.12592431	0.1257486	0.12734	0.1281341
Bus	<u>99</u>	100	101	102	103	104	105

PTDF slack 70	0.12733238	0.126666494	0.126287464	0.1259511	0.126666494	0.12666649	0.126666
Bus	106	107	108	109	<u>110</u>	111	112
PTDF slack 70	0.126666	0.126666	0.126666	0.126666	0.126666	0.126666	0.126666
Bus	113	114	115	116	117	118	
PTDF slack 70	0.393894	0.4609	0.460712	0.153525	0.373998	0.078035	
			<u>.</u>	<u>.</u>	<u>.</u>	<u>.</u>	
Bus	1	2	3	4	5	6	7
PTDF slack 85	0.248855253	0.249168601	0.24872226	0.248053843	0.2479518	0.248651626	0.248921188
Bus	8	9	10	11	<u>12</u>	13	14
PTDF slack 85	0.24670847	0.24670847	0.24670847	0.248933578	0.249361817	2.49E-01	0.249698457
Bus	15	16	17	18	<u>19</u>	20	21
PTDF slack 85	0.250626954	0.25246271	0.25915899	0.256589533	0.254081129	0.294030281	0.323019024
Bus	22	23	<u>24</u>	25	26	<u>27</u>	28
PTDF slack 85	0.356139261	0.410429134	-0.49984814	0.342543773	0.313182135	0.334734192	0.326101322
Bus	29	30	31	32	33	34	35
PTDF slack 85	0.316579922	0.244325764	0.31323784	0.33737277	0.21668869	0.17883092	0.1784545
Bus	36	37	38	39	<u>40</u>	41	42
PTDF slack 85	0.17855827	0.17794887	0.1770331	0.1592781	0.1486217	0.1400983	0.1164708
Bus	43	44	45	46	47	48	<u>49</u>
PTDF slack 85	0.145102	0.095863	0.0777846	0.0636649	0.0524733	0.06063998	0.05983173
Bus	50	51	52	53	54	55	56
PTDF slack 85	0.059127	0.0582781	0.05806917	0.05748826	0.0570548	0.0568537	0.0569659
Bus	57	58	<u>59</u>	60	<u>61</u>	62	63
PTDF slack 85	0.0578712	0.0577182	0.0546356	0.0537695	0.05365437	0.05391284	5.36E-02
Bus	64	65	66	67	68	69	<u>70</u>
PTDF slack 85	0.05307511	0.05158647	0.05494181	0.05446382	0.0288895	-0.0047139	-0.1246357
Bus	71	72	73	74	75	76	77
PTDF slack 85	-0.15700522	-0.32113215	-0.1570052	-0.0723772	-0.0563402	-0.0355866	-0.0056205
Bus	78	79	80	81	82	83	84
PTDF slack 85	-0.0044632	-0.0021861	0.00438388	0.0198556	-0.00115539	-0.00080541	-0.00026327
Bus	<u>85</u>	86	87	88	89	90	91
PTDF slack 85	0	0	0	0.00048673	0.0008265	0.00090226	0.0009995
Bus	92	93	94	95	96	97	98
PTDF slack 85	0.00114745	0.00129803	0.00142802	0.00128857	0.0011128	0.0027043	0.0034984
Bus	<u>99</u>	100	101	102	103	104	105

PTDF slack 85	0.00269664	0.00203075	0.00165172	0.0013153	0.00203075	0.00203075	0.002031
Bus	106	107	108	109	<u>110</u>	111	112
PTDF slack 85	0.002031	0.002031	0.002031	0.002031	0.002031	0.002031	0.002031
Bus	113	114	115	116	117	118	
PTDF slack 85	0.269259	0.336264	0.336076	0.02889	0.249362	-0.0466	
		·	·		·	·	
Bus	1	2	3	4	5	6	7
PTDF slack 99	0.246158614	0.246471962	0.246025621	0.245357204	0.24525516	0.245954988	0.246224549
Bus	8	9	10	11	<u>12</u>	13	14
PTDF slack 99	0.244011832	0.24401183	0.244011832	0.24623694	0.246665178	2.47E-01	0.247001818
Bus	15	16	17	18	<u>19</u>	20	21
PTDF slack 99	0.247930315	0.24976607	0.25646235	0.253892894	0.251384491	0.291333642	0.320322386
Bus	22	23	<u>24</u>	25	26	<u>27</u>	28
PTDF slack 99	0.353442623	0.407732495	-0.50254478	0.339847134	0.310485496	0.332037554	0.323404683
Bus	29	30	31	32	33	34	35
PTDF slack 99	0.313883284	0.241629125	0.3105412	0.33467613	0.21399205	0.17613429	0.17575786
Bus	36	37	38	39	<u>40</u>	41	42
PTDF slack 99	0.17586163	0.17525224	0.1743365	0.1565815	0.145925	0.1374016	0.1137741
Bus	43	44	45	46	47	48	<u>49</u>
PTDF slack 99	0.1424053	0.0931663	0.0750879	0.0609682	0.0497767	0.05794334	0.05713509
Bus	50	51	52	53	54	55	56
PTDF slack 99	0.0564304	0.0555815	0.05537254	0.05479162	0.0543582	0.054157	0.0542693
Bus	57	58	<u>59</u>	60	<u>61</u>	62	63
PTDF slack 99	0.0551746	0.0550215	0.0519389	0.05107286	0.05095773	0.0512162	5.09E-02
Bus	64	65	66	67	68	69	<u>70</u>
PTDF slack 99	0.05037847	0.04888983	0.05224517	0.05176718	0.02619287	-0.0074105	-0.1273324
Bus	71	72	73	74	75	76	77
PTDF slack 99	-0.15970186	-0.32382879	-0.1597019	-0.0750738	-0.0590368	-0.03828324	-0.0083171
Bus	78	79	80	81	82	83	84
PTDF slack 99	-0.0071599	-0.0048828	0.00168725	0.0171589	-0.00385203	-0.00350205	-0.00295991
Bus	<u>85</u>	86	87	88	89	90	91
PTDF slack 99	-0.0026966	-0.00269664	-0.00269664	-0.00220991	-0.0018702	-0.00179438	-0.00169714
Bus	92	93	94	95	96	97	98
PTDF slack 99	-0.00154919	-0.00139861	-0.00126862	-0.00140807	-0.0015838	7.65E-06	0.0008017
Bus	<u>99</u>	100	101	102	103	104	105

PTDF slack 99	0	-0.00066589	-0.00104492	-0.0013813	-0.00066589	-0.00066589	-0.00067
Bus	106	107	108	109	<u>110</u>	111	112
PTDF slack 99	-0.00067	-0.00067	-0.00067	-0.00067	-0.00067	-0.00067	-0.00067
Bus	113	114	115	116	117	118	
PTDF slack 99	0.266562	0.333568	0.333379	0.026193	0.246665	-0.0493	
Bus	1	2	3	4	5	6	7
PTDF slack 110	0.246824502	0.247137851	0.24669151	0.246023092	0.24592105	0.246620876	0.246890438
Bus	8	9	10	11	<u>12</u>	13	14
PTDF slack 110	0.24467772	0.24467772	0.24467772	0.246902828	0.247331067	2.47E-01	0.247667707
Bus	15	16	17	18	<u>19</u>	20	21
PTDF slack 110	0.248596203	0.25043196	0.25712824	0.254558783	0.252050379	0.291999531	0.320988274
Bus	22	23	<u>24</u>	25	26	<u>27</u>	28
PTDF slack 110	0.354108511	0.408398384	-0.50187889	0.340513023	0.311151385	0.332703442	0.324070572
Bus	29	30	31	32	33	34	35
PTDF slack 110	0.314549172	0.242295014	0.31120709	0.33534202	0.21465794	0.17680017	0.17642375
Bus	36	37	38	39	<u>40</u>	41	42
PTDF slack 110	0.17652752	0.17591812	0.1750024	0.1572473	0.1465909	0.1380675	0.11444
Bus	43	44	45	46	47	48	<u>49</u>
PTDF slack 110	0.1430712	0.0938322	0.0757538	0.0616341	0.0504426	0.05860923	0.05780098
Bus	50	51	52	53	54	55	56
PTDF slack 110	0.0570962	0.0562473	0.05603842	0.05545751	0.055024	0.0548229	0.0549351
Bus	57	58	<u>59</u>	60	<u>61</u>	62	63
PTDF slack 110	0.0558404	0.0556874	0.0526048	0.05173875	0.05162362	0.05188209	5.16E-02
Bus	64	65	66	67	68	69	<u>70</u>
PTDF slack 110	0.05104436	0.04955572	0.05291106	0.05243307	0.02685875	-0.0067446	-0.1266665
Bus	71	72	73	74	75	76	77
PTDF slack 110	-0.15903597	-0.3231629	-0.159036	-0.074408	-0.0583709	-0.03761735	-0.0076512
Bus	78	79	80	81	82	83	84
PTDF slack 110	-0.006494	-0.0042169	0.00235313	0.0178248	-0.00318614	-0.00283616	-0.00229402
Bus	<u>85</u>	86	87	88	89	90	91
PTDF slack 110	-0.0020308	-0.00203075	-0.00203075	-0.00154402	-0.0012043	-0.00112849	-0.00103125
Bus	92	93	94	95	96	97	98
PTDF slack 110	-0.0008833	-0.00073272	-0.00060273	-0.00074218	-0.0009179	6.74E-04	0.0014676
Bus	<u>99</u>	100	101	102	103	104	105

PTDF slack 110	0.00066589	0	-0.00037903	-0.0007154	0	0	0
Bus	106	107	108	109	<u>110</u>	111	112
PTDF slack 110	0	0	2.41E-35	1.20E-35	0	0	0
Bus	113	114	115	116	117	118	
PTDF slack 110	0.267228	0.334234	0.334045	0.026859	0.247331	-0.04863	•
	1		1				
Bus	1	2	3	4	5	6	7
ZS slack 12	-0.229468006	-0.22541591	-0.22684564	-0.22095527	-0.2202362	-0.22181306	-0.22166374
Bus	8	9	10	11	<u>12</u>	13	14
ZS slack 12	-0.21700847	-0.2201711	-0.21559244	-0.22221924	0	-2.23E-01	-0.21870422
Bus	15	16	17	18	<u>19</u>	20	21
ZS slack 12	-0.21112969	-0.2216001	-0.2173941	-0.21653133	-0.2136824	-0.25478029	-0.28347339
Bus	22	23	<u>24</u>	25	26	<u>27</u>	28
ZS slack 12	-0.31250845	-0.35177285	0.581722946	-0.28184065	-0.25527284	-0.30094506	-0.29375801
Bus	29	30	31	32	33	34	35
ZS slack 12	-0.28302169	-0.20220587	-0.2781919	-0.3026363	-0.1611607	-0.1042271	-0.1041867
Bus	36	37	38	39	<u>40</u>	41	42
ZS slack 12	-0.104418	-0.1026171	-0.1009993	-0.0744939	-0.057291	-0.0439554	-0.0067029
Bus	43	44	45	46	47	48	<u>49</u>
ZS slack 12	-0.0529269	0.0238101	0.051614	0.0719533	0.0858693	0.07573218	0.07599278
Bus	50	51	52	53	54	55	56
ZS slack 12	0.0797009	0.0847963	0.08628733	0.08853439	0.0882456	0.0888806	0.0884888
Bus	57	58	<u>59</u>	60	<u>61</u>	62	63
ZS slack 12	0.0851215	0.0866623	0.0887703	0.08750072	0.08712248	0.08690864	8.76E-02
Bus	64	65	66	67	68	69	<u>70</u>
ZS slack 12	0.08659565	0.08416061	0.08057457	0.08428337	0.11508629	0.1526866	0.2744479
Bus	71	72	73	74	75	76	77
ZS slack 12	0.30132103	0.44216469	0.3002473	0.2385946	0.2214001	0.21153287	0.1742184
Bus	78	79	80	81	82	83	84
ZS slack 12	0.174214	0.171705	0.15935886	0.1316361	0.18508597	0.18785845	0.18965545
Bus	<u>85</u>	86	87	88	89	90	91
ZS slack 12	0.1883848	0.19054819	0.1858847	0.18584795	0.1779356	0.19007944	0.19063796
Bus	92	93	94	95	96	97	98
ZS slack 12	0.18646348	0.18850214	0.18742126	0.18772556	0.1825975	1.72E-01	0.1713672
Bus	<u>99</u>	100	101	102	103	104	105

ZS slack 12	0.18153122	0.185543625	0.189993047	0.1883795	0.194975869	0.20471157	0.207887
Bus	106	107	108	109	<u>110</u>	111	112
ZS slack 12	0.208056	0.213228	2.09E-01	2.09E-01	0.208291	0.206575	0.20853
Bus	113	114	115	116	117	118	
ZS slack 12	0.208056	0.213228	2.09E-01	2.09E-01	0.208291	0.206575	